# Andrews University Digital Commons @ Andrews University

## Dissertations

Graduate Research

1989

## The Relationship Between Moral Judgment, Selected Personality Factors, and Marital Satisfaction

Thomas C. Wallace Andrews University, wallacet@andrews.edu

Follow this and additional works at: https://digitalcommons.andrews.edu/dissertations

Part of the Other Psychology Commons

## **Recommended Citation**

Wallace, Thomas C., "The Relationship Between Moral Judgment, Selected Personality Factors, and Marital Satisfaction" (1989). *Dissertations*. 1525. https://digitalcommons.andrews.edu/dissertations/1525 https://dx.doi.org/10.32597/dissertations/1525/

This Dissertation is brought to you for free and open access by the Graduate Research at Digital Commons @ Andrews University. It has been accepted for inclusion in Dissertations by an authorized administrator of Digital Commons @ Andrews University. For more information, please contact repository@andrews.edu.



Seek Knowledge. Affirm Faith. Change the World.

Thank you for your interest in the

## Andrews University Digital Library of Dissertations and Theses.

Please honor the copyright of this document by not duplicating or distributing additional copies in any form without the author's express written permission. Thanks for your cooperation.

## **INFORMATION TO USERS**

The most advanced technology has been used to photograph and reproduce this manuscript from the microfilm master. UMI films the text directly from the original or copy submitted. Thus, some thesis and dissertation copies are in typewriter face, while others may be from any type of computer printer.

The quality of this reproduction is dependent upon the quality of the copy submitted. Broken or indistinct print, colored or poor quality illustrations and photographs, print bleedthrough, substandard margins, and improper alignment can adversely affect reproduction.

In the unlikely event that the author did not send UMI a complete manuscript and there are missing pages, these will be noted. Also, if unauthorized copyright material had to be removed, a note will indicate the deletion.

Oversize materials (e.g., maps, drawings, charts) are reproduced by sectioning the original, beginning at the upper left-hand corner and continuing from left to right in equal sections with small overlaps. Each original is also photographed in one exposure and is included in reduced form at the back of the book. These are also available as one exposure on a standard 35mm slide or as a  $17" \ge 23"$ black and white photographic print for an additional charge.

Photographs included in the original manuscript have been reproduced xerographically in this copy. Higher quality 6" x 9" black and white photographic prints are available for any photographs or illustrations appearing in this copy for an additional charge. Contact UMI directly to order.

## U·M·I

University Microfilms International A Bell & Howell Information Company 300 North Zeeb Road, Ann Arbor, MI 48106-1346 USA 313/761-4700 800/521-0600

----

Order Number 9007145

The relationship between moral judgment, selected personality factors, and marital satisfaction

Wallace, Thomas Claude, Ph.D. Andrews University, 1989

Copyright ©1990 by Wallace, Thomas Claude. All rights reserved.



Reproduced with permission of the copyright owner. Further reproduction prohibited without permission.

\_

- ---

Andrews University School of Education

### THE RELATIONSHIP BETWEEN MORAL JUDGMENT, SELECTED PERSONALITY FACTORS, AND MARITAL SATISFACTION

A Dissertation Presented in Partial Fulfillment of the Requirements for the Degree Doctor of Philosophy

> by Thomas C. Wallace June 1989

\_\_\_\_\_

#### THE RELATIONSHIP BETWEEN MORAL JUDGMENT, SELECTED PERSONALITY FACTORS, AND MARITAL SATISFACTION

A dissertation presented in partial fulfillment of the requirements for the degree Doctor of Philosophy

by

Thomas Claude Wallace

APPROVAL BY THE COMMITTEE:

Chair Habenicht Donna

Member:

Member: Selma

Member: Sara Terian

Ex

Director, Doctoral Programs in Education

Dean, Education of

Copyrighted by Thomas Claude Wallace

All Rights Reserved

## ABSTRACT

## THE RELATIONSHIP BETWEEN MORAL JUDGMENT, SELECTED PERSONALITY FACTORS, AND MARITAL SATISFACTION

Ву

Thomas Claude Wallace

Chair: Donna J. Habenicht, Ed.D.

.....

Reproduced with permission of the copyright owner. Further reproduction prohibited without permission.

\_\_\_\_

## ABSTRACT OF GRADUATE RESEARCH

Dissertation

Andrews University

School of Education

#### Title: THE RELATIONSHIP BETWEEN MORAL JUDGMENT, SELECTED PERSONALITY FACTORS, AND MARITAL SATISFACTION

Name of researcher: Thomas Claude Wallace
Name and degree of faculty advisor: Donna J. Habenicht, Ed.D.
Date completed: June, 1989

#### **Problem**

The purpose of this study was to observe relationships between moral judgment, personality, and marital satisfaction.

#### <u>Method</u>

The population included all white, American, married couples whose religious orientation was Church of Jesus Christ of Latter Day Saints, Seventh-day Adventists, or United Methodists. A total of 264 subjects (132 married couples) completed a demographic questionnaire, the Dyadic Adjustment Scale (DAS), the Defining Issues Test (DIT), and the California Psychological Inventory (CPI). Data were analyzed using multiple-regression analysis and analysis of variance. The influence of stage of family life cycle, educational achievement level, and religion were controlled.

#### <u>Results</u>

Means and variances for the DAS and CPI resembled normative data. DIT means, however, were more than one standard deviation below normative data. Overall results showed that subjects who scored above the sample mean in norm-favoring (CPI) were more satisfied in marriage (DAS) than those who scored lower. Subjects who scored in the lowest one-third of the DIT obtained the highest marital satisfaction scores. Subjects who scored in the top onethird of the DIT obtained the lowest marital satisfaction scores.

Husbands who scored above the sample mean in selfrealization scored higher in marital satisfaction. Fathers of young children scored higher in marital satisfaction if they also scored above the sample mean in norm-favoring. Fathers of teenage children scored higher in marital satisfaction if they tended to be extraverted.

Wives who scored above the sample mean in normfavoring scored higher in marital satisfaction. Wives who had no more than a high-school education tended to be less satisfied in marriage if they scored higher than their husbands in moral judgment.

#### <u>Conclusion</u>

Husbands tend to be more satisfied in marriage when they are free of neurotic trends, mature, insightful, optimistic, and have a wide range of interests. Wives tend to be more satisfied in marriage when they are well-organized, conscientious, conventional, and dependable. Wives who have no more than a high-school education tend to be much more satisfied in marriage if the husband's level of moral judgment is equal to or higher than their own.

## TABLE OF CONTENTS

LIST OF TABLES	• •	• •	• •	•	• v
Chapter					
I. INTRODUCTION	• •	•••	• •	•	. 1
Statement of the Problem					. 4
Purpose of the Study				•	
Research Questions			•••	•	. 7
Theoretical Framework .					
Cognitive Development					
Cognitive Development i					
Personality in Marriage					
Family Life Cycle	•	• •	• •	•	. j
Moral Judgment and Mari	•••	24-11	•••		. 10
Importance of the Study	Lai				
Delimitation of the Study	•••		•••		
			• •		
Assumptions	•••	•••	• •	•	. 12
Definition of Terms		•••	• •	•	
Organization of Study .	•••	•••	• •	•	. 16
II. REVIEW OF THE LITERATURE	•••	•••	• •	•	. 17
Marital Satisfaction					. 18
Interactive Variables in M	arit	al			
Satisfaction Research			• •		. 23
Objectivity in Marital Sat				-	
Research					. 39
Moral Judgment	•••	•••	•••	•	. 42
Marital Satisfaction and M					
Moral Judgment and Religion			•••		
Education and Moral Judgmer	11L . +:	• • • • •	• •	•	. 48
Personality and Marital Sat			.on	• •	
Summary	• •	• •	• •	•	. 55

iii

----

Reproduced with permission of the copyright owner. Further reproduction prohibited without permission.

-----

. ......

\_\_\_\_

III.	METHODOLOGY	57
	Type of Research	57
	Population and Sample	57
	Instrumentation	60
	Dyadic Adjustment Scale	60
	Defining Issues Test	64
	California Psychological Inventory	67
	Procedures	70
	Procedures	73
	Research Question 1	73
	Research Question 2	75
IV.	FINDINGS.	80
	FINDINGS	80
	Correlations among the Variables	90
	Testing the Hypotheses	90
	Hypotheses tested by regression	90
	analysis	90
	Hypotheses tested by analysis of	
	Variance	98
	Summary	150
v.	SUMMARY, DISCUSSION, CONCLUSIONS, AND RECOMMENDATIONS	152
	Summary	152
	The problem	152
	Literature	153
	Purpose	156
	Methodology	157
	Findings	158
	Discussion	169
	Conclusions	177
	Recommendations for Further Study	178
	Recommendations for further Study	1/8
APPI	ENDICES	180
	A Correspondence	101
	A Correspondence	181
	C Data matrix	186
		188
REFI	ERENCES	196
VITA	A	209

iv

\_ \_

----

. \_\_\_\_ \_ \_

\_ \_

## LIST OF TABLES

1.	Reliability Data for CPI Vectors	70
2.	Frequency Distribution for Stages of Family Life Cycle	83
3.	Frequency Distribution for Religion	83
4.	Frequency Distribution for Educational Achievement	84
5.	Educational Level of U.S. White Population	84
6.	Group Means for Marital Satisfaction, Moral Judgment, and Personality	85
7.	Dyadic Adjustment Scale Results Compared to Spanier's Married Group	85
8.	Moral Judgment and Religious Affiliation	87
9.	Intercorrelation Matrix	92
10.	Summary for Correlation of DIT, P Index, Religion, Family Life Cycle, and Education to Dyadic Adjustment Scale	95
11.	Marital Satisfaction Means for Various Levels of Moral Judgment at Each Level of Control Variables	102
12.	Analysis of Variance for Various Levels of Moral Judgment at Each Level of Control Variables	103
13.	Marital Satisfaction Means for Various Levels of Moral Judgment by Internality at Each Level of Control Variables	105
14.	ANOVA for Various Levels of Moral Judgment by Internality at Each Level of Control Variables	106

v

15.	Marital Satisfaction Means for Various Levels of Moral Judgment by Norm-favoring at Each Level of Control Variables	108
16.	ANOVA for Various Levels of Moral Judgment by Norm-favoring at Each Level of Control Variables	109
17.	ANOVA for Subjects at the High School Level at Various Levels of Moral Judgment Differences and Above and Below the Mean for Norm-favoring	110
18.	Marital Satisfaction Means for Various Levels of Moral Judgment by Self-realization at Each Level of Control Variables	114
19.	ANOVA for Various Levels of Moral Judgment by Self-realization at Each Level of Control Variables	115
20.	Marital Satisfaction Means for Husbands at Various Levels of Moral Judgment Differences at Each Level of Control Variables	117
21.	Analysis of Variance for Husbands at Various Levels of Moral Judgment at Each Level of Control Variables	118
22.	Marital Satisfaction Means for Husbands at Various Levels of Husband/Wife Moral Judgment Differences by Internality at Each Level of Control Variable	120
23.	ANOVA for Husbands at Various Levels of Husband/Wife Moral Judgment Differences by Internality at Each Level of Control Variables	121
24.	Marital Satisfaction Means for Husbands at Various Levels of Husband/Wife Moral Judgment Differences by Norm-favoring at Each Level of Control Variables	123
25.	ANOVA for Husbands at Various Levels of Husband/Wife Moral Judgment Differences by Norm-favoring at Each Level of Control Variables	124

vi

26.	Marital Satisfaction Means for Husbands at Various Levels of Husband/Wife Moral Judgment Differences by Self-realization at Each Level of Control Variable	127
27.	ANOVA for Husbands at Various Levels of Husband/Wife Moral Judgment Differences by Self-realization at Each Level of Control Variables	128
28.	Marital Satisfaction Means for Wives at Various Levels of Husband/Wife Moral Judgment Differences at Each Level of Control Variables	131
29.	ANOVA for Wives at Various Levels of Husband/Wife Moral Judgment Differences at Each Level of Control Variables	132
30.	Marital Satisfaction Means for Wives at Various Levels of Husband/Wife Moral Judgment Differences by Internality at Each Level of Control Variables	134
31.	ANOVA for Wives at Various Levels of Moral Judgment Differences by Internality at Each Level of Control Variables	135
32.	Marital Satisfaction Means for Wives at Various Levels of Husband/Wife Moral Judgment Differences by Norm-favoring at Each Level of Control Variables	138
33.	ANOVA for Wives at Various Levels of Moral Judgment Differences by Norm-favoring at Each Level of Control Variables	139
34.	ANOVA for Wives With College Degrees at Various Levels of Moral Judgment Differences and Above and Below the Mean for Norm-favoring	141
35.	Marital Satisfaction Means for Wives at Various Levels of Husband/Wife Moral Judgment Differences by Self-realization at Each Level of Control Variables	144
36.	ANOVA for Wives at Various Levels of Moral Judgment Differences by Self-realization at Each Level of Control Variables	145

vii

 .

.

\_\_\_\_\_

-

37.	ANOVA for Wives With no Children at Various Levels of Moral Judgment Differences and Above and Below the Mean for Self-realization	147
38.	ANOVA for Methodist Wives at Various Levels of Moral Judgment Differences and Above and Below the Mean for Self-realization	149
39.	Variables Which Account for Higher Marital Satisfaction for Both Spouses at Various Levels of the Control Variables	170
40.	Variables Which Account for Higher Marital Satisfaction for Husbands at Various Levels of the Control Variables	171
41.	Variables Which Account for Higher Marital Satisfaction for Wives at Various Levels of the Control Variables	172

viii

- ----

\_\_\_\_\_

#### CHAPTER I

#### INTRODUCTION

Marital satisfaction is a pivotal determinant of family stability. Rising divorce rates reflect a need for exploration of alternative perspectives regarding factors influencing the marital relationship (Garrett, 1982, p. 423). Various theoretical models focus upon different aspects of marital and family concerns. Theoretical perspectives of marital and family therapy include psychodynamic models, experiential/humanistic models, the Bowen model, the structural model, the communication model, and behavioral models (Goldenberg & Goldenberg, 1985). Each model addresses specific needs (variables) within the marital relationship. The following briefly summarizes currently accepted models for marriage and family therapy.

In the psychodynamic model, therapy focuses upon the backgrounds and experiences of each family member as well as the family as a unit. A basic assumption in the psychodynamic model is that the degree to which an individual is still attached to the past is a factor in

1

present marital satisfaction. A couple's marital distress is related to the pathogenic introjects each partner brings to the relationship.

The experiential/humanistic models are essentially nontheoretical and nonhistorical. Action is stressed over insight or interpretation. Therapeutic goals consider growth through interactions between the family members and the therapist.

The primary focus of the Bowenian model is the concept of differentiation of self. Emotional health of the family is established by marital partners with similar levels of differentiation, where differentiation refers to the ability to discriminate intellectual from emotional functioning. Those who portray the greatest difficulty in differentiation function most poorly emotionally and become dysfunctional even under low stress levels.

The structural model is similar to the experiential/humanistic model by stressing action over insight or understanding. All behavior is observed within the family structure. Therapy concentrates upon demonstrations of typical conflict situations and reframing the problem as a function of the family structure.

The communication model provides for very direct intervention. Information is offered to encourage families to change patterns and rules.

The behavioral model of marital and family therapy approaches remediation from data-based intervention procedures. Positive and negative reinforcement, shaping, extinction, and social learning are among therapeutic techniques designed to confirm functionality of the family.

These models focus upon affective, behavioral, and structural aspects of the marital relationship. However, marriage is a dynamic progression of sequential stages (Burr, 1970) which may also include cognitive and moral development. The median age for marriage is 24.2 years of age for men and 21.8 years of age for women (Garrett, 1982, p. 152), while the process of moral development continues further into adulthood (Rest, 1979, p. 106).

Moral maturity may be viewed as a component of general intelligence. In a review of 52 correlations, Rest (1979, p. 147) observed that there appears to be a moderate correlation between the Defining Issues Test (a measure of moral judgment) and intellectual aptitude and achievement. Eighty-three percent of the correlations ranged from the .20s to the .50s. Verbal subtests were not more highly correlated with the Defining Issues Test

(DIT) than the nonverbal subtests, thus suggesting that the relationship between intellectual aptitude and the DIT is a "common general factor of intelligence."

Vines (1979) submitted a view of marital conflict as a function of dissonance in stages of moral development achieved by marriage partners. Subsequently, Nims (1984) conducted research which tended to support Vines' theory. Differences in moral-development scores between married couples were supportive of marital satisfaction for the spouse who revealed deficits in moral judgment. Peterson, Hey, and Peterson (1979) suggested the existence of a relationship between one's stage of moral development and choice of marital partner and/or satisfaction with the partner over the life cycle.

#### Statement of the Problem

Research findings are inconsistent in determining the extent of relationship between marital satisfaction and moral judgment. Register (1975) did not observe significant correlation between the Kohlberg Moral Maturity Scale (MMS) and the Locke-Wallace Short Marital Adjustment Test (MAT). However, Deters (1983) studied the relationship between moral judgment (using the DIT) and marital adjustment (using the Locke-Wallace Marital Adjustment Test (MAT)) among Catholics and reported a

Reproduced with permission of the copyright owner. Further reproduction prohibited without permission.

significant zero-order correlation which accounted for 17% of the variance of marital satisfaction. He was unable to replicate the results in a sample which included a multiplicity of religions, suggesting an interaction effect of religion upon the relationship between moral judgment and marital satisfaction.

Nims (1984) also examined the relationship between marital satisfaction and moral development utilizing the Defining Issues Test (DIT) and the MAT. He did not find statistical significance between spouses' moral judgment scores and their marital adjustment scores. Wives scoring above the median on the Defining Issues Test (DIT) revealed greater marital satisfaction than those below the median on the DIT. However, if the wife scored below the median on the DIT, she enjoyed greater marital satisfaction if her husband scored above the median on the DIT (Nims, 1984).

The observed inconsistencies in the zero-order correlation between marital satisfaction and moral judgment in previous studies suggest existence of extraneous interacting variables, in addition to possible differences in measures and populations. Two variables include religion (Deters, 1983) and the influence of the husband's moral judgment upon the wife's marital satisfaction (Nims, 1984). Locke (1947), in a preliminary report of a study of adjustment in marriage, concluded

that marital satisfaction was highly correlated with happiness of parents' marriages, length of acquaintance, conventionality, and sociability. These findings demonstrate the influence of family trends and personality.

Lifton (1985) studied sex, gender, and personality variables related to moral development. Moral judgment correlated positively with social introversion and social identity.

Previous studies have not considered stages of family life cycle as a possible interacting variable. Burr's (1970) results, for example, demonstrated consistently depressed satisfaction in social activities, tasks, companionship, sex, children, and the way families were handled during the "school age" phase of the family life cycle, the period of time when the oldest child was between ages 6 and 12.

#### Purpose of the Study

The purpose of this study was to observe relationships between moral judgment, personality, and marital satisfaction. The influence of stage of family life cycle, educational achievement level, and religion were controlled.

#### Research Questions

When the variables of the family life cycle, religion, and education are controlled, the following questions emerge:

 Is there a relationship between moral judgment, personality, and marital satisfaction?

2. Do differences between spouses in moral judgment and personality explain differences in marital satisfaction?

Null hypotheses are found in Chapter 3.

#### Theoretical Framework

#### Cognitive Development

Piaget (Rychlak, 1981, p. 669) viewed life as a continuing interaction of assimilation and accommodation with the environment. "Assimilation" referred to the taking in of the surrounding environment while "accommodation" accounted for the readjustment of behavior as a result of the assimilated experience. Equilibrium maintained between assimilation and accommodation accounted for how well the individual was adjusted. As the individual became better skilled in assimilating and accommodating, cognitive development occurred. Motivation for improved cognitive development was not present, however, unless there was disequilibrium between

assimilation and accommodation. As the individual experienced cycles of disequilibrium, assimilation, and accommodation, higher levels of structures developed.

Rest (1979, p. 111) reported several studies which consistently agreed that educational achievement is significantly correlated with moral development. Age, however, did not correlate well. Application of Piaget's theory of cognitive development would suggest that the formal educational process is an assimilation process which creates disequilibrium. Moral judgment is most likely a process of accommodation to the disequilibrium created by formal education. Consequently, educational achievement must be considered in order to adequately observe cognitive variance for moral judgment.

Cognitive Development in Marriage Marriage not only requires assimilation, accommodation, and equilibrium for one individual, but is subject to major adjustments by two individuals. Marriage and family are natural social systems which possess properties all their own. Within these systems evolve rules, roles, a power structure, forms of communication, and ways of negotiation and problem solving which allow tasks to be performed effectively (Goldenberg & Goldenberg, 1985, p. 17). If one or both

Reproduced with permission of the copyright owner. Further reproduction prohibited without permission.

partners have not satisfactorily developed the necessary cognitive structures to fluctuate between disequilibrium and equilibrium, the system of the relationship risks dysfunction (Peterson, Hey, & Peterson, 1979).

## Personality in Marriage

Personality variables have been identified which relate to marital satisfaction. Pickford, Signori, and Rempel (1966) employed the Guilford-Zimmerman Temperament Survey to compare three groups of married couples designated as happily married (Group A), having trouble (Group B), and on the verge of separation (Group C). Higher scores on the objectivity, friendliness, and personal relations traits were associated with marital happiness in men, but the trait of emotional stability was more important for marital happiness in women than in men.

A number of studies have shown that neuroticism, impulse control, and conscientiousness are important considerations in predicting marital happiness. These studies and the three largest longitudinal studies of marital satisfaction (Terman, 1938; Burgess & Wallin, 1953; Kelly & Conley, 1987) are discussed further in Chapter 2.

#### Family Life Cycle

Normal family development incorporates a predictable process over the family life cycle (Goldenberg & Goldenberg, 1985, p. 17). Circumstances arising at any phase of the family life cycle have forceful leverage upon relationships in subsequent categories. Goldenberg and Goldenberg (1985, p. 21) stress the need to face new tasks and learn new adaptational techniques at each phase of the family development. Successful adaptation at any phase is dependent upon the family's mastery of the tasks required by the previous phase.

Moral Judgment and Marital Adjustment

Mastery of problem solving is a key assumption in Kohlberg's cognitive-developmental theory of moral development. The development of higher levels of moral judgment as an adult task is illustrated by Kohlberg's findings of increased moral judgment scores well into adulthood (Kohlberg, 1976). Cycles of assimilations and accommodations simultaneously influencing early phases of the family life cycle and latter stages of moral development suggest that relationships exist between marital adjustment and moral judgment.

This study seeks to test the theory that moral judgment and personality are related to marital satisfaction. Due to a large number of influences upon the

marital relationship, relationships may not be directly observable without accounting for certain variables which may influence the results. In order to observe the relationships, the stage of the family life cycle, religion, and educational achievement must be controlled. In other words, the influences of the stages of the family life cycle, religion, and educational achievement are related to both moral judgment and marital satisfaction to such an extent that their influences must be removed before the relationship may be adequately studied. The theoretical basis for this study states that, once the influences of these extraneous variables are removed, moral judgment and personality are important factors in marital happiness.

### Importance of the Study

The marital relationship is the most intimate relationship in adult life. When children reach a certain age, they differentiate themselves from the family of origin in order to establish new dimensions of their own identity. The process precipitates change with every member of the immediate family. In spite of changes brought about by birth, maturation, or absence of children, married partners normally continue in a special one-to-one relationship. For the greatest degree of happiness, it is important that all aspects of

Reproduced with permission of the copyright owner. Further reproduction prohibited without permission.

the special relationship remain intact. This research expands understanding of how the newly defined orthogonal personality vectors defined in the revised California Psychological Inventory (Gough, 1987) relate to marital satisfaction. In addition, this research increases knowledge of marital relationships by observing the function of moral judgment within the marital relationship.

## Delimitation of the Study

Since family structures vary with different ethnic, religious and cultural groups (Axelson, 1985, p. 241), the sample under study was delimited to White American married couples who attended the Church of Jesus Christ of Latter Day Saints, the Seventh-day Adventist Church, or the United Methodist Church. The present marriage of these couples was the first marriage for both spouses who were parents. Previous relationships were acceptable to the study providing the subject did not procreate prior to the present marriage. Interpretation of the results should be restricted to this population.

#### <u>Assumptions</u>

The following assumptions were made for the purpose of this study:

Reproduced with permission of the copyright owner. Further reproduction prohibited without permission.

1. It was assumed that the subjects understood the instructions of the various instruments used in this study and possessed the appropriate reading and writing skills necessary to respond to the research instruments.

2. It was assumed that the subjects provided valid and reliable responses essential for the analyses.

3. Two of the religions (LDS and SDA) which were chosen for study were assumed to be conservative.

4. It was assumed that the tests reflected the subjects' true opinions about social issues, personality, and marital satisfaction.

#### Definition of Terms

The following terms used in this study are defined as follows:

Dyadic Adjustment. Within the context of this study, dyadic adjustment refers to the amount of ease with which a married couple interact and function with each other.

Extraversion. Extraversion describes individuals who are predominantly outgoing, confident, talkative, and have social poise and presence (Gough, 1987, p. 18).

Family Life Cycle. The family life cycle is a sequence of characteristic stages beginning with family formation and continuing through the life of the family to its dissolution (Duvall & Miller, 1985, p. 21).

Stages of family life cycle as used in this study are defined in Chapter 3.

Internality. Internality identifies a personality scale of the California Psychological Inventory (Gough, 1987). Subjects who score high on internality are viewed as introverted. Those who score low on internality are viewed as extraverted.

Introversion. Introversion describes individuals who are predominantly reticent, shy, reserved, moderate, modest, and reluctant to initiate or take decisive social action (Gough, 1987, p. 18).

<u>Marital Adjustment</u>. Marital adjustment is synonymous to dyadic adjustment.

<u>Marital Relationship</u>. Marital relationship is synonymous to marriage.

<u>Marital Satisfaction</u>. Marital satisfaction is synonymous to dyadic adjustment.

Marriage. Marriage refers to a monogamous, dyadic relationship between one man and one woman (Duvall & Miller, 1985, p. 6). It is a relationship recognized by society which provides for sexual relations, legitimizes childbearing, and establishes a basic system for a family.

Moral Development. Moral development is the nature and course of development of an individual's basis for making moral decisions. Kohlberg has defined a predictable sequence by which individuals develop which further expands Piaget's theory of moral development (Kohlberg, 1976).

Moral Judgment. Moral judgment refers to the underlying cognitive organization of thinking which is the basis for decisions. Such cognitive organization develops through a definite succession of transformations (Rest, 1979, p. 7, & Kohlberg, 1976).

Norm-favoring. Norm-favoring describes individuals who tend to be seen as well-organized, conscientious, conventional, dependable, and controlled (Gough, 1987, p. 19).

Norm-Nonfavoring. Norm-nonfavoring describes individuals who tend to be seen as rebellious, restless, pleasure-seeking, and self-indulgent (Gough, 1987, p. 19).

Orthogonal Personality Factors. Orthogonal personality factors refer to personality variables which are essentially unrelated to each other. The present study includes two orthogonal personality factors which have been defined through factor analyses of the California Psychological Inventory (Gough, 1987, p. 16). The two factors include measures of internality

and norm-favoring. A third factor, self-realization, correlates -.17 with internality and -.18 with norm-favoring.

<u>Self-realization</u>. Self-realization describes individuals who are relatively free of neurotic trends and conflicts, moderate, mature, insightful, optimistic, and have a wide range of interests (Gough, 1987, p. 20).

#### Organization of Study

This study is organized into five chapters. Chapter 1 presents the statement of the problem, purpose of the study, the research questions, theoretical framework, importance of the study, delimitation of the study, assumptions, and definition of terms. Chapter 2 contains a survey of the literature pertaining to marital relationships and moral development. Chapter 3 defines the procedures used for the selection of the sample. Also included in chapter 3 is a discussion of methodology employed in data collection and statistical analysis. Chapter 4 presents an analysis of the data and interpretation of findings. Chapter 5 discusses the findings and includes conclusions and recommendations.

#### CHAPTER II

#### REVIEW OF THE LITERATURE

Three areas of the literature are considered in Chapter 2: (1) marital satisfaction, (2) moral judgment, and (3) personality and marital satisfaction. The section on marital satisfaction first reports some of the major studies which serve as historical guides within the marital satisfaction literature and ends with a review of the major topics over the past 10 years (1979-1988). This is followed by a review of problems encountered by attempting objectivity in the selfreporting of marital satisfaction. This section closes with a discussion of variables which are known to be related to marital satisfaction and may interact with other correlates.

The second major section of the chapter reviews the literature related to moral judgment. First is a review of Kohlberg's theory of moral judgment, its basis in Piagetian theory of cognitive development, and how the theory is applied in current research. Second, past research relating moral judgment and marital

satisfaction is reviewed. The third section includes a review of studies which relate moral judgment to religion. The final section is a review of studies which relate moral judgment and level of educational achievement.

The third major section of the chapter includes a review of the most comprehensive study to be undertaken on the relation between personality features and marital compatibility. The latter part of this section includes a review of the major studies involving personality and marital satisfaction over the past 10 years (1979-1988).

### Marital Satisfaction

In spite of concentration on marital satisfaction by the research community, continued evidence is present of a decline in the positive relationship between married subjects and happiness (Glenn & Weaver, 1988). However, while the concept of marital satisfaction continues to be widely explored, significant questions regarding the definitive values of marital satisfaction research have been presented and discussed (Hansen, 1981). Attempts by researchers to ascertain the levels of marital satisfaction and to anticipate the success of marriage go back to the earlier part of this century and continue today. An example of marital research during the early part of this century was reported by Terman

(1950). Terman cited Patterson's findings which were reported in <u>The Journal of Delinquency</u> in 1922. Patterson stated that there was no greater tendency for marital difficulties among subjects who were married before 26 years of age than among those who married later in life.

Hart and Shield (1926) attempted to compare 500 couples who suffered from marital discord and who appeared in the Philadelphia Court of Domestic Relations to a random sample of 500 couples who applied for marriage licenses. The independent variable was age. They concluded that couples who married under the age of 20 years were at greater risk of divorce than those who waited until they were older. In fact, the risk was 2 to 10 times greater than for older couples. It is interesting to note that the Patterson study and the Hart and Shield study were both conducted with Philadelphia subjects.

Burgess and Cottrell (1939) conducted a largescale study of marital adjustment. In addition to responding to questions which addressed marital adjustment, the subjects also disclosed information which pertained to premarital background of both spouses and attitudes and experiences which had surfaced subsequent to marriage. Couples were also asked to rate the happiness of their marriages. Burgess and Cottrell

found socialization, conformity to social rules, and respect for convention to be important in preparing a person for adjustment in marriage (p. 346).

The three largest longitudinal studies of marital satisfaction have been conducted by Terman (1938), Burgess and Wallin (1953), and Kelly (Kelly & Conley, 1987). Locke was a contemporary researcher during these early investigations into marital satisfaction. His studies are reported in chronological sequence with the longitudinal studies (1947).

Terman's (1938) study of marital adjustment examined the relationship between sexual factors, personality, and marital adjustment. He concluded that happily married men tend to be more conservative than unhappily married men (p. 163). This appeared to be especially true in the areas of sexual morality and religious attitudes.

In a study of 925 married and divorced subjects, Locke (1947) compared marriages which ended in divorce with marriages considered to be most happy. He observed that 41% of the divorced men, as compared to 23.8% of the married men, did not belong to a church. Among women, 28.6% of the divorced women and 14.9% of the married women did not belong to a church. He concluded that affiliation with a church was associated with marital adjustment. He also reported that the longer an

individual attended Sunday School as a child, the greater were the chances of marital success later in Those who terminated Sunday school attendance too life. early were more closely associated with the divorced group. Forty-six percent of the men in the divorced group never attended church as compared to 15% of the married men. Thirty-nine percent of the women in the divorced group had never attended church as compared to 15% of the happily married group of women. Locke concluded that church attendance was a major factor in marital adjustment. His conclusions were consistent with earlier researchers who proposed that conventionality, sociability, and stability of persons provide the elements for a happy marriage (Burgess & Cottrell, 1939, p. 346).

Burgess and Wallin (1953) deviated from research designs of previous studies by beginning with engaged rather than married couples. After completing the initial questionnaires, the engaged couples were followed three to five years later to ascertain the predictiveness of the premarital testing. Burgess and Wallin (1953) reported mild correlations between emotional health before marriage and marital adjustment scores after three to five years of marriage.

The most comprehensive analysis to be undertaken is the Kelly Longitudinal Study (KLS). This continuing

longitudinal study (Kelly & Conley, 1987) encompasses both marital stability and marital satisfaction for 300 couples over a period of 45 years and is still continuing. Further discussion of the KLS follows later in this chapter, since it focuses upon measures of personality within the marital relationship.

Although each of the landmark studies reviewed above presents a unique contribution to the understanding of marital satisfaction, each also tends to present common elements which involve personality variables and social background influences. The following studies are a review of the literature pertaining to marital satisfaction for the past 10 years.

Central to the question of marital satisfaction is the issue of personal happiness. Wilson and Wilson (1977) described family adjustment as related to successful involvement with people. Costa and McCrae (1980) carried the concept further by examining the relationship between personality and happiness or subjective well-being. They noticed that happier people are well-adjusted, high in ego strength, high in selfesteem, and socially involved. Their proposed model of the relationships between personality and happiness included extraversion and sociability as contributing toward positive affect. Even though these researchers found a positive relationship between extraversion and

happiness in general, extraversion did not appear to relate in the same manner with marital satisfaction.

### Interactive Variables in Marital Satisfaction Research

Schumm, Hess, Bollman, and Jurich (1981) and Schumm and Kirn (1982) expressed a need to attend more carefully to the manner in which family social scientists integrate theory, measurement, and statistical analysis in research. They noted that studies fail many times because adequate attention had not been given to variables which may interact with the independent and dependent variables. Three variables are discussed in view of their potential to mask true relationships within the study through variable interaction. These variables include stage of family life cycle, religion, and educational achievement.

## Family Life Cycle

Duvall and Miller (1985, p. 26) suggested eight stages through which a family typically progresses: The beginning family (I) is a married couple without children. Childbearing families (II) include families whose oldest child ranges from birth to 30 months. Families with preschool children (III) identifies the group which includes an oldest child who is between 30 months and 6 years of age. Families with school children (IV) have

Reproduced with permission of the copyright owner. Further reproduction prohibited without permission.

the oldest child between 6 and 13 years of age. Families with teenagers (V) are those with the oldest child between 13 and 20 years of age. From the time the first child leaves home until the last child leaves is called the launching period (VI). Families in the middle years (VII) include the years after the children have left until retirement. Aging families (VIII) include the time from the beginning of retirement until the death of both spouses.

Burr (1970) randomly sampled and interviewed 116 intact couples. He originally defined the family life cycle levels consistent with Duvall's definition (Duvall & Miller, 1985, p. 26). However, when the sample was divided according to Duvall's levels, several categories were very small. For this reason, Burr collapsed the childbearing and preschool stages (stages II and III) into one stage. His results demonstrated consistently depressed marital satisfaction in social activities, tasks, companionship, sex, children, and the way finances were handled during the "school age" phase of the family life cycle. The most abrupt overall changes seem to be from the preschool stage to the school stage when there is a discernible drop for both husbands and wives in satisfaction in most of the areas. Perhaps the most interesting finding in the Burr study was that on every measure, husbands were less happy in the pre-child

stage than the stage where there were preschool children in the home. The lowest levels of marital satisfaction appeared to be at the school-age stage for both husbands and wives. The adolescent stage was also troublesome.

Burr's overall findings revealed that there is no methodical decline over the life cycle in satisfaction with the six areas of marriage studied. Furthermore, there appears to be a slight increase in several of the areas. This phenomenon appears to be due to the increase in marital satisfaction in most areas during the post-parental and retirement stages.

Ryder (1973), in a longitudinal study which included 112 married couples between the ages of 18 and 27 when first married, defined three groups: (1) Childless couples, (2) Child couples, and (3) Pregnant couples. The purpose was to study marital dissatisfaction and love sickness during the early stages of marriage. He administered measures of marital dissatisfaction and love sickness. The marital dissatisfaction scale was actually the Locke-Wallace Short Marital Adjustment Test (1959) scaled in the negative direction. His measure of lovesickness consisted of 32 items, such as, "I wish my spouse paid more attention to me." Data revealed that women who have a child become more likely to report that their husbands are not paying enough attention to them.

Overall, the three groups did not differ in marital dissatisfaction. Wives who had a child tended to be more "love sick." They felt that their husbands did not pay adequate attention to them. Closer examination of Ryder's data revealed that the highest "love sick" mean was actually among pregnant wives. Even though Ryder did not observe differences among the three groups in marital dissatisfaction, the following study suggests there is additional stress upon the marital relationship during pregnancy.

Snowden, Schott, Awalt, and Gillis-Knox (1988) studied marital satisfaction in a sample of 106 women recruited at a military medical center. The average age of the participants was 24.66 years. Eighty percent were white and 20% were of other racial groups, predominantly black. Nearly all subjects (92.6%) had completed high school and 11.8% had graduated from college. Nearly half of the subjects had given birth to children previously. Most husbands (90.4%) were enlisted military personnel. The women were evaluated both early and late in pregnancy. Stress upon the marital relationship became apparent during pregnancy. Additionally, first-time parenthood and participation in religious activities appeared to be associated with marital satisfaction. In view of these findings, "love

sickness" as described by Ryder (1973) may have actually come into existence during pregnancy.

Many researchers have investigated the relationship between the family life cycle and marital satisfaction. Studies supporting the curvilinear relationship between family life cycle and marital satisfaction have been reported by Campbell (1972), Corrales (1974), Lee (1978), Boles (1984), Otero de Sabogal (1984), and Hunt (1985). The following studies present varying opinions regarding the importance of one stage of family life cycle over the other in the marital relationship.

Anderson, Russell, and Schumm (1983) first reported a significant but slight relationship between family life cycle and marital satisfaction. Upon reexamination, however, Schumm and Bugaighis (1986) reported that the decline in marital satisfaction during the middle stages of the family life cycle can be explained as the result of the impact of preschool children on a specific group of low income mothers. Their results indicated a severe source of distress for a small group of wives rather than a mild problem for all wives.

Other researchers placed much more emphasis upon the influence of family life cycle in relationship to marital satisfaction. In general, Rollins and Feldman

Reproduced with permission of the copyright owner. Further reproduction prohibited without permission.

(1970) observed family life cycle experiences to be more associated with wives than with husbands. Feelings about marital interaction and the dependent children were issues associated with negative feelings about marriage by wives. Rollins and Cannon (1974) suggested that a theory of role strain during the middle years would explain the curvilinear relationship. Their study revealed that marital satisfaction was favorable upon entry into marriage; however, there was a general depression in satisfaction during the years in which the couple was raising children. Satisfaction appeared to increase conspicuously after the children left home. Similar findings were reported by Anderson (1983).

Abbott (1983) also observed greater marital satisfaction among women who had no children compared to women with children. He studied four categories: mothers with infants (2 years and younger), mothers with preschool children (3-5 years), mothers with both infant and preschool children, and wives with no children. He found that differences between mothers and childless wives were primarily observed among mothers who have two children and mothers who have male children. Mothers with female children, especially those with female infants, reported no significant differences in marital adjustment and emotional quality of the home environment when compared with the childless wives.

Steinberg and Silverberg's (1987) study of 129 couples focused upon the reduced level of marital satisfaction during the stage when children are in the teen years. They suggested that two factors which tend to have a negative influence upon marital satisfaction include distancing in the father-son and mother-daughter relationships and wives' concerns about midlife identity issues.

Researchers have not been unanimous in their endorsement of curvilinearity in the relationship between marital satisfaction and the family life cycle. Spanier, Lewis, and Cole (1975) set out to ascertain whether a curvilinear relationship provides a leastsquares regression curve which explains variance better than a linear regression line and to determine the significance of the relationship. Three coordinated studies in Ames, Iowa; Newark, Ohio; and Classic County, Georgia; were conducted. Their results were mixed. The Ohio study revealed a curvilinear relationship. However, the Georgia and Iowa studies were not significantly curvilinear. They observed that linear interpretation accounted for more variance than the curvilinear.

Research data can be misinterpreted, however, when the family life cycle is not considered in marital satisfaction research. Luckey (1966) attempted to observe relationships of personality and marital

satisfaction with age and number of years married for the respondent and the respondent's spouse. Her subjects included 80 married couples who had been married from 2 to 21 years, averaging 7.7 years. The men in her study ranged from age 26 to 46, with a mean of 31.9 years. Women's ages ranged from 26 to 41 years, with a mean of 29.5 years. She discovered a number of personality traits which were positively correlated with marital satisfaction. These included such traits as being well liked by others, self-confidence, independence, modesty, gracefulness and appreciativeness, cooperativeness, and responsibility. With the exception of being friendly and sociable, all personality traits were negatively correlated with number of years married and age. Subjects appeared to see fewer socially desirable personality characteristics in their mates the longer they were married.

This study is best interpreted in view of the total family life cycle. In considering the ages and number of years married, it becomes apparent that the study only represented approximately half of the family life cycle. It has been observed, even though other studies have not fully agreed, that marital satisfaction and family life cycle tend to have a curvilinear relationship. It is likely in the Luckey study that none of the subjects had completed the launching period

(if they had even begun). Furthermore, if Luckey would have included retired couples in her sample and excluded young, childless couples, her results might have yielded a positive correlation between marital satisfaction and age. Furthermore, if she would have included the complete family life cycle, she might have observed no correlation. The study, in its present form, pictures marriage as an experience to avoid. Obviously, such conclusions are not supported in the general literature.

### <u>Religion</u>

Religion, as a variable considered to possibly interact within the present study, is defined denominationally (e.g., Latter Day Saints, Seventh-day Adventists, and Methodists). In the literature, however, religiosity, the intensity of the religious experience, is a more commonly discussed variable than denominationalism. Because the study of religiosity exceeds the scope of the present study, the following studies are presented only as illustrations of studies which have examined relationships between religiosity and marital satisfaction.

Hunt and King (1978) illustrated how religion was multidimensional in its relationship to marital satisfaction in their study of 17 variables within their definition of a religious construct. Six of the

religiously defined variables related to marital satisfaction. These included positive beliefs about religion, greater effort, more religious participation, more agreement about religion, greater tolerance, and higher extrinsic motivation toward religion.

Religiosity was viewed by Wilson and Filsinger (1986) in terms of religious rituals, knowledge, experiences, consequences, and beliefs. In their study of 190 married couples, religiosity appeared to be generally related to marital satisfaction where religious rituals, experiences, and beliefs are concerned. Religious consequences appeared to relate to marital satisfaction for husbands more than wives.

As observed by Snowden, Schott, Awalt, and Gillis-Knox (1988) in their study of 106 wives of military personnel, religious participation appeared to have a small positive influence upon marital satisfaction during pregnancy. Earlier studies, however, help to clarify the relationship between religion and marital satisfaction.

Thomas and Henry (1985) suggested that there was an expressed movement to link religion and family studies in the early 1950s which seemed to end in the early 1960s. The movement appears to have revived during the early 1980s. The revival, however, appears to bring with it an attitude which is generally more

open to investigation than the earlier movement in the 1950s. D'Antonio, Newman, and Wright (1982) observed that only one of the 57 family texts which they surveyed contained a chapter on religion, and only two religious texts published before 1978 had a chapter on the family.

Evidence of this "revival" of religious investigation within marriage and family studies was apparent in a study by Schumm, Bollman, and Jurich They specifically referred to the studies (1982). conducted by Edmonds (1967) and Edmonds, Withers, and Dibatista (1972) which viewed the observed relationship between religiosity and marital satisfaction as "spurious artifacts of the common contamination of such measures with social desirability." Glenn and Weaver (1988) supported the argument to discount the importance of observed relationships between religious variables and marital happiness. Schumm, Bollman, and Jurich (1982) found evidence within their samples that religiosity correlated with marital satisfaction among subjects who did not respond high in conventionality. Thus the stage was set for reinvestigation of religiosity within the marital relationship.

Hatch (1985) observed that church attendance and frequency of prayer did not individually and significantly contribute to marital satisfaction. Religious beliefs and shared devotional times between

husband and wife, however, appeared to be significantly related to the marital satisfaction of seminary couples in her study. She discovered that a change in religiosity that did not nurture emotional intimacy might not have an effect upon marital satisfaction. Such findings suggest that religiosity (i.e., the <u>intensity</u> of the religious experience) and the interacting variables which relate to intimacy appear to be more closely related to marital satisfaction than merely church membership or church attendance. Similar findings were reported by Brown and Annis (1978).

Baxter (1982) found that denominational affiliation makes no difference in marital adjustment. Bahr and Chadwick (1985) also found no difference between Catholics and Protestants. However, there was a significant difference between those who reported a church preference and those who did not. They also discovered that couples who attended church at least monthly tended to marry and stay married more than those who did not. The following study (Heaton, 1984), however, provides evidence of differences between religious orientations as well as differences in marital satisfaction when spouses subscribe to same or different religions.

Heaton (1984) reported a positive relationship between religious homogamy and marital satisfaction. He

reported the ratio of respondents who acknowledged being very happy to those who reported being pretty happy or not too happy for each combination of husbands' and wives' religion. He found that homogamous marriages tended to report greater marital happiness than heterogenous marriages. Among the homogamous groups, the Jewish group revealed the highest ratio for both husbands and wives, while the lowest ratio for marital happiness was observed among the group which did not subscribe to a religion.

Deters (1983) provided the most convincing argument in favor of controlling religion in a study which correlated moral judgment with marital satisfaction. A relationship between the variables was not observed until he controlled for religion. When he observed the relationship only among his Catholic subjects, he discovered that moral judgment accounted for 17% of marital satisfaction.

The literature regarding the relationship between moral judgment and religion is still not well defined. There is general agreement that religion is multidimensional (Brown & Annis, 1978). However, there is disagreement regarding denominational effects of religion upon the marital relationship. Studies conducted by Baxter (1982) and Bahr and Chadwick (1985) seem to indicate that church attendance versus no church

attendance relates to marital satisfaction, but they observed no denominational differences. Heaton's (1984) study, however, suggested that denominational differences are present. All of the investigators seem to agree that there is a difference in marital satisfaction between religiously oriented homes and nonreligiously oriented homes. In view of Deters' findings, however, where a relationship between moral judgment and marital satisfaction was only observed when religious denomination was controlled, religion appears to be interactive between the two variables. Since Deters' findings have advanced more understanding of the relationship between moral judgment and marital satisfaction than any other study to date, religion may be considered an important interacting variable.

# Educational Achievement

In a discussion of homogamy of education for married couples, Garrett (1982, P. 126) reported that the historical tendency across several generations of American marriages has been toward a high degree of educational homogamy between marriage partners. In more recent times, however, educational homogamy has been reduced. This difference appears to have become more observable during the 1960s when the age at marriage was going down. It has become apparent that, although

marital partners have achieved different educational levels when they marry, one or both of the marital partners may plan to seek an academic degree later. In the early part of this century, men and women tended to complete formal education before marrying. This pattern does not seem to be practical or desirable today. In summarizing current trends in coordinating educational achievement with marriage, Garrett (1982, p. 127) observed Hogan's (1978, 1980) findings that men who alter the normal life-cycle transition of schooling, followed by the first employment and then marriage, have a higher rate of divorce than normal.

Houseknecht, Vaughan, and Macke (1984) observed a high rate of separation and divorce among highly educated women in the United States. In their study of 663 women graduates, they discovered that marital disruption tended to follow the women's attempt to combine a family and a career. Disruption often occurred before the women completed their education. The educational experience appeared to produce a considerable amount of tension in marriages. They observed marital disruptions to be more likely among women who begin graduate studies after marriage rather than before. The authors point out that wives whose husbands are unwilling to support the wife's career advancement tend to find it more difficult to progress

in the career. These researchers suggest that Hogan's findings of the "education, job, marriage" sequence most likely applies to women's experience as well as men's.

In a random sample which included 10% of all divorces granted in California in 1977, Maneker and Rankin (1985) studied the relationship between age at marriage and educational levels for husbands and wives with duration of marriage. Educational levels were defined by five groups: eighth grade or less, ninth to eleventh grades, twelfth grade, one to three years of college, and four or more years of college.

Surprisingly, they discovered the longest duration of marriage before the divorce was within the group with eight or fewer years of education. The second longest duration included the group with four or more years of college. The other groups followed with shorter marriages before divorce. This curvilinear relationship between educational achievement and length of marriage before divorce seems to contradict the hypothesis that longer duration of marriage is positively associated with higher educational levels. Even though these findings are available for divorced couples, the question of the relationship between educational achievement and marital satisfaction for healthy and stressful marriages remains.

Reproduced with permission of the copyright owner. Further reproduction prohibited without permission.

Objectivity in Marital Satisfaction Research Perhaps the most profound criticism of marital satisfaction research relates to measures of marital quality which are so tainted by social desirability and conventionality that their value is questioned. Edmonds (1967), one of the early critics of attempts to measure marital satisfaction, defined conventionalization as "the extent to which a person distorts the appraisal of his marriage in the direction of social desirability." In other words, there is a tendency to pretend something is the situation with minimal interest as to whether it is true or not. Edmonds developed long and short scales measuring conventionalization in marriage composed of items to which respondents answered either "true" or "false", relying upon whether the statement was applicable to them, their spouse, or their marriage. The statements were worded in such a manner that the conventional response to each was too perfect to be regarded as possible. The following is an example of a conventional true-false statement: "My mate completely understands and sympathizes with my every mood." Fifty statements were included in the Marital Conventionalization Scale. Edmonds, Withers, and Dibatista (1972) reported correlations between marital conventionalization and marital adjustment of .63, .53, and .70, with the Locke-Wallace short form scale of

marital adjustment (Locke & Wallace, 1959). They asserted these results demonstrated contamination in the Locke-Wallace Short Marital Adjustment Test. They concluded that the respondents tended to mislead themselves and others into believing their marriages were better than what they really were.

Since failure or unhappiness in marriage is considered by many to be a disgraceful experience, respondents who are sensitive to social desirability would be reluctant to admit to marital difficulties. Such a tendency to conceal marital unhappiness would probably not be normally distributed in the population. It would probably be most prominent among respondents who feel a strong need for being considered socially acceptable or presenting themselves in a socially desirable manner. The question arises, "Do the marital adjustment instruments measure the respondent's actual happiness or do they inform the researcher of the degree of happiness thought to be socially correct?" In his study of 48 married couples, Hawkins (1966) administered the Marlowe-Crown Social Desirability Scale (M-C SD) and the Locke-Wallace Short Marital Adjustment Test. The M-C SD items all represented behaviors or feelings which are considered extremely desirable in American culture, but which are unlikely to occur. Some of the items come from the MMPI "lie" scale. The Locke-Wallace Short

Marital Adjustment Test (1959) is a highly reliable measure of marital adjustment. Hawkins found negligible correlation between the two instruments and concluded that negligible contamination of marital satisfaction by social desirability was present.

In an effort to eliminate confusion regarding the effects of marital conventionalization and social desirability upon self-reported marital adjustment scales, Hansen (1981) actually contributed greater confusion. He questioned the validity of Edmonds' true/false format on the Marital Conventionalization (MC) Scale (Edmonds, 1967). In replicating the study, he offered the true/false format as well as a forcedchoice format. Results surprisingly yielded correlation coefficients of .648 (males), .629 (females), and .636 (total) using the true/false format and correlation coefficients of .734 (males), .787 (females), and .762 (total) using the forced-choice format. All correlations were significant at the .001 level. There appears to be a rather strong tendency for respondents to distort the appraisal of their marriage in the same direction as the actual adjustment of the marriage. This did not appear to be contaminated, however, by the respondent's perception of social desirability separately from perception of the quality of his/her marriage. Furthermore, it did not suggest that marital

adjustment measures are contaminated by conventionality to the point that they must not be used. It simply means that a self-reporting measure of marital satisfaction is not an unbiased measure of the construct. Fortunately, however, the bias appears to be in the same direction as adjustment, thus avoiding complication by interaction.

# <u>Moral Judgment</u>

The three major theories of morality include Kohlberg's cognitive developmental theory (Locke, 1986), Haan's interactive theory (Haan, 1986), and Hogan's socioanalytic theory (Lifton, 1985). The cognitive developmental theory, which dominates the literature and is fundamental to the present study, is the only theory discussed in this literature review. It was based upon Piaget's studies of moral development of children (Piaget, 1965) and may be summarized under four major points. First, an individual has an underlying cognitive structure with which to interpret moral issues. Second, Piaget viewed the child's first experience of social rules as an heteronomous morality followed by autonomous morality. Heteronomous morality typifies a child's thinking until about 10 years of age. Rules are of primary importance and consequences of an act determine guilt regardless of the intentions of the

individual. Autonomous morality is observed in children 11 years of age and older. The older children consider rules to be flexible. Intentions are considered in determining guilt. The third issue referred to the concept of justice in moral development. Rest (Rest, 1979, p. 20) viewed Piaget's concepts of fairness and justice as opinions about the balancing of individual interests and the benefits of cooperation. Piaget's fourth point reflected a fundamental reorganization of an individual's behavior and thinking as development progressed from heteronomous to autonomous morality.

Kohlberg (1976) defined three levels of moral judgment which were expressed in six stages of reasoning. Level I (preconventional) includes stages 1 and 2. Stage 1 reasoning reveals heteronomous morality which typifies moral reasoning to avoid punishment. Stage 2 reasoning recognizes the interests of others. Moral actions reflect mutuality of interest.

Level II (conventional) consists of stages 3 and 4. Stage 3 moral judgment relates to social expectations, trust, loyalty to society, and gratitude. Stage 4 considers the social system and conscience. In stage 4 reasoning, laws and rules are to be upheld except in those few cases in which the laws or rules conflict with other social duties.

Level III (postconventional) incorporates stages 5 and 6. Stage 5 reasoning considers social contract or utility and individual rights. The individual whose modal preference is at stage 5 is aware of a variety of values and opinions held by others. Rules are relative to the larger group. Stage 6 reasoning exemplifies universal ethical principles. When laws infract upon the individual's principles, the person reasons in accordance with the principle.

## <u>Marital Satisfaction and</u> <u>Moral Judgment</u>

Several attempts have been made to observe a relationship between marital satisfaction and moral judgment. Register (1975) did not observe a correlation between the Kohlberg Moral Maturity Scale (MMS) and the Locke-Wallace Short Marital Adjustment Test (MAT). Nims (1984) also examined the relationship between marital satisfaction and moral development utilizing the Defining Issues Test (DIT) and the MAT. Even though he failed to find statistical significance between spouses' moral-judgment scores and their marital-adjustment scores, wives scoring above the median on the Defining Issues Test (DIT) revealed greater marital satisfaction than those below the median on the DIT. However, if the wife scored below the median on the DIT, her marriage tended to be more satisfying to her if her husband

scored above the median on the DIT (Nims, 1984). Nims suggested stage of family life cycle should be controlled in subsequent studies.

Much is yet to be learned about the influence of religion upon moral judgment, since religion appears to be multidimensional. However, Deter's (1983) results suggest that religion may be an interactive extraneous variable when observing relationships and differences in moral judgment and marital satisfaction. He studied the relationship between moral judgment and marital adjustment among Catholics and reported a zero-order correlation which accounted for 17% of the variance of marital satisfaction. He was unable to replicate the results in a sample which did not control for religion.

### Moral Judgment and Religion

Kohlberg (1967, p. 180) originally found no differences in moral judgment due to religious belief. He further noted that children go through the same stages at much the same rate when social class and village-urban differences are held constant. He appeared to have extended his theory beyond his ability to test, however, when he theorized (Kohlberg, 1967, p. 180) that moral action is not justified in terms of conformity to God or to the religious community. Conforming to the religious community may be tested by

objective measures. However, in this study it is assumed conformity to God surpasses any researcher's capacity for observation. An error of such magnitude, however, does not diminish the value of Kohlberg's concept of sequential developmental stages of moral judgment, although it may affect the stage responses of individuals who think in terms of conformity to God.

Ernsberger and Manaster (1981) found differences among denominations in moral judgment. Unitarian-Universalists scored the highest on the Defining Issues Test (DIT) (mean = 51.9), while the Conservative Baptists scored lowest (mean = 31.4). United Methodists and Missouri Synod Lutherans fell in the middle (means = 40.4 and 40.6, respectively). They concluded that their findings challenged Kohlberg's assertion that religious variables have little effect on moral judgment. Such a conclusion appears to be an overstatement. These findings may challenge Kohlberg's statements regarding the rate and extent of moral judgment among different religions. However, the study does not dispel the concept of sequential stages of moral judgment for different religions.

Rest (1979, p. 115) reported striking differences in moral judgment scores between the liberal and conservative church groups. Individuals who attend liberal churches tend to score higher in moral judgment

than those who attend conservative churches. Congregations varied according to the official thinking of the church. Interpretation of these results is speculative at best. One view suggested that conservative churches with well-defined codes or standards may not tolerate experimentation in the presence of cognitive disequilibrium, thus creating a cognitive impoverishment. Another view, which is consistent with the Social Learning perspective, suggests that these findings are due to imitation of the moral judgments of others within their respective groups. Sapp's (1986) study of 1,270 active members of six major denominations supported the concept of differences in moral judgment between different denominations. He concluded that if denominations were ranked on a fundamental-to-liberal theological continuum, increases in DIT scores would follow the continuum.

O'Gorman (1979) explored the relationship of religious knowledge and moral judgment. Both Catholic and non-Catholic high school students who had higher religious knowledge scores also had higher DIT scores. Childerston (1985) discovered that religiosity was related to one's level of moral judgment, and personality type may be one factor which contributes to the maintenance of an ideological commitment.

Subjects from different religious denominations apparently respond differently to moral judgment testing. There is no evidence in the literature, however, suggesting there is a difference in developmental sequencing for individuals of different religions. The predictability of the developmental sequence appears to be consistent regardless of religious orientation. The only developmental differences appear to be in extent and rate of development. Even these concepts may be related more to religiosity than to religion.

#### Education and Moral Judgment

Formal education is highly associated with development in moral judgment (Rest, 1988). In studying the relationship between education and moral judgment among school-age children, the question of age as a meaningful variable is an issue. Rest (1979, p. 111) reviewed four independent studies which examined the relationship between education and moral judgment among adults. These studies were conducted by Dortzbach (1975), Coder (1975), Crowder (1976), and Rest (1977). A 10-point average increase in DIT scores appears with each increase in the level of education (high school, college, graduate school).

Dortzbach (1975) randomly selected an adult sample from the Eugene, Oregon, voter registration lists and

administered the DIT. His results indicated that moral judgment was more highly correlated with education than age. Coder (1975), Crowder (1976), and Rest (1977) found similar results in their respective studies of subjects in a religious education program, volunteers from a naval base, and subjects randomly selected from the Minneapolis, Minnesota, telephone book. Moral judgment appears to increase as educational level increases.

# Personality and Marital Satisfaction

Kelly and Conley (1987) described two major theoretical perspectives on the problem of marital compatibility. The first position suggests intrapersonal causes of compatibility and is most accepted among psychoanalysts. In the intrapersonal theoretical perspective, the personality characteristics of the marriage partners influence the stability and contentment of the relationship. They noted that neuroticism has historically been the most often identified source of marital instability.

The second viewpoint involves interpersonal sources of compatibility and is most favored among behaviorally oriented marriage therapists. Dysfunctioning marriages are seen as dysfunctional behavioral exchanges that are typified by high ratios of punishments to rewards. The spouses are viewed as

lacking in basic social skills and more likely to react to and reciprocate the inappropriate behavior of the other partner.

To Kelly and Conley (1987) these two perspectives on marital compatibility are not mutually exclusive. They also consider high levels of neuroticism of either partner to result in dysfunctional behavioral exchanges. They consider high levels of neuroticism on the part of either partner to be related to a conflictual physiological connection and the establishment of dysfunctional cognitive and behavioral patterns. This view has been supported by other researchers. For example, Vaillant (1978) reported a strong relationship between psychopathology and poor social adjustment as well as low levels of marital adjustment.

In the Kelly Longitudinal Study (KLS), stability and satisfaction were analyzed both separately and in combination. Personality, social background, and attitudes before marriage were assessed. The findings of the KLS portray the most exhaustive analysis yet to be published during the present century regarding the relation between personality characteristics and marital satisfaction (Kelly & Conley, 1987). The subjects in this study were chosen by E. Lowell Kelly between 1935 and 1938 for a study of marital compatibility. Three hundred couples were followed from their engagements in

the 1930s until 1980. All were volunteers who had been recruited through newspaper and other advertisements. At the time of entry, 90% of the subjects lived in Connecticut or a neighboring state. The mean IQ for the subjects was 113. Ninety-four percent of the subjects ranged between 20 and 30 years of age at the time of entry into the study. All were white. In 1980, the median age of the subjects was 68 years. Sixty-nine percent specified they were Protestant, 9% Catholic, 7% Jewish, and 15% reported no religious affiliation.

Major data collections took place during 1935-38, 1954-55, and 1980-81. Annual reports were provided by the subjects from 1935 to 1941. Data collection was disrupted in 1941 due to World War II.

The major findings of this study have shown that men who divorced late in life differed from the stably married group in having higher levels of neuroticism and social extraversion and lower levels of agreeableness. They were also inclined toward closer families of origin and more premarital sexual activity. The only difference between the women who divorced late in life compared to stably married women was a higher level of neuroticism in the divorced women. Women who divorced late in life were similar to women who divorced early in life in sexual attitudes and behavior. For both sexes, there was a positive correlation between parental

Reproduced with permission of the copyright owner. Further reproduction prohibited without permission.

divorce and one's own divorce. The relationship did not reach statistical significance for either men or women, however.

Marital satisfaction among intact marriages was negatively related to neuroticism for both sexes. Impulse control was positively related to marital satisfaction in both sexes in the 1980 data. The level of tension found in the family of origin negatively predicted marital satisfaction for men. Findings were different for women. The psychosocial instability of their original family and their preadult nonconformity to social ideals negatively predicted marital satisfaction among women. The number of stressful life events experienced during early adulthood was a negative predictor of marital satisfaction for both sexes. Favorable attitudes toward conventional order in the family was a major predictor of marital satisfaction for Premarital romantic and sexual involvements were men. negative predictors of marital satisfaction for both men and women. Frequent sexual intercourse was a positive predictor of marital satisfaction.

For men who were divorced or dissatisfied in a stable marriage, neuroticism was considerably higher than among men who were satisfied in their marriages. Among women, the divorced group was higher in

neuroticism than either the satisfied or dissatisfied stably married group.

Three personality characteristics predicted approximately 15% of the variance for a composite measure of marital compatibility. These included husband's neuroticism, wife's neuroticism, and husband's impulse control. The overall predictability of marital compatibility for prenuptial characteristics of the spouses included 24% of the variance. Sexual history variables accounted for 9% of the variance. Results similar to the KLS have been observed by Terman (1950) and Burgess and Wallin (1953).

Other studies have revealed a positive relationship between neuroticism and marital dysfunction. Neuroticism seems to point toward a number of behaviors which are seen in troubled marriages. Findings from Billings (1979), Margolin and Wampold (1981), and Gray-Little and Burks (1983) suggest that distressed couples tend to use more coercive control techniques in the relationship than couples in a healthy relationship. Noller (1980) reported distressed couples tend to make more misinterpretations of basic communications than nondistressed couples. A strong relationship between psychopathology and poor social adjustment has been found to relate to poor marital adjustment (Vaillant, 1978). Additionally, Markman

Reproduced with permission of the copyright owner. Further reproduction prohibited without permission.

(1981) identified unrewarding communication patterns during the premarital period as predictive of low levels of marital satisfaction in a five-year longitudinal study. Furthermore, findings of Bentler and Newcomb (1978) identify impulse control and conscientiousness of the husband as predictive of marital happiness.

Eysenck (1980) interviewed and administered the Eysenck Personality Questionnaire to 1,500 married or divorced men and women. He found extraversion to be related to divorce among men. Eysenck suggests the phenomenon may be explained by the tendency for extraverted men to be more promiscuous. The comparison was made between 35 male divorced spouses and 326 male married spouses. Even though the comparison was significant (p > .05), a replication of the study would be advisable. He further observed that divorces were more frequent among the psychiatrically abnormal (high psychoticism, high neuroticism), especially among the women.

Personality variables have been identified which relate to marital satisfaction. Pickford, Signori, and Rempel (1966) employed the Guilford-Zimmerman Temperament Survey in comparing three groups of married couples designated as happily married (Group A), having trouble (Group B), and on the verge of separation (Group C).

Higher scores on the objectivity, friendliness, and personal relations traits were associated with marital happiness in men, but the trait of emotional stability was more important for insuring marital happiness in women than in men.

#### Summary

Based upon the literature reviewed, there appears to be a need to expand upon previous studies which have attempted to discover a relationship between moral judgment and marital satisfaction and to compare the relationship with other measures of personality which have been shown to relate to marital satisfaction. The literature is quite clear, however, that there are certain variables which are influential, including stage of family life cycle, religion, and education. Burr (1977) and others demonstrated the influence of the stage of family life cycle on marital satisfaction.

The literature is not consistent regarding the influence of different religious denominations on the marital relationship. Bahr and Chadwick (1985) discovered no differences in their study. However, Heaton (1984) observed differences in marital satisfaction among different religious denominations. There is general agreement, however, that moral judgment is different among different religions (Ernsberger & Manaster, 1981).

Moral judgment is highly associated with education (Rest, 1988). For this reason, the present study examines the relationship between moral judgment and marital satisfaction within defined ranges of educational achievement. Without restricting educational variance, a false positive relationship may be observed.

The most influential personality variables which appear to relate to marital satisfaction include, but are not limited to, neuroticism for both spouses and impulse control for men (Kelly & Conley, 1987). These findings appear to be well accepted and cross-verified by other studies. Such findings, of course, have greatly influenced therapeutic intervention. For this reason, it is necessary to include a measure of personality within the present study to compare with moral judgment in relating to marital satisfaction.

#### CHAPTER III

#### METHODOLOGY

#### Type of Research

This research used a correlational design to explore relationships among marital satisfaction, moral judgment, and three personality factors (internality, norm-favoring, and self-realization) and an ex post facto design to study differences on these variables with respect to the stage of family life cycle, religion, and educational achievement.

#### Population and Sample

The population under study included white, American, married couples whose religious orientation included Church of Jesus Christ of Latter Day Saints, Seventh-day Adventists, and United Methodists. The marriage at the time of data collection was the first marriage for both spouses if children were involved. Previously married individuals were included in the study providing the subject did not procreate prior to the present marriage. The purpose was to avoid data

Reproduced with permission of the copyright owner. Further reproduction prohibited without permission.

contamination which might occur from unique issues relating to stepchildren.

The sample included married couples representing designated stages of family life cycle and educational achievement. Sources of respondents included local church groups within communities in Washington, Utah, and Oklahoma. Educational achievement was controlled at four levels: (1) attended and/or graduated from high school, (2) attended college (but did not graduate), (3) graduated from college, (4) completed graduate education (i.e., M.A., Ph.D., M.D., etc.). Due to the extensive testing involving both spouses, random sampling was not considered an economical option. A purposive sampling procedure was applied.

Hypotheses requiring multiple-regression analysis apply to respondents as individuals (not couples). Using Cohen's formula (Cohen, 1977, p. 439) for defining sample size in multiple-regression analysis, 122 subjects were required for a power of .80 with five variables at a projected alpha level of .05 when the researcher seeks to explain 10% or more of the population variance. Even though only four independent variables are involved, power was estimated by including the control variables which were introduced into the data analysis.

Reproduced with permission of the copyright owner. Further reproduction prohibited without permission.

As regression analysis depends upon the correlation matrix, it is essential that this matrix be considered stable. For this purpose, Kendall (1975, p. 11) recommends a minimum of ten persons per variable. As the sample was divided into four family life cycle stages with four variables involved, a total of 4 X 40 subjects (160) were needed. Hypotheses requiring analysis of variance apply to respondents as individuals and as married couples. The largest number of cells in the two-way analysis of variance is 12 (a = 3, b = 4). Using Winer's (1962, p. 220) formula for defining power and assuming equal cell sizes of ten married couples per cell, power was estimated to be greater than .99 when alpha equals .05. Cell sizes were not equal. Thus power was reduced somewhat from the stated value. In considering both types of analysis, it was apparent that a sample size of 80 married couples would be appropriate.

Since it was not feasible to sample all levels of all combinations of the control variables (religion, family life cycle, & educational achievement), sample size was based upon designated stages of family life cycle.

#### Instrumentation

Three instruments were used to measure marital satisfaction, moral judgment, and personality. Marital satisfaction was measured by the Dyadic Adjustment Scale (DAS) (Spanier, 1976). Moral judgment was measured by the Defining Issues Test (DIT) (Rest, 1979). Personality was measured by the California Psychological Inventory (CPI)(Gough, 1987). Three personality factors were defined in the CPI and were used in this study. These factors include internality, norm-favoring, and self-realization. Each of these instruments is described in detail below.

#### Dyadic Adjustment Scale

#### **Description**

The Dyadic Adjustment Scale is a well known and highly reliable measure of marital adjustment (Cronbach's Coefficient Alpha is .96) (Spanier, 1976). Thirty-two objective concepts relating to dyadic adjustment are contained in the instrument. Scores for the Dyadic Adjustment Scale (DAS) range from 0-151. The scale is two pages in length and can be completed by most people in only a few minutes.

The instrument is designed in the form of a Likert scale containing six ordinal categories which include: "Always Agree," "Almost Always Agree,"

"Occasionally Disagree," "Frequently Disagree," "Almost Always Disagree," and "Always Disagree" (Spanier, 1976).

#### Development

In developing the scale, the researchers identified items used in previously developed scales measuring marital adjustment or a related concept. Their search yielded a pool of approximately 300 items (Spanier, 1976). All duplicate items were eliminated from the pool of items. Three judges other than the principal investigator examined the items for content validity.

The remaining 200 items contained several new items which the researcher chose to include. The questionnaire was administered to a purposive sample of 218 married persons in central Pennsylvania. The sample consisted primarily of middle-class residents who worked for one of four industrial or corporate firms which agreed to participate in the study. In addition, questionnaires were mailed to every person in Centre County, Pennsylvania, who had obtained a divorce decree during the 12 months preceding the mailing. Four hundred divorced persons were located. Ninety-four responded.

The researcher analyzed frequency distributions for each item and eliminated all items with low variance

or high skewness. Questions which appeared to contain alternative wording, structure, or category were examined in greater detail. The items with lesser variation were eliminated.

Spanier then analyzed the remaining variables using a t-test for significance of difference between means of the married and divorced samples, as described above. Items with the lowest t-value were eliminated. Forty items remained.

The remaining 40 items were then factor analyzed. This resulted in the elimination of 8 additional items which had low factor loadings, leaving 32 items.

Content validity was established by evaluation of the instrument by three judges. Items were included which were (1) relevant measures of dyadic adjustment for contemporary relationships, (2) consistent with the nominal definitions established by Spanier (1976) for adjustment and three factors including satisfaction, cohesion, and consensus, and (3) accurately worded with proper fixed-choice responses. Some of the items included in the instrument are found in previously developed scales. Other items have been modified from other scales, and others were developed specifically for this scale (Spanier, 1976).

Even though Spanier identified four oblique subscales, Sharpley and Cross (1982) were unable to

replicate the findings by repeating the factor analysis on a new sample. For this reason, the subscales were not applied in the present study. Total scale internal reliability was .96, exactly replicating Spanier's findings.

The correlation between the Locke-Wallace Marital Adjustment Test and the DAS was .86 among married respondents and .88 among divorced respondents (p < .001). The instrument was administered to a sample including 218 married persons and 94 recently divorced persons. Each of the 32 items correlated with the respondent's marital status. Each item differed beyond the .001 level of significance using a t-test for assessing differences between sample means. The mean total scale score was 114.8 for the married group and 70.7 for the divorced group.

Subsequent factor analysis was applied to define four interrelated scales including dyadic satisfaction, dyadic cohesion, dyadic consensus, and affectional expression. These four scales were not used in the present study due to a lack of replicability (Sharpley & Cross, 1982).

# Scoring

The respondent is asked to mark the appropriate box "to the approximate extent of agreement or

disagreement between you and your partner for each item." Values are assigned in a directional manner from 0-5. A respondent's total score is the sum of the values.

Defining Issues Test

#### Description

The DIT was developed by James R. Rest in 1974 based on Kohlberg's theory of moral judgment. It contains six stories portraying moral dilemmas. A list of 12 questions or statements relating to the dilemma follows each story. The P score is the sum of the weighted ranks given to Stage 5A, 5B, and six items. It is the most widely used index of the DIT.

The M score is a measure of sophisticated but meaningless items. It does not represent any stage of thinking but measures the subject's tendency to subscribe to statements for their pompousness rather than meaning. Rest (1986a) suggests discarding a protocol with an M score of 8 or greater.

He also described the D score as an overall index of moral judgment which uses information from all stages rather than from only Stage 5 and 6 items. If the subject gave high ratings to the high stage items, his score was higher than if he gave high ratings to low stage items. In some studies the D score has exhibited superior performance to the P index. Other recent studies, however, have not demonstrated consistent superiority to the P index. For this reason, Rest recommends the use of the P and D indexes. The correlation between the P and D scales was .78 on a sample of 1,080 subjects.

The instrument is designed to be used by individuals or groups who are capable of reading at least as well as an 11-year-old child (Rest, 1986). Because of the heavy dependence upon reading skills, its use is limited among minority groups and subjects whose first language is not English.

# Development

McCrae (1985) described Rest's Defining Issues Test (DIT) as a "rare example of test construction at its best." In developing the Defining Issues Test, Rest abided by the following guidelines:

1. He did not rely on a single test instance. He presumed that the more information there was available, the greater the chance that extraneous factors would cancel each other out, leaving greater accuracy.

 He provided diversity of the test stimulus.
 This was to provide a representative sampling of possible circumstances.

3. He wanted to standardize test stimuli, test administration and setting, and scoring procedures as much as possible.

#### Scoring

The respondent is asked to check one of the spaces to the left of each statement which best describes the importance of the statement as it applies to the given dilemma. To the left of each statement are five choices which are designed in a Likert-type scale (Mueller, 1986). The choices include: Great, Much, Some, Little, and No. In this way, the respondent rates each question or statement according to its level of importance.

The respondent is then asked to identify the four most important questions or statements starting from most important to fourth most important. The principled morality score (P score) is derived from these four choices.

At the present time, three methods of scoring the DIT are available. First, scoring may be done without use of mechanical or electronic devices simply by following instructions in the manual (Rest, 1986a, p. 3.1). Hand scoring the DIT takes approximately 6 to 10 minutes per subject and requires no special skills.

Secondly, scoring may be done by employing one of the available computer programs listed in the Manual for the Defining Issues Test, Third Edition. The third method of scoring the DIT is to send the data to James R. Rest at the Center for the Study of Ethical Development, University of Minnesota where the data may be scored by computer.

California Psychological Inventory Description

The California Psychological Inventory (CPI) is a personality test which was developed in its original form in 1957 (Groth-Marnat, 1984, p. 309). The original test did not focus primarily upon psychometric elegance. In its original form, the lack of a strong theoretical basis drew a considerable amount of criticism. Another criticism of the CPI was that the scales were intercorrelated. Baucom (1985), in reviewing the instrument, emphasized that the original purpose was not to provide an orthogonal factor analytic assessment instrument, but to assess 18 "folk" concepts in normal individuals.

In response to those who expressed concern over the lack of construct validity of the CPI, Gough revised the instrument. The revised CPI now consists of 20 folk scales and three orthogonal personality vectors. The

present study did not apply the 20 folk scales. Instead, the three personality vectors were used to correlate personality with marital satisfaction. The three vectors have been identified as internality, norm-favoring, and self-realization. The test has been administered under a number of conditions, ranging from formal testing conditions to "take-home" testing. Internal indicators of reliability and dependability showed satisfactory results in every condition.

## Development

The CPI was originally developed to assess enduring personality characteristics within a normal population. Gough published his original scales in 1948. However, it was not until 1957 that the full 18 scales were published by Consulting Psychologists Press. Many of the items were taken from the Minnesota Multiphasic Personality Inventory. Eysenck (1985) and others have criticized Gough for not factor analyzing the CPI to provide construct validity and demonstrate a unique personality model.

In constructing the revised instrument, Gough (1987), through factor analysis, sought items which correlated significantly for both sexes with one of the criteria, but at or near zero with the two other criteria. He reported that this process was not

difficult for the first two criteria, but it was much more difficult to intercorrelate items for the third criterion in an effort to be orthogonal to the first two. In the end, he did develop two orthogonal personality scales. The third scale correlated -.17 with the first scale and -.18 with the second scale (Gough, 1987, p. 16). In vector 1, higher scores related to an internal orientation or introversion and lower scores related to an external orientation or extraversion. High scores on vector 2 related to normfavoring or norm-following and lower scores related to norm-doubting or norm-denying. High scores on vector 3 related to a strong sense of self-realization and fulfillment.

The CPI was chosen for the present study for three reasons. (1) The three unrelated personality factors offer greater clarity in observing relationships between variables than other instruments. (2) Reliability data are exceptionally impressive in comparison to other personality measures. (3) A contribution may be made to the literature in the employment of the CPI as a newly revised instrument.

## Scoring

The CPI is a self-administered, 462-item test printed in a reusable booklet. The answers are recorded

by the respondent on a sheet which can be hand scored with stencils by the researcher or returned to Consulting Psychological Press for processing. Internal reliability data for the CPI are reported in Table 1.

#### Table 1

# Reliability Data for CPI Vectors

College Students			High School Students					
Alpha Coefficients			Parallel Test-retest					
Scale	No. of	200	200	400	85	38	102	128
	e Items	Male	Female	Total	Male	Female	Male	Female
v.1	34	.81	.77	.81	.81	.78	.69	.76
v.2	36	.78	.81	.78	.67	.60	.61	.75
v.3	58	.85	.83	.85	.78	.74	.75	.71

Gough (1987, p. 32)

## Procedures

Data collection procedures included soliciting volunteers for testing from local churches and university classes. Couples who had had previous marriages participated if no children had been produced from the previous marriage. Ethical practices were maintained as legislated by Principle 9, Ethical Principles of Psychologists (American Psychological Association, 1981). Principle 9 confirms the psychologist's professional

Reproduced with permission of the copyright owner. Further reproduction prohibited without permission.

responsibility to carry out investigation with respect and concern for the dignity and welfare of those who take part as research subjects.

Testing included administration of the CPI, DIT, DAS, and a short questionnaire which addressed the demographic variables. The researcher invited subjects to participate who were affiliated with the United Methodist, Latter Day Saints, or Seventh-day Adventist churches by mailing a letter (Appendix A) to the stratified purposive sample. Subjects who were members of the Latter Day Saints Church (LDS) church were selected from the Orem/Provo, Utah, area, since the LDS religion is the predominant religion of that area. Seventh-day Adventists (SDA) were sampled from the Walla Walla Valley in southeastern Washington and northeastern Oregon, since the concentration of SDAs is relatively high in that area.

United Methodists are included in this study primarily because they represent a moderate Christian church (Rest, 1979, p. 116). Methodists were sampled from the greater Oklahoma City area and the southern portion of Idaho.

A letter was sent to prospective subjects requesting their cooperation. If no response was received, a local research assistant followed the letter with a telephone call approximately one week later. The research assistant made the first follow-up telephone call approximately two

weeks after the materials were delivered to the subjects. The purpose of the telephone call was to encourage subjects to complete the instruments as soon as possible and return the packet. The research assistant made two additional follow up telephone calls approximately 10 days apart to encourage subjects to complete and return the tests. When the subjects indicated to the research assistant that they had returned the test packets, the assistant discontinued any additional follow-up telephone calls. Subjects completed the questionnaires in the privacy of their homes. No time limit was imposed during the testing period, however, subjects were encouraged to respond to questions quickly to avoid fatigue.

Subjects responded anonymously. Identification numbers were assigned to test packets for the purpose of identifying paired respondents as married couples. Anonymity was preserved. A respondent who wished to know the results of the tests completed the Request for Test Results form (Appendix C). The respondent was the only person who could complete the form accurately, since no one else, including the research staff, knew the identification number for any respondent. By completing the Request for Test Results form, the respondent knew that the tests were no longer anonymous. Test results were given to the respondent confidentially by the

researcher. After the report, all identifying data were destroyed.

Upon completion of the tests, the respondents returned the instruments to the original envelopes and mailed the packet to the researcher or returned it to the research assistant.

No individual test results were disclosed to anyone without a formal, written request from the respondent naming the individual who was to receive the test results, title, address, and telephone number. Additionally, the Request for Test Results form was completed and signed by the respondent in order to identify the tests.

The DIT was computer scored by using a program developed by Lonnie Millett, Student Life Research Center, Brigham Young University, Provo, Utah (Rest, 1986a, p. 3.7). The CPI was hand scored by templates puchased from Consulting Psychologists Press (Gough, 1987). The DAS was hand scored according to instructions provided by Spanier (1976).

# Null Hypotheses and Statistical Analysis

The research questions were answered by testing and interpreting findings for the following hypotheses:

# Research Question 1

Is there a relationship between moral judgment, personality, and marital satisfaction?

# Hypothesis 1

There is no correlation between moral judgment and marital satisfaction.

## Hypothesis 2

There is no correlation between moral judgment and marital satisfaction when the effect of family life cycle, educational achievement, and religious affiliation are controlled.

#### <u>Hypothesis 3</u>

Moral judgment and personality are not related to marital satisfaction.

#### <u>Hypothesis 4</u>

Moral judgment and personality are not related to marital satisfaction when the effect of family life cycle, educational achievement, and religious affiliation are controlled.

Hypotheses 1 through 4 were analyzed by multipleregression analysis and stepwise-regression analysis.

#### <u>Hypothesis 5</u>

Differences in moral judgment do not account for differences in marital satisfaction when family life cycle, educational achievement, and religion are controlled individually.

#### Hypothesis 6

Differences in moral judgment and internality do not account for differences in marital satisfaction when family life cycle, educational achievement, and religion are controlled individually.

## <u>Hypothesis 7</u>

Differences in moral judgment and norm-favoring do not account for differences in marital satisfaction when family life cycle, educational achievement, and religion are controlled individually.

## Hypothesis 8

Differences in moral judgment and self-realization do not account for differences in marital satisfaction when family life cycle, educational achievement, and religion are controlled individually. Hypotheses 5 through 8 were tested by one-way and two-way analysis of variance (Dixon, 1983, p. 359).

# Research Question 2

Do differences between spouses in moral judgment and personality explain differences in marital satisfaction?

# Hypothesis 9

Differences between husbands and wives in moral judgment do not account for differences in marital

satisfaction for husbands when family life cycle, educational achievement, and religion are controlled individually.

#### Hypothesis 10

Internality and differences between husbands and wives in moral judgment do not account for differences in marital satisfaction for husbands when stage of family life cycle, educational achievement, and religion are controlled individually.

#### Hypothesis 11

Norm-favoring and differences between husbands and wives in moral judgment do not account for differences in marital satisfaction for husbands when stage of family life cycle, educational achievement, and religion are controlled individually.

# Hypothesis 12

Self-realization and differences between husbands and wives in moral judgment do not account for differences in marital satisfaction for husbands when family life cycle, educational achievement, and religion are controlled individually.

# Hypothesis 13

Differences between husbands and wives in moral judgment do not account for differences in marital

Reproduced with permission of the copyright owner. Further reproduction prohibited without permission.

satisfaction for wives when stage of family life cycles, educational achievement, and religion are controlled individually.

#### Hypothesis 14

Internality and differences between husbands and wives in moral judgment do not account for differences in marital satisfaction for wives when family life cycle, educational achievement, and religion are controlled individually.

# Hypothesis 15

Norm-favoring and differences between husbands and wives in moral judgment do not account for differences in marital satisfaction for wives when family life cycle, educational achievement, and religion are controlled individually.

# Hypothesis 16

Self-realization and differences between husbands and wives in moral judgment do not account for differences in marital satisfaction for wives when family life cycle, educational achievement, and religion are controlled individually.

Hypotheses 9 and 13 were tested by one-way analysis of variance. Hypotheses 10 through 12 and 14 through 16 were tested by two-way analysis of variance. Moral judgment was analyzed at three levels including (1) husband higher than wife, (2) no difference between husband and wife, and (3) husband lower than wife. Moral judgment differences between spouses which were found between the 40th and 60th percentile rank were defined as "no difference."

In each of the analyses, three levels of moral judgment and two levels of each of the personality variables were observed. For moral judgment differences, the wife's score was subtracted from the husband's score. If the two scores were equal, the difference was zero. If the husband's score was higher, the difference was a positive number. If the wife's score was higher, the difference was a negative number. If the median difference between the husbands' and wives' scores was zero, then the 50th percentile rank represented a score which communicated "no difference" between a husband and wife. By arbitrarily defining a confidence interval of 10 points, scores below the 40th percentile rank were identified as the wife having a higher moral judgment score than the husband. Scores above the 60th percentile rank were identified as the husband having a higher moral judgment score than the wife. Personality scores above the median were identified as high scores and scores below the median were considered low scores.

Reproduced with permission of the copyright owner. Further reproduction prohibited without permission.

Stage of family life cycle was controlled and delimited only to the following four stages: (1) a married couple who has never had children, (2) a married couple who has children, none of whom are in their teens, (3) a married couple who has children with one or more who is in the teen years, and (4) a married couple who has children who are grown and no longer live at home.

Religion was controlled at three levels. These included Church of Jesus Christ of Latter Day Saints, Seventh-day Adventists, and United Methodists.

Educational achievement was controlled at four levels: (1) attended and/or graduated from high school, (2) attended college (but did not graduate), (3) graduated from college, (4) completed graduate education (i.e., M.A., Ph.D., M.D.).

While only main-effects hypotheses were stated, several interaction hypotheses were tested. Any significant interaction led to a series of simple effects tests.

The analyses which were statistically significant and had more than two cells received the Newman-Keuls test of significance if cell sizes were relatively equal. If the cells were not equal, the Scheffé test of significance was applied.

## CHAPTER IV

#### FINDINGS

This chapter reports the research findings and is divided into four major sections. The first section includes a description of the population, basic data on the variables, and some additional findings. The second section includes analyses of the correlations among the variables and discusses the tests for the first four hypotheses, partially answering the first research question. In the third major section, the remaining 12 hypotheses are tested, addressing the remainder of the first research question and answering the second question. Research findings are summarized in the fourth section. The analyses presented in this chapter are based upon the data set found in Appendix D.

# Description of the Population

The population for this study consisted of all married individuals who were Seventh-day Adventists (SDA) in the Walla Walla Valley (consisting of Walla Walla County, Washington, and Umatilla County, Oregon), Church of Jesus Christ of Latter Day Saints (LDS) in Utah County, Utah, and Methodists in Oklahoma City,

80

Reproduced with permission of the copyright owner. Further reproduction prohibited without permission.

Oklahoma, and Kimberly, Idaho. Ministers and bishops who supported this research were told that religion was considered to be a control variable. It was neither the interest nor intention of this research to compare religious denominations. For this reason, setting bias was deliberately incorporated into the research design.

There were 132 married couples in this purposive sample. All subjects were Caucasian, non-Hispanic. None had children from previous marriages. All were citizens of the United States. All subjects were members of the same denomination as their spouse.

Subjects reported their stage of family life cycle by marking the stage which best described their own from among seven stages. These included (1) no children, (2) expecting the first child, (3) have at least one infant or preschool child, but none over six years of age, (4) have at least one child over six years of age, but no teenagers, (5) have at least one teenager, (6) at least one child no longer lives at home, but not all children have left, and (7) all children have left home. Each stage of family life cycle was assigned a numerical value. All seven stages were represented in the sample. The frequency distribution for stages of family life cycle is presented in Table 2. Religious orientation of the subjects is reported in Table 3.

Reproduced with permission of the copyright owner. Further reproduction prohibited without permission.

Subjects were also asked to report their highest level of academic achievement. The frequency distribution for highest level of educational achievement is presented in Table 4. The sample in this study presented higher educational levels than the national average for the white population. According to a 1986 survey (U. S. Bureau of the Census, 1987), 20.1% of the white, U.S. population completed a four-year college degree (Table 5). However, more than half (51.5%) of the subjects in this study had completed a four-year college degree or an advanced degree.

Group means for marital satisfaction, moral judgment and personality are presented in Table 6. Marital satisfaction, as measured by the Dyadic Adjustment Scale (DAS) (Spanier, 1976) yielded a mean of 113.64 with a standard deviation of 14.13 (Table 7). These results were similar to Spanier's (1976) results from a sample of 218 married subjects where a mean of 114.8 with a standard deviation of 17.8 were reported. Welkowitz, Ewen, and Cohen (1982, p. 219) present a type of Z score for comparing means. An effect size may be calculated by subtracting the means of the two groups and dividing by the standard deviation. By applying this method of calculating effect size for the purpose of comparing the means of the present study with the Spanier statistics for married subjects, no

Table 2

Frequency	<u>Distribution</u>	for	Stages	of	Family	Life	Cvcle

Stage	Frequency	Percent
No children	47	17.8
Expecting first child	6	2.3
No child over 6 years	52	19.7
Child over 6, no teens	49	18.6
At least one teen	36	13.6
At least one left home	33	12.5
All children left home	41	15.5
	264	100.0

# Table 3

Frequency Distribution for Religion

Frequency	Percent
84	32.2
128	48.1
52	19.7
264	100.0
	84 128 52

\*

Seventh-day Adventist Church of Jesus Christ of Latter Day Saints \*\*

\*\* Methodist

Reproduced with permission of the copyright owner. Further reproduction prohibited without permission.

# Table 4

Frequency Distribution for Educational Achievement

Education	Frequency	Percent	
Attended HS	34	12.9	
Attended College*	94	35.0	
Graduated College	93	35.2	
Graduate degree	43	_16.3	
-	<u>_43</u> 264	100.0	

\*Attended but did not graduate

# Table 5

Educational level of U.S. white population

	High school or less	Attended College	College
Total	62.8	17.1	20.1
Male	58.7	17.3	24.0
Female	66.7	16.9	16.4

(U.S. Bureau of the Census, 1987)

-

-----

# Table 6

# <u>Group Means for Marital Satisfaction, Moral Judgment,</u> and Personality

Variable*	Mean	S.D.	Min. Max		
DAS	113.636	14.128	61	144	
DIT, P index	21.318	7.403	0	46	
DIT, D index	25.359	7.238	2	44	
CPI, Internality	18.504	7.516	4	47	
CPI, Norm-favoring	25.098	4.136	12	34	
CPI, Self-realization	37.072	7.781	7	55	

\*DAS Dyadic Adjustment Scale DIT Defining Issues Test CPI California Psychological Inventory

# Table 7

# Dyadic Adjustment Scale results compared to Spanier's married group

	Mean	SD	n
Present research	113.64	14.13	264
Spanier results*	114.8	17.8	218

\* Spanier, 1976

significant difference was found between the two means. By applying the standard deviation for the present study, an effect size of .08 was found. When Spanier's standard deviation was applied, the effect size was .06. Welkowitz, Ewen, and Cohen (1982, p. 220) proposed that an effect size of .20 was small, .50 was medium, and .80 was large. Furthermore, as sample size increases, there is a tendency for scores to regress toward the mean, resulting in smaller variance. The sample size in the present study was larger than Spanier's sample of married subjects and the standard deviation was appropriately smaller. The median score for this sample was 114, suggesting a fairly symmetrical distribution when compared to the mean.

Moral judgment was measured by the Defining Issues Test (DIT). Seventh-day Adventists (SDA) and Mormons (LDS) are generally considered conservative denominations. These two religions constituted 80% of the sample. Rest (1986b, p. 178) observed that religion, when depicted as liberal versus conservative, is related to DIT scores. Liberal religious ideology is associated with higher DIT scores. More conservative religions tend to emphasize obedience to external authority and doctrines and DIT scores tend to be lower. Because setting bias was deliberately integrated in the research design, religions cannot be compared in this

study. However, Rest's general observations were upheld (Rest, 1979, p. 101). He reported that the P index portrays the sum of weighted ranks given to items considered to be principled moral responses. The D index is a measure of overall moral development. This sample obtained a mean P index for the DIT of 21.32 with a standard deviation of 7.4, while the D index was 25.36 with a standard deviation of 7.24. When comparing the P% means of this sample with Rest's (1979, p. 116) findings from several religions, these scores appear to be very low. Table 8 depicts a summary of DIT scores reported by Rest (1979, p. 116).

Table 8

## Moral Judgment and Religious Affiliation

	n	P%Mean	S.D.
Unitarian-Universalist	41	52.6	15.2
United Methodist	49	46.6	15.2
Missouri Synod Lutheran	38	34.9	12.5
Conservative Baptist	41	30.1	11.1

(Rest 1979, p. 116)

Three personality variables were measured by the California Psychological Inventory (Gough, 1987). Since the mean educational achievement level for this sample suggests that the average subject has attended college

or has completed a college degree, the most appropriate comparison group provided by Gough (1987) is the "college student" group, even though Gough's students are not demographically comparable to this sample. For the first personality variable (internality), the mean for this sample was 18.5 with a standard deviation of 7.52. Gough's (1987, p. 17) college-student group of 3,236 male students had a mean of 16.14 with a standard deviation of 5.93. His 4,126 female students had a mean of 17.91 with a standard deviation of 6.01. By subtracting the mean for this sample from the mean of Gough's male college students and using Gough's standard deviation, a moderate effect size of .40 was observed. This means that there is a moderate difference between this overall sample and Gough's male college students. When Gough's female college students were compared with this sample, an effect size of .098 was observed. This means there was no difference between the mean of these scores and Gough's sample of female college students.

The second personality variable was normfavoring. The mean for this sample was 25.10 and the standard deviation was 4.14. Comparing these results to Gough's (1987, p. 18) college student group, a sample of 3,236 males yielded a mean of 22.56 with a standard deviation of 5.82. His sample of 4,126 female college students yielded a mean of 23.86 with a standard

deviation of 5.22. By subtracting the mean for this sample from the mean of Gough's male college students and using Gough's standard deviation, a moderate effect size of .44 was observed. This means that there is a moderate difference between this overall sample and Gough's male college students. When Gough's female college students were compared with this sample, an effect size of .24 was observed. This means there is a small difference between the mean of these scores and Gough's sample of female college students.

The third personality variable was self-realization. The mean for this sample was 37.07 and the standard deviation was 7.78. Gough's (1987, p. 19) sample of 3,236 male college students yielded a mean of 38.88 with a standard deviation of 7.78. His sample of 4,126 female college students yielded a mean of 40.97 with standard deviation of 7.09. By subtracting the mean for this sample from the mean of Gough's male and female college students and using Gough's standard deviation, only small effect sizes were observed for each gender. This means there is a small difference between this overall sample and Gough's college students. There are no large differences between these personality scores and scores from Gough's research with subjects who had attained a similar educational level.

# Correlations Among the Variables

This research examines the relationships and differences between the dependent variable (marital satisfaction) and the independent variables, including moral judgment, internality, norm-favoring, and selfrealization. Certain demographic variables served as control variables. All coefficients were rounded to two decimal places. Table 9 is an intercorrelation matrix for the variables included in this study. It is assumed for the use of the Pearson coefficient of correlation that the two correlated variables have a linear relationship.

### Testing the Hypotheses

Each of the 16 null hypotheses stated in Chapter 3 was tested for statistical significance. The first four hypotheses were tested by using regression analysis and the last 12 hypotheses were tested by analysis of variance.

Hypotheses Tested By Regression Analysis

For the regression portion of the study two moral judgment scores were used: the P index (Rest, 1979, p. 100) and the D index (p. 236).

Reproduced with permission of the copyright owner. Further reproduction prohibited without permission.

## Hypothesis 1

There is no correlation between moral judgment and marital satisfaction.

As observed in the intercorrelation matrix (Table 9), there was no correlation between the Defining Issues Test (DIT) P score and the Dyadic Adjustment Scale. The Dyadic Adjustment Scale correlated .04 with the D score of the DIT and was not statistically significant. Therefore, the null hypothesis was retained.

# Additional Findings from The Intercorrelation Matrix

In addition to the results related to the formal hypotheses, additional relationships were observed in the intercorrelation matrix.

1. Moral judgment was negatively correlated to the number of years the subjects were married. The correlation for the P index of the Defining Issues Test was -.17 (p < .01). The correlation for the D index of the Defining Issues Test was -.25 (p < .001) with number of years married. Subjects who reported being married longer tended to score lower on the DIT.

 Educational achievement level and moral judgment were positively correlated. The D index of the DIT produced a correlation coefficient of .2136 (p < .001). The correlation for the P index of the DIT

#### Intercorrelation Matrix

	Sex	Years Married	Family Life Cyclo	Doligio-	Educational Achievement
	Jex	Marrieu	Cycle	Religion	Level
S	1.00				
YM	.01	1.00			
FLC	01	.90***	1.00		
Rel	.00	01	.01	1.00	
Ed	19***	05	02	.06	1.00
DAS	.02	02	05	02	.06
Ρ	.05	17**	12*	.08	.13*
D	.06	25***	15**	.12*	.21***
V1	.20***	.18**	.17**	24***	03
V2	01	.21***	.17**	03	.06
V3	.05	.00	.00	.07	.19

\* p < .05 \*\* p < .01 \*\*\* p < .001

Note. Abbreviations in the left column are for variable names across top. In addition, the following abbreviations are used: DAS Dyadic Adjustment Scale P the P index of the Defining Issues Test D the D index of the Defining Issues Test V1 Internality scale of the CPI V2 Norm-favoring scale of the CPI V3 Self-Realization scale of the CPI

Table 9, continued

	DAS	P	D	Vl	V2	V3
DAS	1.00	<u> </u>				
P	.00	1.00				
D	.04	.61***	1.00			
Vl	.01	17**	18***	1.00		
V2	.32**	*02	.00	04	1.00	
V3	.18	.37***	36***	34***	.06	

\*\*\* p < .01 \*\*\* p < .001 produced .1349 (p < .05), substantiating Rest's (1979, p. 110) findings that moral judgment increases with more education.

3. The highest correlation observed was between self-realization and moral judgment. Self-realization yielded a correlation coefficient of .3655 (p < .001) with the P index of the DIT. Self-realization correlated .3645 (p < .001) with the D index of the DIT. Since self-realization, as evaluated by the CPI (Gough, 1987), is a measure of maturity, this finding suggests that moral judgment increases as one becomes more mature.

4. Internality was negatively correlated with moral judgment, suggesting that subjects with higher moral judgment scores tended to be more extraverted.

#### Hypothesis 2

There is no correlation between moral judgment and marital satisfaction when the effect of family life cycle, educational achievement, and religion are controlled individually.

When family life cycle, educational achievement, religion, and the P index for the DIT were entered one at a time as independent variables into a stepwise multiple-regression model, no correlation was observed. The four variables accounted for less than 1% of variance. The standardized regression coefficient value for the P index of the DIT was -.01 and was not statistically significant. The correlations were not significant.

When family life cycle, educational achievement, religion, and the D index were introduced as independent variables, no correlation was observed (Table 10). The standardized regression coefficient value for the D index of the DIT was .026, and was not statistically significant. Therefore, the null hypothesis was retained.

Table 10

Summary for Correlation of DIT, P Index, Religion, Family Life Cycle, and Education to Dyadic Adjustment Scale

Variable	Standardized Regression Coefficient	Probability
Religion	024	.6945
Family Life Cycle	051	.4123
Educational Achievement Level	.058	.3537
Defining Issues Test, P Index	013	.8389

#### Hypothesis 3

Moral judgment and personality are not related to marital satisfaction.

Stepwise multiple regression was used for this analysis. Using the established .05 level of significance for entry into the regression model, normfavoring and self-realization accounted for 12.5% of the variance of marital satisfaction, where p < .001. Normfavoring was the first personality variable to enter the model, accounting for 10% of the common variance in marital satisfaction at the .001 level of significance. The standard partial regression coefficient for selfrealization was .16 at the .01 level of statistical significance, explaining an additional 2.6% of the common variance with marital satisfaction. The P index of the DIT and the internality scale of the personality inventory did not enter the regression model. When the same analysis was done using the D index of the DIT in lieu of the P index, the same results were observed. Because of the strong influence of the norm-favoring personality variable and the smaller contribution of the self-realization variable, the null hypothesis was rejected.

#### Hypothesis 4

Moral judgment and personality are not related to marital satisfaction when the effect of family life cycle, educational achievement, and religion are controlled individually.

Family life cycle, educational achievement, and religion were entered into the stepwise-multiple regression analysis before any predictor variables. These three control variables together did not account for any variance in marital satisfaction. After forcing the control variables into the model using stepwise multiple-regression analysis, the norm-favoring personality variable contributed a correlation coefficient of .33 and accounted for 11% of the common variance with marital satisfaction. This personality variable was statistically significant at the .001 level. Self-realization entered next, explaining an additional 2.89% of marital satisfaction variance. With these two variables in the regression model (normfavoring and self-realization), 14% of the variance for marital satisfaction was explained at the .001 level of statistical significance. Internality and the D index of the DIT did not enter the regression model at the .05 level of significance. Using the same stepwise approach, the same results were obtained when the P index for the DIT was used in place of the D index.

Moral judgment, as measured by the DIT, did not account for marital satisfaction in this study. However, since the two personality variables, norm-favoring and selfrealization, accounted for 14% of the variance of marital satisfaction, the null hypothesis was rejected, even though family life cycle, educational achievement, and religion did not influence the results.

Hypotheses Tested by Analysis of Variance

Since the literature (Burr, 1970) suggests a curvilinear relationship between family life cycle and marital satisfaction, it is conceivable that the above regression analyses portray an incomplete explanation of the common variance between the predictor variables (personality and moral judgment) and the dependent variable (marital satisfaction). Regression analysis assumes a linear relationship. For this reason, further analysis was undertaken to view the variables from another perspective. The following analyses incorporated analysis of variance.

In the following analyses which involve moral judgment, only the D index of the DIT was used. Scores on the D index were used to divide the subjects into three equal groups. The first group included subjects who scored low on the DIT (2-22). The second group was considered to be within the average range for this

sample (23-28). The third group scored highest within this sample (29-44). Each of the three DIT cells contained 88 subjects.

The seven stages of family life cycle were categorized into four groups. The first group included those who did not have children, even though six of the subjects (three couples) were expecting their first child at the time of data collection. The second group was identified as subjects who had children, but no teenagers. The third group was identified as subjects who had at least one teenager in the home. The fourth group contained subjects whose children were launched.

In several of the following analyses which include more than one independent variable, interaction was observed between the variables. This means that the influence of one independent variable upon the dependent variable depends on the level of another independent variable (Kerlinger, 1986, p. 230). Interacting independent variables received further analysis.

In the following hypotheses, stage of family life cycle, level of educational achievement, and religion are presented as control variables. Hence, all analyses were done at each separate level of education, each separate stage of family life cycle, and each religion.

#### Hypothesis 5

Differences in moral judgment do not account for differences in marital satisfaction when family life cycle, educational achievement, and religion are controlled individually. This required 11 one-way analyses of variance. Moral judgment was the independent variable. Table 11 shows the mean DAS score for each level of moral judgment. The results of the ANOVA are given in Table 12.

The first analyses were conducted at each stage of family life cycle. Differences in moral judgment did not account for differences in marital satisfaction when controlled by four stages of family life cycle.

The second analyses controlled for education. The first level of educational achievement was significant for differences in moral judgment in accounting for differences in marital satisfaction. In this case, those who scored low in moral judgment scored high in marital satisfaction. Statistical significance was confirmed at a level of .001 for the first level of educational achievement only. The Scheffé method for testing contrasts between means was applied to the means for moral judgment. Significant differences were observed between the high and low groups and between the middle and low groups. The difference between the middle and high groups, however, was not significant.

Marital Satisfaction Means for Various Levels of Moral Judgment

at Each Level of Control Variables

		4	Moral Jud	Judgment		
Control Variable Level	Low		Middle	le	High	
No children	121.33	(6)	109.56	(6)	116.125	(8)
Young children	109.37	(16)	112.68	(61)	113.06	(16)
Teenagers	115.45	(11)	108.33	(12)	114.82	(11)
Children launched	110.36	(11)	124.00	(9)	114.50	(4)
High school	121.37	(61)	100.75	(12)	92.67	(3)
Attended college	111.10	(08)	114.55	(33)	117.00	(31)
B.A./B.S.	111.43	(28)	111.25	(28)	114.76	(11)
Graduate degree	120.00	(11)	114.87	(12)	114.76	(11)
SDA	115.00	(12)	113.20	(12)	109.71	(7)
LDS	114.00	(11)	113.17	(23)	109.71	(23)
Methodist	108.11	(6)	108.75	(8)	109.89	(6)

<u>Analysis of Variance for Various Levels of Moral Judgment</u>

at Each Level of Control Variables

		Moral Judgment	
Control Variable Level	ſτą	đf	đ
No children	2.10	2,23	.15
Young children	0.27	2,48	.77
Teenagers	0.93	2,31	.41
Children launched	1.93	2,18	.17
High school	7.59	2,31	*100.
Attended college	1.40	2,91	.25
B.A./B.S.	0.78	2,90	.46
Graduate degree	0.62	2,40	.55
SDA	0.38	2,40	. 69
IDS	0.45	2,60	.64
Methodist	0.07	2,23	.93
	•		

The third analysis was conducted for each religious group. Differences in moral judgment did not account for differences in marital satisfaction when controlled for each religion. In this analysis, differences in moral judgment did not explain marital satisfaction. Because differences were found between low, middle, and high groups in moral judgment scores for the group in the first level of educational achievement, the null hypothesis was rejected.

## <u>Hypothesis</u> 6

Internality and differences in moral judgment do not account for differences in marital satisfaction when family life cycle, educational achievement, and religion are controlled individually.

This required 11 two-way analyses of variance. Moral judgment and internality were the independent variables. Table 13 gives the mean DAS score for each level of moral judgment and internality. Table 14 shows the results of the analysis of variance. The sample size for each subgroup is given in parentheses. The first four analyses were conducted at each stage of family life cycle. Differences in moral judgment and differences in internality did not account for differences in maritalsatisfaction when controlled by four stages of family life cycle. In this analysis,

neither moral judgment nor internality explained differences in marital satisfaction and no interaction was observed (Table 14).

The next four analyses controlled for education. The mean scores for marital satisfaction for the first level of educational achievement were significantly different at various levels of moral judgment. However, the means for marital satisfaction for those who scored above and below the mean for internality were not significantly different.

The third analysis was conducted for each religious group. Neither differences in moral judgment nor differences in internality were statistically significant when controlled for each religion.

Internality was not statistically significant within any of the control variables. A Scheffé test for contrast of means was used to determine which means were different among the moral judgment scores. It was determined that only the low and high cells were different at the .90 level of significance. The null hypothesis was rejected.

<u>Satisfaction Means for Various Levels of Moral Judgment by</u> lity at Each Level of Control Variables Internality at Each Level <u>Marital</u>

				Moral Judgment	gmen		
Control Variable Level	Internality	Low		Middle	٩	High	
No children	Above	113.50	(4)	106.33	(3)	128.00	(1)
Young children	Above	115.2	(15) (15)	111.1/	<u>0</u>	114.43 117.83	(18)
	Below	105.27	(15)	110.21	(29)	112.29	(21)
Teenagers	Above Relow	110.36	(11)	114 00	(17)	112.40	(10)
Children launched	Above	115.23	(17)	118.67	<u>)</u>	106.33	(3)
	Below	113.625	(8)	122.25	(4)	112.00	(e)
High school	Above	120.15	(13)	99.25	(4)	93.50	(2)
	Below	124	(9)	101.5	(8)	91.00	(1)
Attended college	Above	112.20	(12)	112.07	(15)	118.025	(8)
	Below	111.00	(12)	110.011	(18)	118.025	(8)
B.A./B.S.	Above	111.00	(13)	112.00	(10)	116.11	(61)
i	Below	114.44	(6)	110.83	(18)	112.89	(18)
Graduate degree	Above	112.60	(2)	114.625	(8)	117.67	(9)
	Below	117.83	(9)	115.14	(2)	113.18	(11)
SDA	Above	112.96	(27)	110.12	(11)	115.89	(6)
	Below	120.00	(13)	108.91	(53)	105.00	(2)
LDS	Above	116.93	(12)	111.94	(16)	116.33	(24)
	Below	115.31	(13)	113.76	(29)	119.33	(30)
Methodist	Above	112.90	(10)	113.00	(4)	106.50	(2)
	Below	108.70	(10)	110.45	(11)	108.40	(15)

<u>ANOVA for Various Levels of Moral Judgment by Internality at Each Level of</u>

**Control Variables** 

Control Variahle	Moral	Moral judgment	ent	Int	Internality	ĹτΥ	Inte	Interaction	
Level	Ĺμ	đf	đ	ы	đf	d	ы	đf	đ
No children	2.51	2,20	.11	0.10	1,20	.76	1.73	2,20	.20
Young children	1.26	2,95	.29	1.72	1.72 1,95	.19	0.88	2,95	.42
Teenagers	0.48	2,63	.62	3.25	1,63	• 08	0.30	2,63	.74
Children launched	1.38	1.38 2,35	.26	0.26	0.26 1,35	. 62	0.24	2,35	.79
High school	6.66	6.66 2,28	<.01*	0.02 1,28	1,28	. 89	0.04	2,28	.96
Attended college	1.41	1.41 2,88	.25	0.06	0.06 1,88	.81	0.04	2,28	.96
B.A./B.S.	0.54	2,37	.58	0.00 1,87	1,87	66.	0.73	2,87	.48
Graduate degree	0.55	2,37	.58	0.45 1,37	1,37	.51	0.17	2,37	.85
SDA	2.47	2,79	.09	0.29	0.29 2,79	.59	2.72	2,79	.07
LDS	1.18	1.18 2,121	.31	0.13	0.13 1,121	.72	0.20	2,121	.82
Methodist	0.40	2,46	.67	0.20 1,46	1,46	.66	0.23	2,46	.80
<pre>* Statistically significant</pre>	lifican	t							

106

# <u>Hypothesis 7</u>

Differences in moral judgment and normfavoring do not account for differences in marital satisfaction when family life cycle, educational achievement, and religion are controlled individually.

This required 11 two-way analyses of variance. Moral judgment and norm-favoring were the independent variables. Table 15 gives the DAS means. Table 16 shows the ANOVA results for all The first four analyses were conducted at tests. each stage of family life cycle. Differences in moral judgment and differences in norm-favoring did not account for differences in marital satisfaction at either of the first, third, or fourth stages of family life cycle. These stages included subjects who did not have children, subjects who had teenage children, and subjects whose children were grown and no longer lived at home. Neither moral judgment nor norm-favoring explained differences in marital satisfaction. No significant interaction was evident in any of these four analyses.

Subjects who scored above the mean in norm-favoring in the second stage of family life

Marital Satisfaction Means for Various Levels of Moral Judgment by Norm-

favoring at Each Level of Control Variables

				Moral Judgment	lgment		
Control Variable Level	Norm- favoring	LOW		Middle	e	High	-
No children	Above	114.14	(2)	116.43	(2)	116.36	(11)
Young children	Below Above	122.75 115.54	(12) (13)	110.50 116.33	(10)	<b>118.33</b> 122.42	(9) (19)
l	Below	106.46	(13)	105.19	(21)	107.65	(20)
Teenagers	Above Below	115.33	(9)	111.71	(14)	118.93 108.75	(14) (12)
Children launched	Above	117.06	(16)	120.86	(5)	107.33	(3)
	Below	110.56	(6)	86.57	(2)	81.00	(1)
High school	Above	123.50	(10)	120.60	(2)	98.50	(2)
	Below	00.011	(6)	86.57	(2)	81.00	(1)
Attended college	Above	115.08	(13)	117.75	(12)	122.335	(18)
	Below	108.06	(11)	112.71	(21)	109.625	(13)
B.A./B.S.	Above	112.27	(12)	113.06	(18)	119.30	(20)
	Below	110.46	(13)	108.00	(10)	108.94	(11)
Graduate degree	Above	112.75	(4)	114.875	(8)	115.43	(2)
	Below	124.14	(2)	114.86	(2)	114.30	(10)
SDA	Above	118.30	(20)	113.82	(11)	108.80	(10)
	Below	112.20	(20)	106.94	(11)	113.82	(11)
LDS	Above	115.64	(14)	117.40	(22)	122.00	(34)
	Below	116.71	(14)	107.75	(20)	111.20	(20)
Methodist	Above	110.125	(8)	111.86	(2)	108.33	(9)
	Below	111.25	(12)	110.50	(8)	108.09	(11)

<u>ANOVA for Various Levels of Moral Judgment by Norm-favoring at Each Level</u>

of Control Variables

Control Variahle	Moral	Moral judgment	ent	Norm-	Norm-favoring	ng	Inte	Interaction	
Level	伍	đf	đ	F	đf	đ	F	df	đ
No children	0.74	2,47	.48	0.19	1,47	. 66	1.45	2,47	.24
Young children	0.97	0.97 2,95	.38	15.44 1,95	1,95	*00*	0.32	2,95	.73
Teenagers	0.79	0.79 2,63	.46	2.53	2.53 1,63	.12	0.61	0.61 2,63	.55
Children launched	1.15	1.15 2,35	.33	0.13	0.13 1,35	.73	0.37	0.37 2,35	. 69
High school	10.30 2,28	2,28	•00•	8.37	8.37 1,28	*10.	4.32	4.32 2,28	.02*
Attended college	0.93	0.93 2,88	.48	8.64	8.64 1,88	*00*	0.68	0.68 2,88	.51
B.A./B.S.	0.81	0.81 2,87	.45	5.22 1,87	1,87	.02*	1.08	1.08 2,87	.34
Graduate degree	0.27 2,37	2,37	.45	0.63	0.63 1,37	.43	0.76	0.76 2,37	.48
SDA	1.25	1.25 2,79	. 29	3.64	3.64 1,79	.06	0.03	0.03 2,79	.97
LDS	0.94	2,121	.39	5.27	5.27 1,121 .02	. 02	1.54	1.54 2,121	.22
Methodist	0.36	0.36 2,46	.70	0.00	0.00 1,46	.96	0.06	0.06 2,46	.94

Statistically significant

\*

17	
Table	

<u>ANOVA for Subjects at the High School Level at Various Levels of Moral</u> <u>Judgment Differences and Above and Below the Mean for Norm-favoring</u>

Effect	E4	đf	ď
Moral judgment differences at low norm-favoring	3.28	2,14	3.28 2,14 .0677
Moral judgment differences at high norm-favoring	11.37	2,14	11.37 2,14 .0012*
Norm-favoring at low DIT	1.02	1,17	1.02 1,17 .3261
Norm-favoring at mid DIT	9.92	9.92 1,10	.0103*
Norm-favoring at high DIT	1.81	1,1	1.81 1,1 .4065

Statistically significant

\*

cycle scored higher in marital satisfaction than those who scored below the mean in norm-favoring. The differences were significant at the .001 level. The next four analyses controlled education at four levels. Because significant interaction between the means for moral judgment and norm-favoring was present at the high school level, it was necessary to test each level of norm-favoring by each level of moral judgment in addition to testing each level of moral judgment by each level of norm-favoring. The means were tested for statistical significance. Table 17 shows the results of the simple-effects tests. By applying the Scheffé test for contrasting means to the moral judgment means for the group which scored below the mean in norm-favoring, differences were present between the first and second mean and the first and third mean. No significant difference was observed between the second and third means.

Subjects who scored above the mean in norm-favoring in the second level of educational achievement scored higher in marital satisfactionthan those who scored below the mean in norm-favoring. The means were tested for significance.

Subjects within the middle group for moral judgment who scored above the mean for norm-favoring, scored

higher in marital satisfaction than those who scored below the mean for norm-favoring (Table 17).

Subjects had attended college and who scored above the mean on norm-favoring scored higher in marital satisfaction than those who scored below the mean in norm-favoring. Subjects who had attained B.A. or B.S. degrees and who scored above the mean in norm-favoring scored higher in marital satisfaction than those who scored below the mean in norm-favoring. Differences in moral judgment and norm-favoring did not account for differences in marital satisfaction for subjects who had attained advanced academic degrees.

The third analysis was conducted for each religious group. Differences in moral judgment and norm-favoring did not account for differences in marital satisfaction. Then null hypothesis was rejected.

## Hypothesis 8

Differences in moral judgment and self-realization do not account for differences in marital satisfaction when family life cycle, educational achievement, and religion are controlled individually.

This required 11 two-way analyses of variance. Moral judgment and self-realization were the independent variables. Table 18 shows the mean DAS score for each level of moral judgment above and below the mean for

self-realization. Table 19 shows the results of the ANOVA. The first analyses were conducted at each stage of family life cycle. Differences in moral judgment did not account for differences in marital satisfaction when controlled by four stages of family life cycle.

The next four analyses controlled for education. The mean scores for marital satisfaction for highschool-level respondents were not significantly different above and below the mean for self-realization. However, the means for marital satisfaction were significantly different at various levels of moral judgment. By applying the Scheffé test for contrasting means to the moral judgment means, differences were present between the first and second mean and the first and third mean. The second and third means were not significantly different. Because moral judgment was significant at the first level of educational achievement, the null hypothesis was rejected.

The third analysis was conducted for each religious group. Differences in moral judgment and

Marital Satisfaction Means for Various Levels of Moral Judgment by Self-

realization at Each Level of Control Variables

			-	Moral Judgment	gment		
Control Variable Level	Self- realization	Гом		Middle	Ð	High	
No children	Above	129.43	(2)	110.00	(9)	119.10	(10)
	Below	113.83	(12)	114.00	(11)	114.14	( <i>L</i> )
Young children	Above	111.80	(2)	112.72	(18)	116.79	(28)
	Below	110.81	(21)	106.94	(18)	109.91	(11)
Teenagers	Above	116.43	(2)	112.71	(11)	115.47	(11)
	Below	112.55	(11)	104.50	(8)	111.89	(6)
Children launched	Above	113.75	(8)	121.33	(9)	113.50	(4)
	Below	115.18	(11)	118.25	(4)	100.50	(3)
High school	Above	126.50	(4)	111.75	(4)	106.00	(T)
	Below	120.00	(12)	95.25	(8)	86.00	(2)
Attended college	Above	114.89	(6)	114.67	(21)	118.615	(23)
	Below	109.48	(21)	114.	(12)	112.375	(8)
B.A./B.S.	Above	116.50	(8)	110.20	(01)	116.95	(22)
	Below	109.40	(20)	111.83	(18)	111.00	(12)
Graduate degree	Above	119.67	(9)	115.175	(12)	113.15	(13)
	Below	120.40	(2)	115.17	(12)	113.15	(13)
SDA	Above	120.00	(11)	108.16	(19)	112.375	(8)
	Below	113.45	(53)	112.78	(6)	109.33	(6)
LDS	Above	117.58	(12)	117.50	(20)	120.21	(38)
	Below	115.125	(16)	109.60	(22)	112.75	(16)
Methodist	Above	114.75	(4)	116.75	(8)	108.54	(13)
	Below	109.81	(16)	104.71	(2)	107.00	(4)

<u>ANOVA for Various Levels of Moral Judgment by Self-realization at Each</u>

Level of Control Variables

Control Variable	Moral	Moral judgment	ent	Self-	Self-realization	ation	Intei	Interaction	e
Level	٤ų	đf	đ	F	đf	d	ſ¥,	đf	ď
No children	2.48	2,47	60.	3.03	1,47	60.	2.59	2,47	60.
Young children	0.42	2,95	. 66	1.56	1,95	.22	0.20	0.20 2,95	.82
Teenagers	1.18	2,63	.31	16.0	0.91 1,63	.13	0.20	0.20 2,63	.82
Children launched	1.62	2,35	.21	0.91	0.91 1,35	.35	0.65	2,35	.53
High school	5.57	2,28	.01*	3.08	1,28	60.	0.35	2,28	.71
Attended college	0.37	0.37 2,88	. 69	1.62	1.62 1,88	.21	0.35	2,88	• 69
B.A./B.S.	0.45	2,87	.64	2.02	1,87	.16	1.01	2,87	.37
Graduate degree	0.44	2,37	. 65	0.18	0.18 1,37.	. 68	0.28	2,37	.76
SDA	1.87	2,79	.16	0.26	1,79	.61	1.22	2,79	.30
LDS	0.48	2,121	. 62	4.12	4.12 1,121	.04*	0.30	2,121	.74
Methodist	0.69	2,46	.51	3.94	1,46	.053	1.03	2,46	.36
* Statistically significant	lificar	t l							

self-realization did not account for differences in marital satisfaction when the sample included SDA and Methodists. LDS subjects who scored above the mean in self-realization, scored higher in marital satisfaction than LDS subjects who scored below the mean in self-realization. Differences in moral judgment were not statistically significant for LDS subjects. Because self-realization was significant at the .05 level, the null hypothesis was rejected.

In each of the following analyses, difference scores between husbands and wives in moral judgment were established by subtracting the wives' scores from the husbands' scores. The difference scores were divided into three equal groups.

# Hypothesis 9

Differences between husbands and wives in moral judgment do not account for differences in marital satisfaction for husbands when stage of family life cycle, educational achievement, and religion are controlled individually.

This required 11 one-way analyses of variance. Difference scores in moral judgment comprised the independent variable. Table 20 gives the mean DAS scores for each level of moral judgment. The results of the ANOVA are given in Table 21. The first analyses were conducted at each stage of family life cycle.

<u>Marital Satisfaction Means for Husbands at Various Levels of</u> <u>Moral Judgment Differences at Each Level of Control</u> Variables

		I	Moral Juc	Judgment		
Control Variable Level	Wife higher		Equal		Husband higher	
No children	118.22	(6)	114.63	(8)	114.00	(6)
Young children	107.29	(11)	116.50	(20)	110.43	(14)
Teenagers	115.45	(11)	108.67	(6)	113.21	(14)
Children launched	113.50	(9)	111.57	(7)	119.25	(8)
High school	114.67	(3)	116.22	(6)	102.75	(4)
Attended college	112.17	(12)	113.00	(11)	120.10	(10)
B.A./B.S.	111.65	(11)	107.20	(01)	110.00	(16)
Graduate degree	113.73	(11)	120.875	(8)	115.93	(12)
SDA	114.87	(12)	111.25	(12)	113.94	(16)
LDS	110.64	(22)	118.64	(22)	115.74	(61)
Methodist	113.67	(9)	(6) 106.10	(10)	(10) 108.90	(10)

<u>Analysis of Variance for Husbands at Various Levels of Moral</u>

Judgment at Each Level of Control Variables

	4	Moral Judgment	
Control Variable Level	Ĺ	đf	ሲ
No children	0.27	2,23	.77
Young children	1.71	2,48	.19
Teenagers	0.59	2,31	.56
Children launched	0.65	2,13	.54
High school	0.93	2,36	.41
Attended college	0.93	2,40	.66
B.A./B.S.	0.43	2,40	.66
Graduate degree	0.65	2,31	.53
SDA	0.23	2,40	.79
LDS	1.42	2,60	.25
Methodist	1.08	2,23	.36

Differences in moral judgment did not account for differences in marital satisfaction at any of the four stages of family life cycle, four stages of educational level, or three different religions. The null hypothesis was retained.

#### Hypothesis 10

Differences between husbands and wives in moral judgment and internality do not account for differences in marital satisfaction for husbands when family life cycle, educational achievement, and religion are controlled individually.

This required 11 two-way analyses of variance. Table 22 shows the mean DAS scores for each level difference scores between husbands and wives in moral judgment. The independent variables were differences between husbands and wives in moral judgment and internality scores for husbands. The results of the ANOVA are given in Table 23. The first analyses were conducted at each stage of family life cycle. When controlled for husbands who are fathers of teenage children, those who scored above the mean in internality scored lower in marital satisfaction than husbands who scored below the mean in internality.

<u>Marital Satisfaction Means for Husbands at Various Levels of Husband/Wife</u> Moral Judgment Differences by Introversion at Each Level of Control

S
ส
Iat
ar
2

				Moral Judgment	Igmen		
Control Variable Level	Introversion	Wife higher		Egual	4	Husband higher	
No children	Above Below	112.20 125.75	(5) (4)	111.00 115.14	<b>1</b>	114.50 113.86	(2)
Young children	Above Below	<b>110.375</b> 104.56	(8) (6)	114.67 116.82	(3)	114.80	(2)
Teenagers	Above Belcw	116.25 115.00	(4)	96.80 123.50	(2)	115.60	( <del>)</del>
Children launched	Above Below	119.00 102.50	(4)	110.00	0.00	115.80	(5)
High school	Above Below	115.00	500	108.20		114.00	253
Attended college	Above	115.20	<u>)</u>	106.75		115.335	600
B.A./B.S.	Above	112.20		92.500	(r) (r)	105.86	56
Graduate degree	Above Below	114.50	5.55	115.67	() () () () () () () () () () () () () (	122.90 122.00	(10)
SDÀ	Above Relow	115.56	(6)	109.625		117.43	
LDS	Above	112.00	66	105.20	(5) (5)	112.14	
Methodist	Above Below	112.33 115.00	(f) (f) (f) (f) (f) (f) (f) (f) (f) (f)	60.721 97.00 11.71	(1)	112.14 102.00 110.625	853

<u>ANOVA for Husbands at Various Levels of Husband/Wife Moral Judgment</u> <u>Differences by Internality at Each Level of Control Variables</u>

Control Variahle	Moral	judgment	ent	Inte	Internality	t y	Inte	Interaction	
Level	F	df	đ	Ы	df	đ	F	df	đ
No children	0.36	2,20	.70	0.73	1,20	.40	0.55	2,20	.59
Young children	06.0	2,45	.42	0.47	0.47 1,45	.50	0.28	2,45	.76
Teenagers	0.52	2,28	.60	6.14	6.14 1,28	.02*	3.04	3.04 2,28	.06
Children launched	06.0	2,15	.43	0.21	0.21 1,15	.66	1.16	2,15	.34
High school	0.30	2,10	.75	0.01	0.01 1,10	.93	0.76	2,10	.49
Attended college	0.70 2,33	2,33	.50	0.36	0.36 1,33	.55	0.66	2,33	.52
B.A./B.S.	1.65	2,37	.21	3.87	3.87 1,37	.06	1.72	2,37	.19
Graduate degree	0.42	2,28	.66	0.02 1,28	1,28	06.	0.96	2,28	.39
SDA	0.12	2,37	.89	0.05	0.05 1,37	.83	0.47	2,37	. 63
TDS	0.37	2,57	. 69	2.64	2.64 1,57	.11	1.79	2,57	.18
Methodist	1.64	2,20	.22	1.85	1.85 1,20	.19	0.20	0.20 2,20	.82

Statistically significant

\*

Differences in moral judgment and husbands' internality did not account for differences in marital satisfaction at any of the four stages of educational level or three different religions. The null hypothesis was rejected.

#### Hypothesis 11

Differences between husbands and wives in moral judgment and norm-favoring do not account for differences in marital satisfaction for husbands when family life cycle, educational achievement, and religion are controlled individually.

This required 11 two-way analyses of variance. Table 24 shows the mean DAS score for each level of moral judgment above and below the mean for normfavoring. Difference scores between husbands and wives in moral judgment and norm-favoring for husbands were the independent variables. The results of the ANOVA are given in Table 25. The first four analyses were conducted at each stage of family life cycle. Differences in moral judgment and differences in normfavoring did not account for differences in marital satisfaction at the first, third, or fourth stages of family life cycle. These stages included subjects who did not have children, subjects who had teenage

Marital Satisfaction Means for Husbands at Various Levels of Husband/Wife Judgment Differences by Norm-favoring at Each Level of Control <u>Moral Judg</u> Vari<u>ables</u>

Ľ	

			Σ	Moral Judgment	gment		
Control Variable No Level	Norm- favoring	Wife higher		Equal	ч	Husband higher	
No children	Above	113.00		111.20	(5)	114.00	(5)
Young children		112.33	(9)	128.00		112.625	£ @ (
Teenagers		115.00	(2)	114.25	(4)	00.01	<u>)</u>
	Below	116.25	<del>4</del>	104.20	(2)	113.43	(2)
CULLATEN LAUNCNEQFF 1	ADOVE Below		<u>.</u>		<u> 6</u>		(4 (4
High school	Above	127.00	त)	128.75	(4)	104.00	(2)
	Below	108.50	(3)	106.20	(2)	113.00	(T)
Attended college		113.375	(8)	115.67	(12)	120.89	(6)
		109.75	(4)	106.60	(2)	113.00	(1)
B.A./B.S.	~	114.80	(07)	112.14	6	109.43	(2)
	Below	107.14	6	95.67	(e)	110.44	6)
Graduate degree	Above	96.00	(e)	120.25	(4)	115.56	6
SDA	Above	2/5.02T	(8)	09.121	<del>4</del> (	116.50 02.011	(4)
	Below	116.20	(5)	114.00	22	106.57	(77) (77)
LDS	Above	110.53	(17)	118.86	(21)	118.13	(15)
	Below	111.00	(5)	114.00	(1)	106.75	(4)
Methodist	Above	115.00	(3)	107.29	(2)	113.00	(9)
	Below	112.33	(3)	103.33	(3)	102.755	(4)

One-way ANOVA's were undertaken. \*\* Empty cell.

ANOVA for Husbands at Various Levels of Husband/Wife Moral Judgment

Differences by Norm-favoring at Each Level of Control Variables

Control Variablo	Moral	Moral judgment	nt	Nori	Norm-favoring	ring	Inte	Interaction	E	
Level	Ŀ	đf	þ	Ĺ	đf	đ	£4	đf	đ	
No children	0.17	2,20	.85	1.33	1,20	.26	0.34	2,20	.71	
Young children	2.24	2,45	.10	6.56	6.56 1,45	.01*	1.15	2,45	• 33	
Teenagers	0.47	2,28	.63	0.29	1,28	.59	0.45	2,28	.64	
Children launched**										
High school	0.84	2,10	-46	1.67	1.67 1,10	.23	0.36	2,10	.70	
Attended college	0.22	2,33	.81	1.02	1.02 1,33	.32	0.10	0.10 2,33	.91	
B.A./B.S.	1.09	2,37	.35	4.05	1,37	.052	1.60	1.60 2,37	.21	
Graduate degree	2.17	2.17 2,28	.13	3.57	3.57 1,28	.07	2.67	2.67 2,28	60.	
SDA	0.27	0.27 2,37	.76	0.00	0.00 1,37	.97	0.06	2,37	-94	
LDS	0.19	0.19 2,57	.82	0.59 1,57	1,57	.44	0.47	0.47 2,57	. 63	
Methodist	1.26	1.26 2,20	.31	1.84	1.84 1,20	.19	0.35	0.35 2,20	.71	
** Empty cell. Unabl	le to r	Unable to proceed with analysis.	with	analys		Dne-wa	ANOV	One-way ANOVA's undertaken.	l	ken.

children, and subjects whose children were grown and no longer lived at home. Neither moral judgment nor normfavoring explained differences in marital satisfaction.

Subjects who scored above the mean in normfavoring in the second stage of family life cycle scored higher in marital satisfaction than those who scored below the mean in norm-favoring. The means were tested for statistical significance. Husbands whose children no longer lived at home produced empty cells in the above analysis. The cells were collapsed. Means for difference scores between husbands and wives on the DIT were 113.50 for the group of husbands whose wives scored higher, 111.57 for the group whose wives scored equal to them on the DIT, and 119.25 for the husbands who scored higher than their wives on the DIT. The F ratio was 0.56 with 2 and 18 degrees of freedom (p = .5807). Husbands who scored above the mean in normfavoring had a marital satisfaction mean of 114.71. Those below the mean in norm-favoring scored 115.05 in marital satisfaction. The F ratio was 0.01 with 1 and 19 degrees of freedom (p = .9421). When controlled for level of educational achievement and religion, individually, no differences were observed in DAS scores. The null hypothesis was retained.

## Hypothesis 12

Differences between husbands and wives in moral judgment and self-realization do not account for differences in marital satisfaction for husbands when family life cycle, educational achievement, and religion are controlled individually.

Table 26 shows 11 two-way analyses of variance to test difference scores between husbands and wives in moral judgment and self-realization for husbands as independent variables. The results of the ANOVA are given in Table 27. The dependent variable was husbands' marital satisfaction. Control variables yielding significant results included the first and third stages of family life cycle, the third level of educational achievement, and LDS and Methodists. DAS means at the first stage of family life cycle were 121.69 above the self-realization mean and 109.61 below the selfrealization mean and significant at the .01 level. DAS means at the third stage of family life cycle were 115.7 above the mean for self-realization and 108.9 below the mean for self-realization. These means were significant at the .05 level.

DAS means at the third stage of educational achievement (graduated from college) were 114.72 above the mean for self-realization and 109.92 below the mean

Marital Satisfaction Means for Husbands at Various Levels of Husband/Wife Moral Judgment Differences by Self-realization at Each Level of Control Variables

le							
	Self- realization	wife higher		Equal	Ц	Husband higher	
	je	124.75	(4)	121.33	(3)	119.83	(9)
IOUNG CULTUREN ADOVE		107.50	c) <del>(</del>	118.08	(13) (13)	102.33 115.375	(m) (80)
Teenscore About		107.23	(13)	113.57	6	103.83	(9)
Below		114.125	<u>)</u>	92,00	(0)	111.25	( A T )
Children launched Above	/e	112.33	(e)	112.67	(e)	00.011	(2)
Below	MC	114.67	(e)	110.75	(4)	119.67	(e)
High school ** Above	/e		() ()		(4)		E.
Below	MC		(3)		(2)		(e)
Attended college   Above	/e	122.75	(4)	113.40	(10)	119.86	(2)
Below	MC	106.875	(8)	112.43	(2)	120.67	(3)
B.A./B.S.   Above	/e	111.50	(4)	114.40	(2)	116.53	(8)
	MC	111.69	(13)	100.00	(2)	103.50	(8)
Graduate degree Above		114.33	(9)	118.50	(9)	114.69	(13)
		113.00	(2)	128.00	(2)	124.00	(2)
SDA Bove		121.40	(2)	111.00	(9)	113.56	(6)
	MC	111.60	(10)	111.50	(9)	114.43	(2)
LDS Above	/e	111.625	(8)	125.00	(13)	119.38	(13)
Below	MC	110.07	(14)	109.44	(6)	107.83	(9)
Methodist   Above	/e	123.00	(1)	108.00	(9)	114.71	(2)
Below	MC	111.80	(2)	103.25	(4)	95.33	(e)

\*\* Empty cell. One-way ANOVA's undertaken.

<u>ANOVA for Husbands at Various Levels of Husband/Wife Moral Judgment</u> <u>Differences by Self-realization at Each Level of Control Variables</u>

	ral	Moral judgment	nt	Self-1	Self-realization	ation	Intei	Interaction	~
control variable Level	F	đf	۵	E4	đf	<u>a</u>	E4	đf	ď
No children 0	16.0	2,20	.42	7.61	1,20	.01*	0.19	2,20	.83
Young children 1	1.24	2,45	.30	1.32	1.32 1,45	.26	0.44	0.44 2,45	. 65
Teenagers 1	1.83 2,28	2,28	.18	4.72 1,28	1,28	.04*	1.90	1.90 2,28	.17
Children launched 0	0.45	2,15	. 65	0.00 1,15	1,15	.96	0.03	0.03 2,15	.97
High school **					<u>.</u>				
Attended college 0.	0.67	2,33	.52	1.04 1,33	1,33	.31	1.01	2,33	.38
B.A./B.S. 0.	0.43	2,37	. 65	6.04 1,37	1,37	.02*	1.58	2,37	.22
Graduate degree 0.	0.97	2,28	.39	0.97 1,28	1,28	.33	0.43	2,28	. 65
SDA 0.	0.42	2,37	.66	0.39 1,37	1,37	.54	0.61	2,37	. 55
LDS 0.	0.91	2,57	.41	5.58 1,57	1,57	.02*	1.14	2,57	.33
Methodist 3.	3.08	2,20	.07	8.96 1,20	1,20	.01*	1.80	2,20	.19

\* Statistically significant \*\* Empty cell. One-way ANOVA's undertaken.

Reproduced with permission of the copyright owner. Further reproduction prohibited without permission.

for self-realization. These means were significant at the .05 level.

DAS means for LDS subjects were 119.70 above the mean for self-realization and 109.41 below the mean. These means were significant at the .05 level. DAS means for Methodist subjects were 112.43 above the mean for self-realization and 104.83 below the mean. These means were significant at the .01 level.

Empty cells were observed at the high school level. The cells were collapsed and analyzed at the main effects level. The mean for husbands' marital satisfaction scores for those whose wives scored higher on the DIT was 114.67. Marital satisfaction score for husbands who were equal to their wives on the DIT was 116.22. Husbands who scored higher than their wives on the DIT scored 102.75 in marital satisfaction. The F ratio was 0.64 with 2 and 13 degrees of freedom (p =.5398). The marital satisfaction means for husbands who were below the mean for self-realization was 105.82. Husbands who scored above the mean on self-realization scored 127.40 in marital satisfaction. The F ratio was 5.42 with 1 and 14 degrees of freedom (p = .0354).

At five of the 11 levels, husbands scoring above the mean in self-realization also scored higher in marital satisfaction than husbands who scored below the

mean in self-realization. The null hypothesis was rejected.

## Hypothesis 13

Differences between husbands and wives in moral judgment do not account for differences in marital satisfaction for wives when family life cycle, educational achievement, and religion are controlled individually.

This required 11 one-way analyses of variance. Difference scores between husbands and wives in moral judgment was the independent variable. Table 28 shows the mean DAS score for each level of moral judgment. The results of the ANOVA are given in Table 29. The first analyses were conducted at each stage of family life cycle. Differences in moral judgment did not account for differences in marital satisfaction at any of the four stages of family life cycle, three of the four levels of educational achievement, or three different religions. Mean differences were present at the first level of educational achievement. A Scheffé test for contrasted means revealed that there was a significant difference of means for wives who scored lower than their husbands on the DIT and for husbands who scored higher than their wives on the DIT. The wives who scored higher than their husbands were much

<u>Marital Satisfaction Means for Wives at Various Levels of</u> of <u>Husband/Wife Moral Judgment Differences at Each Level</u> control Variables

		H	Moral Judgment	dgment		
Control Variable Level	Wife higher		Equal	-	Husband higher	
No children	111.86	(7)	118.14	(7)	116.89	(6)
Young children	106.865	(2)	121.00	(6)	111.70	(10)
Teenagers	120.50	(9)	108.20	(2)	106.50	(8)
Children launched	112.20	(2)	111.86	(7)	120.50	(8)
High school	91.20	(2)	115.17	(9)	120.71	(2)
Attended college	116.29	(14)	114.83	(23)	111.28	(18)
B.A./B.S.	114.55	(20)	117.17	(12)	113.67	(18)
Graduate degree	120.75	(4)	104.33	(3)	120.50	(2)
SDA	109.29	(14)	111.33	(12)	113.26	(19)
LDS	115.87	(23)	119.82	(22)	114.26	(19)
Methodist	110.50	(9)	107.90	(10)	114.80	(10)

ANOVA for Wives at Various Levels of Husband/Wife Moral Judgment

Differences at Each Level of Control Variables

		Moral Judgment	L L
Control Variable Level	<b>Гч</b>	đf	Q
No children	0.48	2,20	. 63
Young children	1.27	2,23	.30
Teenagers	2.16	2,16	.15
Children launched	1.24	2,17	.32
High school	4.44	2,15	• 03*
Attended college	0.63	2,52	.54
B.A./B.S.	0.31	2,47	.73
Graduate degree	1.98	2,6	.22
SDA	0.37	2,39	. 69
LDS	0.73	2,61	.49
Methodist	1.20	2,23	.32
* Statistically sign	significant		

132

less satisfied in marriage then those whose husbands scored higher. Because the differences were statistically significant at the .05 level, the null hypothesis was rejected.

## Hypothesis 14

Differences between husbands and wives in moral judgment and internality do not account for differences in marital satisfaction for wives when family life cycle, educational achievement, and religion are controlled individually.

This required 11 two-way analyses of variance. Table 30 shows the mean DAS score for each level of moral judgment. Difference scores between husbands and wives in moral judgment and internality scores for wives were the independent variables. The results of the ANOVA are given in Table 31. Analyses conducted at each stage of family life cycle, analyses controlling for each religion, and analyses for three out of four levels of education were not statistically significant. Differences in internality accounted for differences in marital satisfaction for wives who had teenage children. Since interaction occurred between the difference scores between husbands' and wives' moral judgment and internality for husbands, further analysis was necessary. The difference scores were tested at each

<u>Table 30</u>

<u>Marital Satisfaction Means for Wives at Various Levels of Husband/Wife</u> <u>Moral Judgment Differences by Internality at Each Level of Control</u> Variables

				Moral Judgment	gmen		
Control Variable Level	Internality	Wife higher		Equal		Husband higher	
No children	Above	120.00	(9)	130.00	(T	114.00	(5)
Young children	Above	108.67 114.375	(m)	118.375 118.36	(11)	120.50	(4)
Teenagers	Below Above	1103.44	(6)	113.375	(8)	108.33	(6)
	Below	120.40	(2)	107.67	<u>)</u>	108.50	(21)
Children launched	Above	114.00	(2)	108.00	(4)	119.00	(9)
נייארי אילע	Below	111.00	(e)	117.00	<b>(E)</b>	125.00	(3)
TOOUDS UDTH	Above	114.00	(3)	109.25	(4)	119.40	(2)
	Below	76.00	(3)	127.00	(2)	124.00	(2)
Arrended college	Above	112.00	6	113.90	(10)	114.00	(6)
	Below	120.57	(2)	115.54	(13)	108.56	(6)
B.A./B.S.	Above	118.09	(11)	120.33	(9)	112.25	(12)
	Below	110.22	(6)	114.00	(9)	116.50	(e)
Graduate degree **	Above		(e)		(2)		(2)
	Below		<del>.</del>		(1)		<u>(</u> )
SUA	Above	114.43	(2)	109.44	(6)	110.77	(13)
	Below	104.14	(2)	117.00	(3)	125.33	(3)
TDS	Above	116.56	(16)	120.22	(6)	00.011	(6)
	Below	114.29	(2)	119.54	(13)	110.00	(10)
Methodist **	Above		<u>(</u> )		(4)		(9)
	Below		(9)		(9)		(4)
-							

\*\* Empty cell. One-way ANOVA's undertaken.

els of Moral Judgment Differences by ontrol Variables

Control Variahle	Mora.	Moral judgment	nent	Inte	Internality	tу	Inte	Interaction	
Level	Êų	đf	đ	н	đf	đ	ц	đf	đ
No children	0.71	2,21	.50	0.77	0.77 1,21	.39	1.19	1.19 2,21	.32
Young children	0.87	0.87 2,44	.43	3.41	3.41 1,44	.07	0.17	0.17 2,44	.84
Teenagers	1.27	1.27 2,29	.30	0.00	0.00 1,29	.97	0.32	0.32 2,29	.73
Children launched	1.15	2,14	.35	0.44	0.44 1,14	.52	0.34	0.34 2,14	.72
High school	5.27	2,12	.02*	0.53	0.53 1,12	.48	5.27	5.27 2,12	.02*
Attended college	0.61 2,49	2,49	.55	0.18	0.18 1,49	.67	1.08	1.08 2,49	.35
B.A./B.S.	0.26 2,44	2,44	.77	0.86	0.86 1,44	.36	1.23 2,44	2,44	.30
Graduate degree **					·				<u> </u>
SDA	1.33	2,36	.28	0.74	1,36	.40	2.95 2,36	2,36	.07
LDS	0.70 2,58	2,58	.50	0.99 1,58	1,58	.33	0.40 2,58	2,58	.67
Methodist **						<u> </u>			
* Statistically significant	ni firan	+							

s undertaken.

<u>Various Leve</u> 1 Level of Co	Leve of Co
Mora	l jud
Ľ4	đf
0.71	2,21
0.87	2,44
1.27	2,29
1.15	2,14
5.27	2,12
0.61	2,49
0.26	2,44
1.33	2,36
0.70	2,58
significant One-way ANOV	cant ANOVA's
	rious Moral Moral 1.15 1.15 1.15 1.15 1.15 0.61 0.61 0.26 1.33 1.33 1.33 1.33 1.33 1.33 1.33 1.3

level of internality and internality was tested at each level of the difference scores. None of the analyses were statistically significant, with the exception of difference scores between husbands and wives where wives scored below the mean for internality. A Scheffé test of contrasts was used to determine cell differences between means. Wives who were below the mean in internality and who scored lower than their husbands in moral judgment scored higher in marital satisfaction than those who were below the mean in internality and who scored higher than their husbands in moral judgment. The null hypothesis was rejected.

Empty cells were found for women with graduate degrees. When the cells were collapsed, wives who were extraverted had a mean of 108.00 in marital satisfaction and the wives who were introverted scored 117.29 in marital satisfaction. The F ratio was 0.78 with 1 and 7 degrees of freedom (p = .4075).

Empty cells were also present for Methodist wives. When the cells were collapsed, wives who scored higher on the DIT than their husbands had a mean of 110.50 in marital satisfaction. Wives who scored equal to their husbands on the DIT scored 107.90 in marital satisfaction. Wives whose husbands scored higher on the DIT had a mean of 114.80 in marital satisfaction. The F ratio was 1.20 with 2 and 23 degrees of freedom

(p = .3204). Methodist wives who were extraverted had a marital satisfaction mean of 108.375. Introverted Methodist wives had a marital satisfaction mean of 115.60. The F ratio was 3.44 with 1 and 24 degrees of freedom (p = .0759). The null hypothesis was rejected.

# Hypothesis 15

Differences between husbands and wives in moral judgment and norm-favoring do not account for differences in marital satisfaction for wives when family life cycle, educational achievement, and religion are controlled individually.

This required 11 two-way analyses of variance. Table 32 shows the mean DAS score for each level of moral judgment. Independent variables included difference scores between husbands and wives in moral judgment and norm-favoring for wives. The results of the ANOVA are given in Table 33. Analyses conducted at the first and last stages of family life cycle, analyses at the highest level of educational achievement, and analyses for Methodists were not statistically significant. Wives who had young children and who scored above the mean in norm-favoring scored higher in marital satisfaction than wives who scored below the mean in norm-favoring. The differences were significant at the .01 level. Norm-favoring was significant when

<u>Marital Satisfaction Means for Wives at Various Levels of Husband/Wife</u> of Control <u>Moral Judgment Differences by Norm-favoring at Each Level</u> Variables

				Moral Judgment	Igmen	ц	
Control Variable	Norm-	Wife		Eanal	- -	puedzuH	
Level	favoring	higher			!	higher	
No children	Above	125.00	(2)	119.67	(9)	116.00	(3)
	Below	113.71	(2)	119.67	(e)	117.33	(9)
Young children	Above	122.83	(9)	120.18	(11)	114.375	(8)
1	Below	100.82	(11)	110.875	(8)	108.17	(9)
Teenagers	Above	123.33	(9)	120.67	(e)	109.43	(2)
	Below	110.83	(9)	103.33	(9)	112.71	(2)
Children launched**	Above		(2)		(3)		(4)
	Below		( <u>)</u>		(4)		(4)
High school	Above	106.33	(3)	122.75	(4)	125.00	(E)
	Below	68,50	(2)	100.00	(3)	117.50	(4)
Attended college	Above	123.83	(9)	119.40	(10)	114.83	(9)
	Below	110.625	(8)	111.31	(13)	109.50	(12)
B.A./B.S.	Above	122.00	(6)	120.00	6)	111.64	(11)
	Below	108.45	(11)	108.67	(E)	116.86	(2)
Graduate degree **	Above		(1)		<u>;</u> 0		(2)
	Below		(3)		(e)		<u>(</u> )
SDA	Above	122.67	(9)	118.75	(4)	114.40	(2)
	Below	99.25	(8)	107.625	(8)	113.09	(11)
ILDS	Above	123.70	(01)	124.07	(15)	115.08	(13)
	Below	109.85	(13)	110.71	(2)	112.50	(9)
Methodist	Above	105.00	(3)	107.25	(4)	116.25	(4)
	Below	116.00	(3)	108.33	(9)	113.83	(9)

\*\* Empty cell. One-way ANOVA's undertaken.

ANOVA for Wives at Various Levels of Moral Judgment Differences by Normfavoring at Each Level of Control Variables

Control Variable	Mora.	Moral judgment	nent	Norm	Norm-favoring	ing	Inte	Interaction	c
Level	îц	đf	đ	F	đf	đ	۲.	đf	đ
No children	0.21	2,21	. 89	0.33	1,21	.57	0.45	2,21	. 65
Young children	0.42	2,44	.66	8.62	8.62 1,44	.01*	1.25	1.25 2,44	.30
Teenagers	0.89	2,29	.42	4.43	4.43 1,29	.04*	2.37	2.37 2,29	.11
Children launched**								-	
High school	9.42	2,12	•00•	12.24 1,12	1,12	*00.	1.88	1.88 2,12	.20
Attended college	0.60	2,49	.55	5.95	1,49	.02*	0.35	0.35 2,49	.70
B.A./B.S.	0.04	2,44	.96	3.45 1,44	1,44	.07	3.47	3.47 2,44	.04*
Graduate degree **									
SDA	0.22 2,36	2,36	.80	10.06 1,36	1,36	*00*	3.16	3.16 2,36	.055
LDS	0.30 2,58	2,58	.74	6.59 1,58	1,58	.01*	0.84	0.84 2,58	.44
Methodist	1.21 2,20	2,20	.32	0.59 1,20	1,20	.45	0.80	0.80 2,20	.46
* Statictically cignificant	i fi ann								

\* SCAUISUICALLY SIGNIFICANT \*\* Empty cell. One-way ANOVA's undertaken.

controlled at the third stage of family life cycle. Norm-favoring and differences between husbands and wives in moral judgment were significant at the first level ofeducational achievement. Norm-favoring was also significant at the second and third levels of educational achievement. Subjects who had attended college and scored above the mean for norm-favoring, had a mean score of 119.36 on the DAS. Subjects who had completed college and who scored above the mean for norm-favoring, had a mean of 114.35 on the DAS. Subjects who had completed college and who scored below the mean for norm-favoring, had a mean of 111.28 on the Interaction was present at the third level of DAS. educational achievement, requiring additional analysis to determine the significance of each variable. The additional analysis (Table 34) supported statistical significance for norm-favoring. Significance of difference scores between husbands and wives was not upheld. Norm-favoring was also significant at the first and second levels of religion (SDA and LDS). SDAs who scored above the mean on norm-favoring, had a mean DAS score of 118.87. SDAs who scored below the mean on norm-favoring, had a mean DAS score of 107.37. LDS subjects who scored above the mean on norm-favoring, had a mean DAS score of 120.90. LDS subjects who scored

Reproduced with permission of the copyright owner. Further reproduction prohibited without permission.

<u>ANOVA for Wives with College Degrees at Various Levels of Moral Judgment</u> Differences and Above and Below the Mean for Norm-favoring

Effect	£۲	đf	đ
Moral judgment differences above mean in norm-favoring 2.93 2,26	2.93	2,26	.07
Moral judgment differences below mean in norm-favoring 1.03 2,18	1.03	2,18	.38
Norm-favoring when wives' DIT higher	6.96	6.96 1,18	.02*
Norm-favoring when spouses DIT equal	3.48	3.48 1,10	60.
Norm-favoring when husbands' DIT higher	0.77	0.77 1,16 .39	.39

\* Statistically significant

below the mean on norm-favoring, had a mean DAS score of 110.69.

Empty cells were present for wives whose children no longer live at home. When the cells were collapsed, the mean marital satisfaction score for wives who scored higher on the DIT than their husbands was 112.20. The mean marital satisfaction score for wives who scored equal to their husbands on the DIT was 111.86. The meanmarital satisfaction score for wives whose husbands scored higher on the DIT was 120.50. The F ratio was 1.24 with 2 and 17 degrees of freedom (p = .3151). The mean marital satisfaction scores for wives who were below the mean for norm-favoring was 110.00. The mean marital satisfaction scores for wives who were above the mean for norm-favoring was 119.00. The F ratio was 2.99 with one and 18 degrees of freedom (p = .1011). Empty cells were also present for the wives who have graduate degrees. The marital satisfaction means foreach level of DIT difference from the husbands' scores was discussed earlier under hypothesis 14. Wives who were below the mean for norm-favoring had a marital satisfaction mean of 111.33, while above the mean normfavoring wives scored of 123.00. The F ratio was 1.78 with 1 and 7 degrees of freedom (p = .224). Based upon all findings, the null hypothesis was rejected.

Reproduced with permission of the copyright owner. Further reproduction prohibited without permission.

## Hypothesis 16

Differences between husbands and wives in moral judgment and self-realization do not account for differences in marital satisfaction for wives when family life cycle, educational achievement, and religion are controlled individually.

This required 11 two-way analyses of variance. The independent variables were differences between husbands and wives in moral judgment and self-realization scores for wives. Table 35 shows the mean DAS score for each level of moral judgment. The results of the ANOVA are given in Table 36. The first analyses were conducted at each stage of family life cycle. No main effect differences were observed. However, interaction was found between moral judgment and internality for subjects who had no children. This required further one-way analyses for moral judgment at each level of self-realization and for self-realization at each level of moral judgment (Table 37). The only significant result from this analysis was self-realization for the group whose husbands scored higher on the DIT. Wives who had no children and whose husbands scored higher on the DIT were more satisfied in marriage if they scored above the mean in self-realization.

Differences in moral judgment and wives' selfrealization did not account for differences in marital

<u>Marital Satisfaction Means for Wives at Various Levels of Husband/Wife</u> <u>Moral Judgment Differences by Self-realization at Each Level of Control Variables</u>

				Moral Judgment	lgmen	<u>ب</u>	
Control Variable Level	Self- realization	Wife higher		Equal	I T	Husband higher	q
No children	Above	105.67	(3)	120.33	(3)	125.75	(4)
Young children	Above	01.011	(10)	119.33 118.27	(11)	113.60	(2) (2)
Teenagers	Below	1106.43	( <u>)</u>	113.50	(8)	110.67	(6)
	Below	117.80	(2)	104.33	<u>)</u>	109.00	(2) (2)
Children launched	Above	116.00	(3)	97.00	(1)	125.67	(2)
High school **	Below	106.50	(2)	114.33	(9)	117.40	(5)
	Below		(2)		(o) (9)		$\left[2\right]$
Attended college	Above	113.00	(8)	117.40	(15)	115.67	) (6)
	Below	120.67	(9)	110.00	(8)	106.89	(6)
B.A./B.S.	Above	114.27	(11)	116.00	(4)	117.00	(8)
tt goodste goodste	Below	114.89	(6)	117.75	(8)	111.00	(01)
Graquate degree **	Above Below		6 6 9		(5)		<u>6</u> 0
SDA	Above	105.43	(2)	110.25	(4)	114.29	(2)
	Below	113.14	(2)	111.875	(8)	112.89	(6)
ILDS	Above	114.85	(13)	122.58	(12)	117.73	(11)
	Below	117.20	(01)	116.50	(01)	109.50	(8)
Methodist	Above	117.00	(e)	103.20	(2)	121.00	(3)
	Below	104.00	(3)	112.60	(2)	112.14	(2)

\*\* Empty cell. One-way ANOVA's undertaken.

ANOVA for Wives at Various Levels of Moral Judgment Differences by Selfrealization at Each Level of Control Variables

Control Variablo	Mora	Moral judgment	nent	Inté	Internality	ty	Inte	Interaction	u
Level	F	đf	d	ы	df	đ	F	đf	ď
No children	0.58	2,21	.57	0.01	1,21	.94	3.79	2,21	.04*
Young children	0.96	0.96 2,44	.39	0.64	0.64 1,44	.43	0.01	0.01 2,44	66.
Teenagers	1.34	1.34 2,29	.28	0.41	0.41 1,29	.52	0.24	2,29	.78
Children launched	2.54 2,14	2,14	.11	0.00	0.00 1,14	.98	1.69	1.69 2,14	.22
High school **									
Attended college	0.70 2,49	2,49	.50	0.60	0.60 1,49	.44	1.84	1.84 2,49	.17
B.A./B.S.	0.19 2,44	2,44	.83	0.11	0.11 1,44	.74	0.46	0.46 2,44	.64
Graduate degree **									
SDA	0.37	0.37 2,36	. 69	0.37	0.37 1,36	.55	0.42	0.42 2,36	• 66
LDS	0.76 2,58	2,58	.47	1.04	1.04 1,58	.31	0.70	0.70 2,58	.50
Methodist	2.09 2,20	2,20	.15	1.20	1.20 1,20	. 29	3.59	3.59 2,20	.046*
<pre>* Statistically significant</pre>	lificar	t –						i	

One-way ANOVA's undertaken. Unable to proceed with analysis. \*\* Empty cell.

satisfaction at any of the four stages of educational level. Wives whose educational level was not past the high-school level and who scored higher than their husbands in moral judgment did not score as high in marital satisfaction as wives who scored equal to or lower than their husbands in moral judgment. By applying the Scheffé test for contrasting means to the difference groups between husbands and wives in moral judgment for wives in the first level of education, differences were present between the first and third marital satisfaction means. Because of the empty cells for wives who had graduate degrees, the two-way analysis was collapsed into two one-way analyses. The mean marital satisfaction score was 120.75 for wives whose DIT scores were higher thantheir husbands. Wives who scored equal to their husbands on the DIT scored 104.33 in marital satisfaction. Wives who scored lower than their husbands on the DIT scored 120.50 in marital satisfaction. Analysis of variance yielded an F ratio of 1.98 with 2 and 6 degrees of freedom (p = .2191).

The mean marital satisfaction scores for the same wives who scored below the mean for self-realization was 118.00. The mean marital satisfaction scores for the wives who scored above the mean for self-realization was

<u>ANOVA for Wives with No Children at Various Levels of Moral Judgment</u> <u>Differences and Above and Below the Mean for Self-realization</u>

Effect	F	đf	đ
Moral judgment differences at low self-realization	2.08	2.08 2,14	.16
Moral judgment differences at high self-realization	1.60	1.60 2,7	.27
Self-realization at wives' DIT higher	2.00	2.00 1,7	.20
Self-realization at spouses equal on DIT	0.02	0.02 1,7	.88
Self-realization at husbands' DIT higher	6.88	6.88 1,7	.03*

Statistically significant

\*

113.83. The F ratio was 0.19 with 1 and 7 degrees of freedom (p = .6795).

Because of the empty cells for wives who had high school education or less, the two-way analysis was collapsed into two one-way analyses. The mean marital satisfaction scores was 91.20 for wives whose DIT scores were higher than their husbands. Wives who score equal to their husbands on the DIT scored 115.17 in marital satisfaction. Wives who scored lower than their husbands on the DIT scored lower than their satisfaction. The analysis of variance yielded an F ratio of 4.44 with 2 and 15 degrees of freedom (p = .0306).

The mean marital satisfaction scores for the same wives who scored below the mean for self-realization was 112.14. The mean marital satisfaction scores for the wives who scored above the mean for self-realization was 105.50. The F ratio was 0.31 with 1 and 16 degrees of freedom (p = .5880).

Interaction between self-realization and differences scores between husbands and wives was present for Methodist wives. This required additional one-way analyses. Moral judgment difference scores were tested at each level of self-realization and selfrealization was tested at each level of moral judgment differences (Table 38). No significant simple effects

ANOVA for Methodist Wives at Various Levels of Moral Judgment Differences and Above and Below the Mean for Self-realization

Effect	<u>ل</u> عا	đf	2
	-		4
Moral judgment differences at low self-realization	1.28	1.28 2,12	.31
Moral judgment differences at high self-realization	3.14	3.14 2,8	.10
Self-realization at wives' DIT higher	3.09	3.09 1,4	.15
Self-realization at spouses equal on DIT	2.18	2.18 1,8	.18
Self-realization at husbands' DIT higher	2.44	2.44 1,8	.16

Statistically significant

\*

emerged from the one-way analyses. However, because significant difference in self-realization scores was found for wives with no children and whose husbands scored higher in moral judgment, the null hypothesis was rejected.

#### Summary

Sixteen research hypotheses involving moral judgment and personality were tested for statistical significance in predicting or accounting for marital satisfaction using stepwise- and multiple-regression analyses and analyses of variance. The sample consisted of 132 married couples who volunteered as subjects for this research study. Forty-two couples were members of the Seventh-day Adventist Church (SDA), 64 couples were members of the Church of Jesus Christ of Latter Day Saints (LDS), and 26 couples were members of the United Methodist church.

Stages of family life cycle were defined and found to be fairly uniform in number across 6 stages. Educational achievement was measured at four levels: attended high school, attended college but did not graduate, graduated from college, and attained an advanced academic degree. Most of the subjects had attended college or graduated from college.

Moral judgment and differences in moral judgment between spouses did not correlate with marital satisfaction. Internality did not correlate with marital satisfaction. However, all of the multipleregression analyses and analyses of variance identified norm-favoring and self-realization as relating to marital satisfaction under certain controls. Wives who rated high in norm-favoring tended to be satisfied in their marriages. Kohlberg's developmental stage theory did not consistently identify subjects who were satisfied in marriage. Husbands who scored high in marital satisfaction also tended to score high in selfrealization on the CPI.

#### CHAPTER V

SUMMARY, DISCUSSION, CONCLUSIONS, AND RECOMMENDATIONS

## Summary

# Problem

Increased divorce rates indicate a necessity for investigation into alternative viewpoints relating to elements motivating the marital relationship. Research findings are not consistent in ascertaining the relationship between marital satisfaction and moral judgment.

As new measures of personality are introduced, it is important to examine their unique application in various human functions. Marriage, as a major function of human experience, warrants such attention. The recent revision of the California Psychological Inventory (Gough, 1987) presents three personality factors. An examination of the correlations of each factor to marital satisfaction was needed. The theoretical basis for this study assumes that, once the influences of stage of family life cycle, level of educational achievement, and religion are removed, moral

152

judgment and personality are important factors in marital happiness.

# Literature

Three areas of the literature were considered: (1) marital satisfaction, (2) moral judgment, and (3) personality and marital satisfaction. Burgess and Cottrell (1939) conducted a large-scale study of marital adjustment. Socialization, conformity to social rules, and respect for convention were important in preparing an individual for adjustment in marriage (p. 346). During the same period, Terman studied marital adjustment in relationship to sexual factors, personality, and marital adjustment. He concluded that happily married men tended to be more conservative than unhappily married men (Terman, 1938, p. 163). This appeared to be especially true in the areas of sexual morality and religious attitudes. The most comprehensive analysis to be undertaken is the Kelly Longitudinal Study (KLS). The study has continued for the past 45 years and encompasses both marital stability and marital satisfaction for 300 couples. Three personality characteristics were identified as impeding marital compatibility. These included husband's neuroticism, wife's neuroticism, and husband's impulse control.

Burr (1970), in a random sample of 116 intact couples, found a curvilinear relationship between marital satisfaction and stage of family life cycle. The lowest levels of marital satisfaction appeared to be during the stages of child rearing. Marital satisfaction appeared to increase during the postparental and retirement stages.

Moral judgment has been described by Kohlberg (1976) as occurring within three level which are expressed in six stages. Level I (preconventional) includes stages 1 and 2. Stage 1 reveals heteronomous morality which typifies moreal reasoning in which the motive is to avoid punishment. Stage 2 is labelled "instrumental" and describes a mode of reasoning which recognizes the interests of others. Moral actions reflect mutuality of interest.

Level II (conventional) consists of stages 3 and 4. Stage 3 moral judgment relates to social expectations, trust, loyalty to society, and gratitude. Stage 4 considers the social system and conscience. In stage 4, laws and rules are to be upheld except in those few cases in which the laws or rules conflict with other social duties.

Level III (postconventional) incorporates stages 5 and 6. Stage 5 considers social contract or utility and individual rights. Stage 5 reasoning is aware of a

variety of values and opinions held by others. Rules are relative to the larger group. Stage 6 exemplifies universal ethical principles. When laws infract upon the individual's principles, the person acts in accordance with the principle.

The relationship between moral judgment and marital satisfaction is still not well defined. Several studies have been conducted for the purpose of understanding the relationship, but the one most applicable to this study was conducted by Deters (1983). His total sample did not correlate moral judgment and marital satisfaction. However, when he controlled religion by only analyzing Catholics, moral judgment accounted for 17% of marital satisfaction. There is general agreement among the researchers that there is a difference in marital satisfaction between religiously oriented homes and nonreligiously oriented homes. In Deters' study, religion appeared to be an interacting variable. Register (1975) did not observe a relationship between moral judgment and marital satisfaction. Nims (1984) also examined the relationship between marital satisfaction and moral development. He concluded that wives scoring above the median in moral judgment revealed greater marital satisfaction than those who scored below the median. However, if the wife scored below the median on the DIT,

her marriage tended to be more satisfying to her if her husband scored above the median on the DIT (Nims, 1984). Nims suggested stage of family life cycle should be controlled in subsequent studies.

## Purpose

The purpose of this study was to observe relationships between moral judgment and personality as they pertain to marital satisfaction. Two research questions were addressed:

1. Is there a relationship between moral judgment, personality, and marital satisfaction?

2. Do differences between spouses in moral judgment and personality explain differences in marital satisfaction?

From these two questions, 16 hypotheses were formulated. Eight hypotheses were used to test the first question, and eight hypotheses tested the second question. Stage of family life cycle was controlled. Level of educational achievement has been identified as a correlate of moral judgment (Rest, 1979, p. 111). To assure moral judgment was relating independent of education, level of educational achievement was controlled.

## Methodology

The population for this study included members of three different religious denominations: Church of Jesus Christ of Latter Day Saints (LDS), Seventh-day Adventists (SDA), and United Methodists. The Latter Day Saints were found in Orem, Utah, United Methodists were predominantly from Oklahoma City, Oklahoma, as well as Kimberly, Idaho, and the Seventh-day Adventists were from the Walla Walla Valley, which includes Walla Walla and College Place, Washington, and Milton-Freewater, Oregon. Ministers and bishops were contacted in each area and invited to offer testing to members of their respective churches. In order to reduce cultural interaction among the variables, data for this study included only married couples who were Caucasian, nonhispanic.

The research instruments included a demographic questionnaire, The Dyadic Adjustment Scale (Spanier, 1976), The Defining Issues Test (a test of moral judgment) (Rest, 1986a), and The California Psychological Inventory (Gough, 1987). A research assistant was employed in each community and was responsible for disseminating test packets, communicating with participating clergymen, and collecting the instruments after completion. Both spouses were requested to complete the instruments at home without consultation with each other.

The major statistical methods used to analyze the collected data included multiple-regression analysis, stepwise-regression analysis, and analysis of variance. Probability was set at .05. The data were initially analyzed descriptively.

## Findings

Descriptive data were analyzed first. The sample included 128 (48.5%) Latter Day Saints, 84 (31.8%) Seventh-day Adventists, and 52 (19.7%) United Methodists. Stage of family life cycle was somewhat evenly distributed in the sample. Three wives were pregnant with the first child. Fifty-two subjects had at least one preschool child, but no children over six years of age. Forty-nine subjects stated that they had at least one child over six years of age, but no teenagers. There were 36 subjects who claimed to have at least one teenage child. Thirty-three subjects stated that one child no longer lives at the subject's home, but not all children have left yet.

Regarding levels of educational achievement, 12.9% had a high-school education, 35.6% Had attended college, but did not graduate, 35.2% graduated from college, and 16.3% had advanced academic degrees.

Marital satisfaction, as measured by the Dyadic Adjustment Scale (Spanier, 1976), presented a mean of 113.64 with a standard deviation of 14.13 for this sample. The mean P score for the DIT was 21.32 with a standard deviation of 7.4 while the D score was 25.36 with a standard deviation of 7.24. Personality was measured by the CPI (Gough, 1987). The first personality variable (internality) presented a mean of 18.5 with a standard deviation of 7.52. The second personality variable (norm-favoring) had a mean of 25.10 with a standard deviation of 4.14. The third personality variable (self-realization) yielded a mean of 37.07 with a standard deviation of 7.78.

This sample presented very low scores on the Defining Issues Test (DIT). Rest (1979, p. 115) presented studies which suggest that people who subscribe to a liberal religious orientation tend to obtain higher scores on the DIT than those who are conservatively oriented. The results of the normfavoring personality variable suggest that the subjects in this research considered aspects of moral judgment which are not defined in the Kohlberg theory. Commitment, conscientiousness, and dependability, features of the norm-favoring variable, may also describe the subject's attitude toward moral issues as well as the marital relationship.

Most moral development research has focused upon chronological development from childhood to adulthood. The present research has demonstrated that there is a negative correlation between moral development and the number of years married. Since there is no evidence in the literature to suggest that moral development reverses during adult life, it is possible that adult moral judgment develops in dimensions which have not been defined. The most probable explanation of the finding is that sociologically there are moral judgment differences between older respondents and younger respondents.

The findings of this study were examined by testing 16 null hypotheses.

## Hypothesis 1

There is no correlation between moral judgment and marital satisfaction.

There was no correlation between the Defining Issues Test (DIT) P score and the Dyadic Adjustment Scale (DAS). The DAS correlated .04 with the D score of the DIT and was not statistically significant. The null hypothesis was retained.

## <u>Hypothesis 2</u>

There is no correlation between moral judgment and marital satisfaction when the effect of family life

cycle, educational achievement, and religion are controlled individually.

Neither measures of moral judgment (P index and D index) correlated with marital satisfaction after the variances for stage of family life cycle, educational achievement, and religion were removed individually. The null hypothesis was retained.

#### Hypothesis 3

Moral judgment and personality are not related to marital satisfaction.

In this analysis, stepwise-multiple regression was used. Norm-favoring and self-realization entered while accounting for 12.5% of the variance of marital satisfaction at the .001 level of significance. Normfavoring was the first personality variable to enter the model, accounting for 10% of the common variance with marital satisfaction. Self-realization accounted for 2.5% of marital satisfaction variance. Moral judgment did not enter the stepwise-multiple-regression model. Based upon the relationship of norm-favoring and selfrealization to marital satisfaction, the null hypothesis was rejected. However, no relationship was observed between moral judgment and marital satisfaction in this regression model.

Reproduced with permission of the copyright owner. Further reproduction prohibited without permission.

#### Hypothesis 4

Moral judgment and personality are not related to marital satisfaction when the effect of family life cycle, educational achievement, and religion are controlled individually.

Moral judgment did not account for marital satisfaction in this study. However, the two personality variables, norm-favoring and selfrealization, accounted for 14% of the variance of marital satisfaction. The null hypothesis was rejected, even though family life cycle, educational achievement, and religion did not influence the results.

### Hypothesis 5

Differences in moral judgment do not account for differences in marital satisfaction when family life cycle, educational achievement, and religion are controlled individually.

In this analysis, moral judgment did not explain marital satisfaction when analyzed with any of the three control variables. The null hypothesis was retained.

#### Hypothesis 6

Differences in moral judgment and internality do not account for differences in marital satisfaction when family life cycle, educational achievement, and religion are controlled individually.

Reproduced with permission of the copyright owner. Further reproduction prohibited without permission.

Independent variables included moral judgment and internality. The dependent variable was marital satisfaction. Two-way analyses were conducted at each level for each of the control variables. Subjects who had attained no more than high-school education and who scored high in moral judgment scored low in marital satisfaction. The null hypothesis was rejected.

### Hypothesis 7

Differences in moral judgment and norm-favoring do not account for differences in marital satisfaction when family life cycle, educational achievement, and religion are controlled individually.

Independent variables included moral judgment and norm-favoring. The dependent variable was marital satisfaction. Two-way analyses were conducted at each level for each of the control variables. Subjects at four particular levels of the control variables who scored above the mean in norm-favoring scored higher in marital satisfaction than those who scored below the mean in norm-favoring. The null hypothesis was rejected.

#### <u>Hypothesis 8</u>

Differences in moral judgment and selfrealization do not account for differences in marital

satisfaction when family life cycle, educational achievement, and religion are controlled individually.

Independent variables included moral judgment and self-realization. The dependent variable was marital satisfaction. Two-way analyses were conducted at each level for each of the control variables. Subjects who had attained no more than high-school education and who scored high in moral judgment scored low in marital satisfaction. The null hypothesis was rejected.

In each of the following analyses, difference scores between husbands and wives in moral judgment were established by subtracting the wives' scores from the husbands' scores. The scores were divided into three equal groups by placing cutpoints at .1667 standard deviation in both directions from the mean. Subjects who scored within the first group were couples in which the wife scored higher in moral judgment than the husband. The middle group consisted of couples who obtained moral judgment scores which were not different from each other. The third group consisted of couples in which the husband scored higher in moral judgment than the wife.

## <u>Hypothesis 9</u>

Differences between husbands and wives in moral judgment do not account for differences in marital

satisfaction for husbands when stage of family life cycle, educational achievement, and religion are controlled individually.

The independent variable was difference score between husbands and wives in moral judgment. The dependent variable was husbands' marital satisfaction. No significant differences were observed. The null hypothesis was retained.

#### <u>Hypothesis 10</u>

Differences between husbands and wives in moral judgment and internality do not account for differences in marital satisfaction for husbands when family life cycle, educational achievement, and religion are controlled individually.

Independent variables included difference scores in moral judgment between husbands and wives and internality for husbands. The dependent variable was marital satisfaction for husbands. When controlled for husbands who were fathers of teenage children, subjects who scored above the mean in internality scored lower in marital satisfaction than husbands who scored below the mean in internality. That is, husbands who tend to be outgoing tended to score higher in marital satisfaction. The null hypothesis was rejected.

#### Hypothesis 11

Differences between husbands and wives in moral judgment and norm-favoring do not account for differences in marital satisfaction for husbands when family life cycle, educational achievement, and religion are controlled individually.

Independent variables included difference scores between husbands and wives in moral judgment and normfavoring. Subjects who scored above the mean in normfavoring in the second stage of family life cycle scored higher in marital satisfaction than those who scored below the mean in norm-favoring. The null hypothesis was rejected.

### Hypothesis 12

Differences between husbands and wives in moral judgment and self-realization do not account for differences in marital satisfaction for husbands when family life cycle, educational achievement, and religion are controlled individually.

At five levels of the control variables, husbands scoring above the mean in self-realization also scored higher in marital satisfaction than husbands who scored below the mean in self-realization. The null hypothesis was rejected.

#### Hypothesis 13

Differences between husbands and wives in moral judgment do not account for differences in marital satisfaction for wives when family life cycle, educational achievement, and religion are controlled individually.

The independent variable was the difference scores between husbands and wives in moral judgment. The dependent score was marital satisfaction for wives. Mean differences were present for subjects who had attained no more than high school education. Wives who scored higher than their husbands in moral judgment scored lower in marital satisfaction. Because the differences were significant at the .05 level, the null hypothesis was rejected.

#### Hypothesis 14

Differences between husbands and wives in moral judgment and internality do not account for differences in marital satisfaction for wives when family life cycle, educational achievement, and religion are controlled individually.

Independent variables included internality and moral judgment difference scores between husbands and wives. The dependent variable was wives' marital satisfaction. Extraverted women (i.e., women who scored

below the mean in internality) who attained no more than high-school education scored higher in marital satisfaction if their husbands scored higher than they in moral judgment. The null hypothesis was rejected.

### Hypothesis 15

Differences between husbands and wives in moral judgment and norm-favoring do not account for differences in marital satisfaction for wives when family life cycle, educational achievement, and religion are controlled individually.

Independent variables included norm-favoring and difference scores between husbands and wives. The dependent variable was marital satisfaction for wives. At five levels of the control variables, wives who scored above the mean in norm-favoring scored higher in marital satisfaction. The null hypothesis was rejected.

#### Hypothesis 16

Differences between husbands and wives in moral judgment and self-realization do not account for differences in marital satisfaction for wives when family life cycle, educational achievement, and religion are controlled individually.

Independent variables included self-realization and moral judgment difference scores between husbands and wives. Wives who had no more than high-school

education and who scored higher than their husbands in moral reasoning did not score as high in marital satisfaction as wives whose husbands scored equal to them or higher. The null hypothesis was rejected.

#### <u>Discussion</u>

When each of the variables was controlled at the various levels of family life cycle, education, and religion, several general themes became apparent (Tables 35 through 37). Norm-favoring appeared to have the greatest overall influence for the sample as a whole. Subjects who scored above the mean in norm-favoring also scored higher in marital satisfaction than those who scored below the mean. Five of the 11 control variables supported the finding. However, when the analyses were conducted on husbands and wives separately, the influence of norm-favoring was primarily an issue with the wives, even though fathers of young children were found to be more satisfied in marriage if they scored above the mean in norm-favoring. Husbands with teenage children differed from other husbands, in that internality was important to marital satisfaction. When the sample was limited to wives, seven of the 11 control levels were significant in identifying norm-favoring as important to marital satisfaction. The most important

Reproduced with permission of the copyright owner. Further reproduction prohibited without permission.

Table 39

Variables Which Account for Higher Marital Satisfaction for Both Spouses at Various Levels of the Control Variables

Control Variable Level	DIT	DIFF	τΛ	V2	V3
No children					
Young children				> mean	
Teenagers					
Children launched					
High school	Low			> mean	
Attended college				> mean	
B.A./B.S.				> mean	
Graduate degree					
SDA					·
LDS					> mean
Methodist					

Table 40

<u>Variables Which Account for Higher Marital Satisfaction for Husbands at</u>

<u>Variables</u>
Control
the
of
Levels
Various

Control Variable Level	DIT	DIT DIFF	ТЛ	V2	V3
No children					> mean
Young children				> mean	
Teenagers			< mean		> mean
Children launched					
High school					
Attended college					
B.A./B.S.					> mean
Graduate degree					
SDA				_	
LDS					> mean
Methodist					> mean

Table 41

<u>Variables Which Account for Higher Marital Satisfaction for Wives at</u> <u>Various Levels of the Control Variables</u>

Control Variable Level	DIT	DIT DIFF	١٨	V2	V3
No children					
Young children				> mean	
Teenagers				> mean	
Children launched					
High school		husband		> mean	
Attended college		-> wire			
B.A./B.S.					
Graduate degree					
SDA				> mean	
LDS				> mean	
Methodist					

variable for husbands' marital satisfaction was selfrealization, which was significant for 5 of the 11 control variables.

There is a general tendency for wives who are "well organized, conscientious, conventional, dependable, and controlled" (Gough, 1987, p. 19) to be more satisfied in marriage. They are inclined to take a serious attitude toward ethical and moral issues. There is also a general tendency for husbands who tend to be "relatively free of neurotic trends and conflicts, moderate, mature, insightful, optimistic, and as having a wide range of interests" (p. 20) to be more satisfied in marriage.

This research identified two general obstacles with the DIT. The first problem is related to the theoretical basis for the DIT. The second concern is with the construction of the instrument.

Kohlberg's theory of moral development examines the cognitive thought processes for the decision, beginning with punishment avoidance and progressing through six stages toward principled morality. Since the norm-favoring variable on the CPI addresses the subjects' commitment to moral behavior, one might expect to find a relationship between Rest's DIT and normfavoring. On the contrary, there was no correlation

between the two variables in this research. Normfavoring was an important variable in marital satisfaction for wives. One of the questions for the norm-favoring variable addressed the importance of ethical and moral issues to the subject. Subjects who scored high in norm-favoring recognized the importance of making moral and ethical decisions. Many of those individuals, however, scored low on the DIT. In view of such results, it is very unlikely that the DIT measures the multidimensional construct of "moral judgment" in more than one direction. Kohlberg's theory focuses upon thinking about issues of importance as one considers a given social issue. There is no consideration for actual behaviors which may result from the cognitive Furthermore, the behaviors which carry out the process. moral intent at the moment of excecution may be based upon different criteria than those considered in the safety of one's home where the issue is only hypothetical. A theory of moral judgment which deliberately omits the "rightness" or "wrongness" of a moral decision and which does not consider the action following a moral decision probably underestimates the construct.

Even though the DIT has been identified as the most reliable objective measure of moral judgment at the present time, the assumption (Chapter 1) that it would

reflect the true opinion of the subjects was not upheld. Numerous subjects in this research reported frustration and hostility toward the DIT. Two general themes were repeated continuously from all three religions. First, many subjects viewed the DIT as addressing issues which were important in the 1960s and 1970s but not the 1980s. For example, it would be unlikely that students would take over the university administration building. There are likely other ways for them to gain influence. In this research, the subjects viewed the DIT as appearing to address the fundamental issues of another culture and another time in American history rather than the fundamental issues of moral judgment.

Second, they reported that the available options for response were too limited. Many subjects felt Heinz, for example, would have more options than to either steal or not steal the drug for his wife. Today he might explore ways to help the druggist market the drug. In exchange, the druggist would provide medication for Heinz' wife. There might be mutual gain. One subject was very upset because there was no option to provide a new trial for the escaped prisoner.

Wives who have teenage children and wives who have no more than a high-school education were more satisfied in marriage when their husbands' moral judgment was equal to or better than their own. Wives

within these groups who scored higher than their husbands in moral judgment scored very low in marital satisfaction. When considering the lower educational level and stressful stage of family life cycle, it is probable that wives whose moral judgment exceeds their husband's experience greater stress in the marital relationship and may tend to be less tolerant of their husbands. This was especially true for extraverted wives.

The Kelly Longitudinal Study (KLS) showed a relationship between social extraversion and men who divorced late in life (Kelly & Conley, 1987). Husbands in the present study who have teenage children and who were inclined toward extraversion scored higher in marital satisfaction than husbands who were inclined toward internality. The apparent inconsistency is resolved by considering the definitions of extraversion. In the KLS, extraverted husbands were also described as having lower levels of agreeableness, higher levels of neuroticism, inclined toward closer families of origin, and having more premarital sexual activity. Gough's (1987, p. 19) definition of the extraverted subject describes him as "outgoing, confident, talkative, and as having social poise and presence."

In short, conscientious, dependable wives tend to be more satisfied in marriage than restless, pleasure-

seeking, self-indulgent wives. For husbands, maturity and insightfulness are important personal qualities for their marital satisfaction.

### <u>Conclusions</u>

A review of the findings which have been presented suggests the following conclusions in response to the original research questions:

1. Is there a relationship between moral judgment, personality, and marital satisfaction?

No relationship was found between moral judgment and marital satisfaction. Norm-favoring and selfrealization were found to be related to marital satisfaction, but internality had very little effect. Norm-favoring was found to be important for wives' marital satisfaction. For husbands, self-realization was most important.

When viewed from the perspective of analysis of variance, high and low scores in moral judgment and internality did not explain differences in marital satisfaction. However, norm-favoring explained differences in marital satisfaction for wives and selfrealization explained differences in marital satisfaction for husbands.

2. Do differences between spouses in moral judgment and personality explain differences in marital satisfaction?

For wives who had teenage children and wives with only a high school education, difference scores in moral judgment between husbands and wives accounted for differences in marital satisfaction. Wives who scored higher in moral judgment than their husbands tended to score low in marital satisfaction. Fathers of teenage children who tended to be extraverted were more satisfied in marriage than introverted fathers. Normfavoring tended to be a significant variable for wives. Self-realization for husbands explained differences in marital satisfaction.

## Recommendations for Further Study

On the basis of the conclusions formulated from this research, specific recommendations for additional research are presented.

1. Additional research is recommended to ascertain the relationship between these variables and a more comprehensive definition of moral judgment.

2. Because the moral judgment scores were low in this sample compared to other studies, this study should

be replicated with individuals from a wider variety of religions to ensure greater variance.

3. Sociological research is required to explain the phenomenon of a negative correlation between moral judgment and number of years married. Since there is no theoretical basis for adults to regress in moral judgment during adulthood, the most logical explanation for the phenomenon must consider age-related, generational cohort cultural differences.

4. Further research is required to determine how this research may apply to clinical practice.

APPENDICES

.

\_\_\_\_

Reproduced with permission of the copyright owner. Further reproduction prohibited without permission.

•

## APPENDIX A

.

## CORRESPONDENCE

Reproduced with permission of the copyright owner. Further reproduction prohibited without permission.

\_\_\_\_\_

c.	Wallace
s.	Main
UT	84058
22	<u>5-3847</u>
	S. UT

Dear Pastor

As a doctoral candidate majoring in Counseling Psychology at Andrews University, I am conducting research which examines the relationship between moral judgment, personality, and marital satisfaction.

There have been numerous studies which have examined the relationship between personality and marital satisfaction. However, the literature is virtually void of studies which have explored the relationship between moral judgment and marital satisfaction. From a religious perspective, we recognize the value of moral conduct. We do not know, however, what the relationship is between our <u>basis</u> for moral conduct and the marital relationship. I consider this study to be an opportunity to make a contribution toward our better understanding of the marital relationship.

The instruments I will be using include The Dyadic Adjustment Scale (DAS), The Defining Issues Test (DIT), and the California Psychological Inventory (CPI). All three instruments have been widely used in contemporary research.

I am requesting your endorsement to conduct this study among members of your church. Testing will require approximately 2 hours of the participants' time. Confidentiality will be maintained at the highest level. Couples may arrange with the examiner to receive their test results.

I anticipate some very interesting results which may be useful in both pastoral and clinical counseling, as well as sermons and lectures. In appreciation to you and the members of your church who volunteer to participate, I will be more than happy to share a compendium of the results with you upon completion of the study.

Within the next few days, a research assistant will contact you personally about the possibility of an appointment.

182

Reproduced with permission of the copyright owner. Further reproduction prohibited without permission.

Thank you for considering this request.

Sincerely,

Thomas C. Wallace Ph.D. Doctoral Candidate in Counseling Psychology

\_\_\_\_\_

Reproduced with permission of the copyright owner. Further reproduction prohibited without permission.

Dear Friends and Neighbors:

Too many people have started marriage with the hope of living blissfully forever but have ultimately become disillusioned. Skillful counselors are constantly seeking better ways to help couples rejuvenate their marriage. As a doctoral candidate in counseling psychology, I am especially interested in studying the marital relationship so I may be more effective in helping people.

This is where you come into the picture. To complete my studies I need your help in "The Marital Interaction Research Project." <u>I am looking for couples who would be</u> willing to be part of this study. Each person will be asked to complete a personality inventory and questionnaires about marital satisfaction and opinions about social problems. It will probably take about two hours to complete these instruments. This can be done in the privacy of your home and at a time convenient to you. <u>Your responses will be completely anonymous</u>. If you would like to know the results of your tests, special arrangements can be made.

I am convinced that knowledge to be gained from this research will be very helpful in counseling couples whose marriages are in crisis. Church leaders from several denominations and counseling professionals have encouraged this study and are looking forward to using the information in their work with people. You may even know someone personally whose marriage could be helped as a result of this study.

Thank you for considering this request. <u>Please complete</u> and return the enclosed postcard. If you have any questions, feel free to call me collect at (801) 225-3847.

I am looking forward to hearing from you soon--and thanks again!

Gratefully yours,

Thomas C. Wallace Ph.D. candidate in Counseling Psychology 1051 S. Main Orem, UT 84058

604 W. John Beers Rd. #142 Stevensville, MI 49127 June 16, 1988

Dr. James R. Rest Center for the Study of Ethical Development University of Minnesota 141 Burton Hall 178 Pillsbury Dr., S.E. Minneapolis, MN 55455

Dear Dr. Rest:

I am a Ph.D. graduate student in Counseling Psychology at Andrews University. I am currently engaged in a doctoral dissertation which investigates relationships of moral judgment and selected personality variables to marital satisfaction. I am asking for your permission to administer the Defining Issues Test (DIT). The test will be administered to 180 American, caucasian, married couples.

The purpose of the study is to observe relationships between moral judgment, personality, and marital satisfaction when stage of family life cycle, religion, or level educational achievement are controlled.

Thank you in advance for your early response.

Sincerely yours,

Thomas C. Wallace Graduate Student

Donna J. Habenicht, Ed.D. Professor of Educational and Counseling Psychology

## APPENDIX B

## REQUEST FOR TEST RESULTS

Reproduced with permission of the copyright owner. Further reproduction prohibited without permission.

\_\_\_\_\_

#### REQUEST FOR TEST RESULTS

I request the results of the following tests which were administered to me as a participant in The Marital Interaction Research Project:

- ( ) Defining Issues Test
- ( ) California Psychological Inventory
- ( ) Dyadic Adjustment Scale

TEST IDENTIFICATION NUMBER (INCLUDE ALL 4 DIGITS):\_\_\_\_\_

NAME:\_\_\_\_\_

ADDRESS:

TELEPHONE NUMBER (INCLUDE AREA CODE IF NOT IN UTAH):

I understand that by revealing my name and test identification number that my test results will not be anonymous. During the time that my test results are not anonymous, I expect Mr. Wallace and/or his employed counselor to treat my test results in a professional, discreet, and confidential manner. After the results have been given to me, I request that Mr. Wallace destroy this form, thus returning my test findings to an anonymous status.

Research Participant

Date

APPENDIX C DATA MATRIX

Reproduced with permission of the copyright owner. Further reproduction prohibited without permission.

-

# DATA MATRIX

The following columns identify the variables in the data matrix used in this research:

<u>Column number</u>	<u>Variable</u>
1-3	Identification number
4	Sex
5,6	Number of years married
7	Stage of family life cycle
8	Religion
9	Educational achievement level
10-12	Dyadic Adjustment Scale
13,14	Defining Issues Test, P Index
15-19	Defining Issues Test, D Index
20,21	CPI, Introversion-extraversion
22,23	CPI, Norm favoring
24,25	CPI, Self-realization

------

Reproduced with permission of the copyright owner. Further reproduction prohibited without permission.

0402165131261627.05162939 0391063121182425.95162845 0392063121093628.03222343 0381104111012123.61232118 0382104121081525.02222823 0371041121043229.81122827

.

\_...

- .....

190

-

\_\_\_\_

0752011221191820.90132636 0651073241312128.37161732 0652063231193934.02112647 0561011221211224.25182625 0562011231312425.27082634 0711021231073025.23073026 0712021231112429.35133035 0881134231252737.94232842 0882134231262925.75172943 0741012221181629.49122344 0742012221202529.96042752 0541023231051822.0021333	0752011221191820.90132636 0651073241312128.37161732 0652063231193934.02112647 0561011221211224.25182625 0562011231312425.27082634 0711021231073025.23073026 0712021231112429.35133035 0881134231252737.94232842 0882134231262925.75172943 0741012221181629.49122344 0742012221202529.96042752	0321457121192024.07292732 0322457131191409.34302331 0311417121041211.30232834 0312417121021014.98292130 0301205141061523.47251939 0302215141151626.04232450 0291417111380912.36292718 0292417111321113.20163036 0281155131201720.68152731 02821551312832.37222634 0271195120991223.23202350 0272195120880615.31312028 0261145141071821.60162349 0262145131033130.58312036 0221155131052435.19121937 0222155111052225.95252447 0451164131082323.98242238 0452164131092733.05142841 0041021121220810.88222129 0042021131123223.89272432 0471104130910711.07132128 0472094130762525.17152139 0611043241103132.44132147 0612043231081323.11163333 0021001121162626.73211839 0022001131101531.92212327 0571053241352728.82122348 0572053221332528.44083055 0721063221112126.42132343 0722063221332429.39212843 0031121241441615.82091938 0032122241331121.18302437 0551012231022527.30202528
	0412467131231801.90232536 0241134131031714.98252732	0021001121162626.73211839 0022001131101531.92212327 0571053241352728.82122348 0572053221332528.44083055 0721063221112126.42132343 0722063221332429.39212843 0031121241441615.82091938 0032122241331121.18302437 0551012231022527.30202528 0552012221303038.57072634 0751011211111619.32202822 0752011221191820.90132636 0651073241312128.37161732 0652063231193934.02112647 0561011221211224.25182625 0562011231312425.27082634 0711021231073025.23073026 0712021231112429.35133035 0881134231252737.94232842 0882134231262925.75172943 0741012221181629.49122344 0742012221202529.96042752 0541023231051822.00213333 0542023220971416.97182624

\_\_\_\_\_

0671053221263031.60102842 0672053221331224.95042542 0701011221282018.12172040 0702011221202822.78202542 0601053221211823.34092930 0602053231252535.22223049 0691011221132225.55122232 0692011221151819.94172932 0621001231112030.78082644 0622001231212336.09252443 0661033231193132.15221643 0662033231101216.29312442 0591083231181518.96112428 0592083221081324.50102434 0761001231271820.58212736 0762001231242726.75232337 0771013231312535.69093144 0772013221223335.38212642 0731033241181928.44142848 0732033231222428.84273229 0631011211332422.69062342 0632011221292524.44152143 0831043221322223.43172837 0832042231301626.26262734 0891104230952221.54202339 0892094221312544.35222537 0851366241221931.25082932 0852366231251834.49213235 0861386221173126.20232744 0862376221242730.50092743 0641313221191515.07153034 0642013221292529.23092846 1271327240821221.82072645 1272337211063032.81282646 0931013220662432.04152336 0932013220772237.68211540 0921336241171927.84162747 0922336221301921.58242444 0911114221151817.18232838 0912114221263029.69312842 0901417241151928.18192640 0902417211131221.42212433 1011216230962429.17192834 1012205230962221.50272940 0811236231101523.98192834 0812236231172633.60222640 0951296241122028.49162646 0952296221041424.18222338 0791165231222430.17152941 0792175221241525.19262741 1601557211252625.45203139 1602557211222328.63172935

1071053211081925.70132435
1072043221221529.80122444
1091386241181720.73212544
1092386221072635.88242441
1141155221311618.18173134
1142155221343037.89123035
1161457221253029.80173145
1162457211351514.93243439
1211276241072627.87232642
1212276221113130.25162340
0961144231042327.57132433
0962144221082129.81132133
1001155241391422.48182451
1002155221131623.91223038
0871165221323030.97172452
0872165221192235.69142440
1121316240992724.20172935
1122316231223730.82232651
1051063220730414.22141728
1052063210612723.92171934
1041001221322630.95162947
1042001231301618.90172841
0991043241082030.59042934
0992043220933627.66142630
0681013221282129.23162643
0682013221252732.18202649
1171124241122725.89212540
1172124231054141.02222240
1081084221142226.50101928
1082084221081926.96152340
1261074211150717.78141931
1262074221111824.86221331
1251084241191842.55143151
1252083211221926.65172934
1201043230853339.63242234
1202043220852025.72121741
1181436241212934.74072350
1182436241221931.13203042
1061033231222331.74132647
1062033221163431.72072327
0941114211412326.62103447
0942114231312230.48282743
0191011131133323.24173142
0192011131122727.89152542
1131316221092840.93172839
1132315230911526.56132736
1101063241402540.25432513
1102063221371718.56433120
1291306241081321.65372121
1291306241081321.65372121
0421316210662024.57332323
0422316210812529.13472126

- --- -

......

\_\_\_\_

1792306331141824.79252733 1821457331151714.19252232	1951011331012524.06222135 1952011321201417.69102031	1821457331151714.19252232
	1792306331141824.79252733 1821457331151714.19252232 1822447321142326.04142745 1951011331012524.06222135	1841021341212930.97163042 1842021331302321.42262542

194

.

\_--

1761124341201319.86192237
1762124331243628.88182054
1551347341033135.17202444
1552347341191826.37253039

.

- - - -----

- ----

\_\_\_\_

REFERENCES

.

Reproduced with permission of the copyright owner. Further reproduction prohibited without permission.

\_\_\_\_\_

-

-

#### REFERENCES

- Abbott, Douglas A. (1983). The relation of child age, gender, and number of children to the marital adjustment of wives (Doctoral dissertation, University of Georgia). <u>Dissertation Abstracts</u> <u>International</u>, <u>44</u>, 2388A.
- Adams, Bert N. (1988). Fifty years of family research: What does it mean? <u>Journal of Marriage and the</u> <u>Family</u>, <u>50</u>, 5-17.
- American Psychological Association. (1981). Ethical principles of psychologists (revised). <u>American</u> <u>Psychologist</u>, <u>36</u>, 633-638.
- Anderson, K. J. (1983). An investigation of some predictors of marital satisfaction. (Doctoral dissertation, The University of Iowa). <u>Dissertation Abstracts International</u>, <u>44</u>, 3596A.
- Anderson, Stephen A., Russell, Candyce S., & Schumm, Walter R. (1983). Perceived marital quality and family life cycle categories: A further analysis. Journal of Marriage and the Family, 45, 127-139.
- Axelson, John A. (1985). <u>Counseling and development in a</u> <u>multicultural society.</u> Monterey, CA: Brooks/Cole.
- Bahr, Howard M., & Chadwick, Bruce A. (1985). Religion and family in Middletown, USA. <u>Journal of</u> <u>Marriage and the Family</u>, <u>47</u>, 407-414.
- Baucom, Donald H. (1985). [Review of <u>California</u> <u>Psychological Inventory</u>]. In James V. Mitchell (Ed.), <u>The nineth mental measurement yearbook</u>, vol. 1 (p. 251). Lincoln, NE: Buros Institute of Mental Measurement, University of Nebraska-Lincoln.
- Baxter, Gary Glen. (1982). Marital adjustment among clergy couples. (Doctoral dissertation, East Texas State University). <u>Dissertation Abstracts</u> <u>International</u>, <u>34</u>, 1832A.

196

- Bentler, P. M., & Newcomb, M. D. (1978). Longitudinal study of marital success and failure. <u>Journal</u> <u>of Consulting and Clinical Psychology</u>, <u>46</u>, 1053-1070.
- Billings, A. (1979). Conflict resolution in distressed and nondistressed married couples. <u>Journal of Consulting and Clinical Psychology</u>, <u>47</u>, 368-376.
- Boles, Abner Joseph III. (1984). Predictors and correlates of marital satisfaction during the transition to parenthood (longitudinal, couples, family formation). (Doctoral dissertation, University of California, Berkeley). <u>Dissertation Abstracts</u> <u>International</u>, <u>46</u>, 634B.
- Brown, Dawn, & Annis, Lawrence. (1978). Moral development level and religious behavior. <u>Psychological Reports</u>, <u>43</u>, 1230.
- Burgess, E. W., & Cottrell, L. S. (1939). <u>Predicting</u> <u>success or failure in marriage</u>. Englewood Cliffs, NJ: Prentice-Hall.
- Burgess, E. W., & Wallin, P. (1953). <u>Engagement and</u> <u>marriage</u>, New York: Lippincott.
- Burr, Wesley R. (1970). Satisfaction with various aspects of marriage over the life cycle: A random middle-class sample. <u>Journal of</u> <u>Marriage and the Family</u>, <u>32</u>, 29-37.
- Campbell, Bruce Larsen. (1972). Some factors affecting the correlation between marital satisfaction and satisfaction with the children at the launching stage of the family life cycle (Doctoral dissertation, University of Minnesota). <u>Dissertation Abstracts</u> <u>International</u>, <u>33</u>, 3812A.
- Childerston, James Kent. (1985). Understanding religious "fundamentalists": A study of typyology and moral judgment (Myers-Briggs type, development, reasoning). (Doctoral dissertation, Fuller Theological Seminary, School of Psychology, 1985). <u>Dissertation</u> <u>Abstracts International</u>, <u>46</u>, 4443B.

- Coder, C. R. (1975). Moral judgment in adults. (Doctoral dissertation, University of Minnesota, 1975). <u>Dissertation Abstracts</u> <u>Interanational</u>. <u>36</u>, 1402B.
- Cohen, Jacob. (1977). <u>Statistical power analysis for</u> <u>the behavioral sciences</u> (rev. ed.). New York: Academic Press, (a subsidiary of Harcourt Brace Jovanovich).
- Corrales, Ramon Garrido. (1974). The influence of family life cycle categories, marital power, spousal agreement, and communication styles upon marital satisfaction in the first six years of marriage (Doctoral dissertation, University of Minnesota). <u>Dissertation</u> <u>Abstracts International</u>, <u>35</u>, 3141A.
- Costa, Paul T., & McCrae, Robert R. (1980). Influence of extraversion and neuroticism on subjective well-being: Happy and unhappy people. Journal of Personality and Social Psychology, <u>38</u>(4), 668-678.
- Crowder, J. W. (1976). Effects of training upon objectivity of moral judgment in grade school children. In James R. Rest, (1979), <u>Development in judging moral issues</u> (p. 112). Minneapolis, MN: University of Minnesota Press.
- D'Antonio, W. V., Newman, W. M., & Wright, S. A. (1982). Religion and family life: How social scientists view the relationship. <u>Journal for</u> <u>the Scientific Study of Religion</u>, <u>21</u>, 218-225.
- Deters, Thomas John. (1983). The relationship of couples' moral reasoning, moral action choices, and sex-role type to marital adjustment. (Doctoral dissertation, Marquette University, 1983) <u>Dissertation Abstracts</u> <u>International</u>, <u>45</u>, 346B.
- Dixon, W. J. (Ed.). (1983). <u>BMDP statistical software</u> <u>manual</u>: <u>1985 printing</u>. Berkeley: University of California Press.

- Dortzbach, J. A. (1983). Moral judgment and perceived locus of control: A cross-sectional developmental study of adults, aged 25-74. (Doctoral dissertation, University of Oregon, 1975) <u>Dissertation Abstracts International</u>, <u>36</u>, 4662B.
- Duvall, Evelyn Millis, & Miller, Brent C. (1985). <u>Marriage and family development</u> (6th 2d.). New York: Harper and Row.
- Edmonds, Vernon H., Withers, Glenne, & Dibatista, Beverly. (1972). Adjustment, conservatism, and marital conventionalization. <u>Journal of</u> <u>Marriage and the Family</u>, <u>34</u>, 96-103.
- Edmonds, Vernon H. (1967). Marital conventionalization: Definition and measurement. <u>Journal of Marriage and the</u> <u>Family</u>, <u>29</u>, 681-688.
- Ernsberger, David J., & Manaster, Guy J. (1981). Moraldevelopment, intrinsic/extrinsic religious orientation and denominational teaching. <u>Genetic Psychology Monographs</u>, <u>104</u>, 23-41.
- Eysenck, H. J. (1985). [Review of <u>California</u> <u>psychological inventory]</u>. In James V. Mitchell (Ed.), <u>The Ninth Mental Measurement Yearbook</u>, Vol. 1 (p. 252). Lincoln, NE: Buros Institute of Mental Measurement, University of Nebraska-Lincoln.
- Eysenck, H. J. (1980). Personality, marital satisfaction, and divorce. <u>Psychological</u> <u>Reports</u>, <u>47</u>, 1235-1238.
- Garrett, William R. (1982). <u>Seasons of marriage and</u> <u>family life</u>. New York: CBS College Publishing.
- Glenn, Norval D., & Weaver, Charles N. (1978). Adjustment, conservatism, and marital conventionalization. <u>Journal of Marriage and</u> <u>the Family</u>, <u>40</u>, 269-282.
- Glenn, RNorval D., & Weaver, Charles N. (1988). The changing relationship of marital status to reported happiness. Journal of Marriage and the Family, 50, 317-324.

\_\_\_\_

Goldenberg, Irene, & Goldenberg, Herbert. (1985). <u>Family therapy: An overview</u> (2nd ed.). Monterey, CA: Brooks/Cole.

- Gray-Little, B., & Burks, N. (1983). Power and satisfaction in marriage: A review and critique. <u>Psychological Bulletin</u>, <u>93</u>, 513-538.
- Groth-Marnat, Gary. (1984). <u>Handbook of psychological</u> <u>assessment</u>. New York: Van Nostrand Reinhold.
- Gough, H. G. (1987). <u>California psychological</u> <u>inventory</u> (rev. ed.). Palo Alto, CA: Consulting Psychologists Press.
- Haan, Norma. (1986). Systematic variability in the quality of moral action, as defined in two formulations. <u>Journal of Personality and</u> <u>Social Psychology</u>, <u>50</u>(6), 1271-1284.
- Hansen, Gary L. (1981). Marital adjustment and conventionalization: A reexamination. <u>Journal</u> <u>of Marriage and the Family</u>, <u>43</u>, 855-863.
- Hart, Hornell, & Shields, Wilmer. (1926). Happiness in relation to age at marriage. <u>Journal of</u> <u>Social Hygiene</u>, <u>12</u>(7), 403-407.
- Hatch, Ruth Cordle. (1985). Marital adjustment and satisfaction as related to perceptions of religious practices and orientations: An examination of graduate and seminary student couples (religiosity, intrinsicextrinsic, quality, dyadic). (Doctoral dissertation, Kansas State University). <u>Dissertation</u> <u>Abstracts Internation</u>, <u>46</u>, 2824A.
- Hawkins, James L. (1966). The Locke Marital Adjustment Test and social desirability. <u>Journal of Marriage and The Family</u>, <u>28</u>, 193-195.
- Heaton, Tim B. (1984). Religious homogamy and marital satisfaction reconsidered. <u>Journal of Marriage</u> <u>and the Family</u>, <u>46</u>, 729-733.
- Hogan, Dennis P. (1978). The variable order of events in the life course. <u>American Sociological</u> <u>Review</u>, <u>43</u>, 573-586.

- Hogan, Dennis P. (1980). The transition to adulthood as a career contingency. <u>American Sociological</u> <u>Review</u>, <u>45</u>, 261-276.
- Houseknecht, Sharon K., Vaughan, Suzanne, & Macke, Anne S. (1984). Marital disruption among professional women: The timing of career and family events. <u>Social Problems</u>, <u>31</u>(3), 273-284.
- Hunt, Richard A., & King, Morton B. (1978). Religiosity and marriage. <u>Journal for the</u> <u>Scientific Study of Religion</u>, <u>17</u>(4), 399-406.
- Hunt, Robert Wesley. (1985). Spousal companionship, children, and marital satisfaction. (Doctoral dissertation, York University, Canada, 1984). <u>Dissertation Abstracts International</u>, <u>47</u>, 432B.
- Kaluger, George, & Kaluger, Meriem Fair. (1984) <u>Human</u> <u>development: The life span</u>. St. Louis, MO: Times Mirror/Mosby College Publishing.
- Kelly, E. Lowell, & Conley, James J. (1987). Personality and compatibility: A prospective analysis of marital stability and marital satisfaction. <u>Journal of Personality and</u> <u>Social Psychology</u>, <u>52</u>, 27-40.
- Kendall, Maurice. (1975). <u>Multivariate analysis</u>. New York: Hafner Press.
- Kerlinger, Fred N. (1986). <u>Foundations of Behavioral</u> <u>Research</u> (3rd Ed.). New York: Holt, Rinehart and Winston.
- Kohlberg, Lawrence. (1967). Moral and religious education and the public schools: A developmental view. In Theodore R. Sizer (Ed.), <u>Religion and Public Education</u> (pp. 164-183). Boston: Houghton Mifflin Company.
- Kohlberg, Lawrence. (1976). Moral stages and moralization: The cognitive-developmental approach. In Thomas Lickona (Ed.), <u>Moral</u> <u>development and behavior: Theory, research,</u> <u>and social issues</u> (pp. 31-53). New York: Holt, Rinehart and Winston.

- Lee, Paul A. (1978). The relationship between marital communication and marital adjustment over the family life cycle. (Doctoral dissertation, The Florida State University). <u>Dissertation</u> <u>Abstracts International</u>, <u>39</u>, 6998A.
- Lifton, Peter D. (1985). Individual differences in moral development: The relation of sex, gender, and personality to morality. <u>Journal</u> <u>of Personality</u>, <u>53</u>(2), 306-334.
- Locke, Don. (1986). A psychologist among the philosophers: Philosophical aspects of Kohlberg's theories. In Sohan Modgil and Celia Modgil (Eds.), <u>Lawrence Kohlberg</u>: <u>Consensus</u> <u>and controversy</u> (pp. 21-38). Philadelphia: Falmer Press.
- Locke, Harvey J. (1947). Predicting marital adjustment by comparing a divorced and a happily married group. <u>American Sociological</u> <u>Review</u>, <u>12</u>, 187-191.
- Locke, Harvey J., & Wallace, Karl M. (1959). Short marital-adjustment and prediction tests: Their reliability and validity. <u>Marriage and Family Living</u>, <u>21</u>, 251-255.
- Luckey, Eleanore Braun. (1966). Number of years married as related to personality perception and marital satisfaction. <u>Journal of Marriage</u> <u>and the Family, 28</u>, 44-48.
- Maneker, Jerry S., & Rankin, Robert P. (1985). Education, age at marriage, and marital duration: Is there a relationship? <u>Journal of</u> <u>Marriage and the Family</u>, <u>47</u>, 675-683.
- Margolin, G., & Wampold, B. E. (1981). Sequential analysis of conflict and accord in distressed and nondistressed marital partners. <u>Journal</u> <u>of Consulting and Clinical Psychology</u>, <u>49</u>, 554-567.
- Markman, H. J. (1979). Application of a behavioral model of marriage in predicting relationship satisfaction of couples planning marriage. <u>Journal of Consulting and Clinical Psychology</u>, <u>47</u>, 743-749.

- Markman, H. J. (1981). Prediction of marital distress: A 5-year follow up. <u>Journal of</u> <u>Consulting and Clinical Psychology</u>, <u>49</u>, 760-762.
- McCrae, Robert R. (1985). In James V. Mitchell (Ed.), <u>The ninth mental measurement yearbook</u>, vol. 1 (p. 304). Lincoln, NE: Buros Institute of Mental Measurement, University of Nebraska-Lincoln.
- Mueller, Daniel J. (1986). <u>Measuring social</u> <u>attitudes</u>: <u>A handbook for researchers and</u> <u>practitioners</u>. New York: Teachers College Press.
- Nims, Donald R. (1984). An examination of the relationship between the moral reasoning of marriage partners and their level of marital adjustment. (Doctoral dissertation, George Peabody College for Teachers of Vanderbilt University, 1984) <u>Dissertation Abstracts</u> <u>International</u>, <u>46</u>, 3507A.
- Noller, P. (1980). Misunderstandings in marital communication: A study of couples' nonverbal communication. Journal of Personality, 39, 1135-1148.
- Norton, Robert. (1983). Measuring marital quality: A critical look at the dependent variable. Journal of Marriage and the Family, <u>45</u>, 141-151.
- O'Gorman, T. P. (1979). An investigation of moral judgment and religious knowledge scores of Catholic high-school boys from Catholic and public schools. <u>Dissertation Abstracts</u> <u>International</u>, <u>40</u>, 1365A.
- Otero de Sabogal, Regina (1984). Effects of parenting on marital quality: A causal analysis (family roles, gender differences, role strain). (Doctoral dissertation, North Texas State University). <u>Dissertation Abstracts</u> <u>International</u>, <u>45</u>, 23188.
- Peterson, Gail Beaton, Hey, Richard N., & Peterson, Larry Richard. (1979). Intersection of family development and moral stage frameworks: Implications for theory and research. Journal of Marriage and the Family, 41, 229-235.

- Piaget, Jean. (1965). <u>The moral judgment of the</u> <u>child</u>. (M. Gabain, trans.) New York: Free Press. (Original work published 1932)
- Pickford, John H., Signori, Edro I., & Rempel, Henry. (1966). The intensity of personality traits in relation to marital happiness. Journal of <u>Marriage and the Family</u>, <u>28</u>, 459-464.
- Register, Mary Claire. (1975). Moral compatibility in married couples: A study of marital satisfaction as related to stage of moral development of spouses. (Doctoral dissertation, University of Georgia, 1975) <u>Dissertation Abstracts</u> <u>International</u>, <u>36</u>, 4176B.
- Rest, G. J. (1977). Voting preference in the 1976 presidential election and the influences of moral reasoning. In James R. Rest (1979), <u>Development in judging moral issues</u> (p. 112). Minneapolis, MN: University of Minnesota Press.
- Rest, James R. (1979). <u>Development in judging moral</u> <u>issues</u>. Minneapolis, MN: University of Minnesota Press, University of Minnesota: Center for the Study of Ethical Development.
- Rest, James R. (1986a). <u>DIT manual for the defining</u> <u>issue test</u>, (3rd ed.). University of Minnesota: Center for the Study of Ethical Development.
- Rest, James R. (1986b). <u>Moral development: Advances</u> <u>in research and theory</u>. New York: Praeger Publishers.
- Rest, James R. (personal communication, June 24, 1988).
- Rollins, Boyd C., & Cannon, Kenneth L. (1974). Marital satisfaction over the family life cycle: A reevaluation. Journal of Marriage and the Family, <u>36</u>, 271-282.
- Rollins, Boyd C., & Feldman, Harold. (1970). Marital satisfaction over the family life cycle. Journal of Marriage and the Family, 32, 20-27.

Rychlak, Joseph F. (1981). <u>Introduction to</u> <u>personality and psychotherapy</u>. Boston: Houghton Mifflin Company.

- Ryder, Robert G. (1973). Longitudinal data relating marriage satisfaction and having a child. Journal of Marriage and the Family, 35, 604-606.
- Sapp, Gary L. (1986). Moral judgment and religious orientation. In Gary L. Sapp (Ed.), <u>Handbook</u> <u>of moral development: Models</u>, <u>processes</u>, <u>techniques</u>, <u>and research</u> (pp. 271-286). Birmingham, AL: Religious Education Press.
- Schumm, Walter R., Bollman, Stephan R., & Jurich, Anthony P. (1982). The "marital conventionalization" argument: Implications for the study of religiosity and marital satisfaction. Journal of Psychology and Theology, 10(3), 236-241.
- Schumm, Walter R., & Bugaighis, Margaret A. (1986). Marital quality over the marital career: Alternative Explanations. <u>Journal of Marriage</u> <u>and the Family</u>, <u>48</u>, 165-168.
- Schumm, Walter R., Hess, James L. Bollman, Stephan R., & Jurich, Anthony P. (1981). Marital conventionalization revisited. <u>Psychological</u> <u>Reports</u>, <u>49</u>, 607-615.
- Schumm, Walter R., & Kirn, Julia E. (1982). Evaluating equity in the marital relationship: An alternative approach. <u>Psychological</u> <u>Reports</u>, <u>51</u>, 759-762.
- Sharpley, C. F., & Cross, D. G. (1982). A psychometric evaluation of the Spanier dyadic adjustment scale. <u>Journal of Marriage and the</u> <u>Family</u>, <u>44</u>(3), 739-741.
- Snowden, Lonnie R., Schott, Tracy L., Awalt, Suzanne J., & Gillis-Knox, Jo. (1988). Marital satisfaction in pregnancy: Stability and change. Journal of Marriage and the Family, 50, 325-333.
- Spanier, Graham B. (1976). Measuring dyadic adjustment: New scales for assessing the quality of marriage and similar dyads. <u>Journal</u> of Marriage and the Family, <u>38</u>(1), 15-29.

Spanier, Graham B., Lewis, Robert A., & Cole, Charles L. (1975). Marital adjustment over the family life cycle: The issue of curvilinearity. <u>Journal of Marriage and the Family</u>, <u>37</u>, 263-275.

- Steinberg, Laurence, & Silverberg, Susan B. (1987).
  Influences on marital satisfaction during the
  middle stages of the family life cycle.
  Journal of Marriage and the Family, 49, 751760.
- Terman, Lewis M. (1938). <u>Psychological factors in</u> <u>marital happiness</u>. New York: McGraw-Hill.
- Terman, Lewis M. (1950). Predicting marriage failure from test scores. <u>Marriage and Family Living</u>, <u>12</u>, 51-55.
- Thomas, Darwin L., & Henry, Gwendolyn C. (1985). The religion and family connection: Increasing dialogue in the social sciences. <u>Journal of</u> <u>Marriage and the Family</u>, <u>47</u>, 369-379.
- U. S. Bureau of the Census. (1987). <u>National data</u> <u>book and guide to statistical abstract of the</u> <u>United States 1988</u> (108th ed.). Washington, DC.
- Vaillant, G.E. (1978). Natural history of male psychological health: VI. Correlates of successful marriage and fatherhood. <u>American</u> <u>Journal of Psychiatry</u>, <u>135</u>, 653-659.
- Vines, Neville R. (1979). Adult unfolding and marital conflict. <u>Journal of Marital and Family</u> <u>Therapy</u>, 5-14.
- Wahrman, Israel S. (1981). The relationship of dogmatism, religious affiliation, and moral judgment development. <u>The Journal of</u> <u>Psychology</u>, <u>108</u>, 151-154.
- Welkowitz, J., Ewen, R., & Cohen, J. (1982). <u>Introductory statistics for the behavioral</u> <u>sciences</u>. San Diego: Harcourt Brace Jovanovich.

- Wilson, Margaret R., & Filsinger, Erik E. (1986). Religiosity and marital adjustment: Multidimensional interrelationships. Journal of Marriage and the Family, 48, 475-479.
- Wilson, John P., & Wilson, Stephen B. (1977). Psychological needs and moral judgment. <u>Psychological Reports</u>, <u>41</u>, 475-479.

## VITA

.

-----

-

	NAME: MARITAL STATUS: CHILDREN: OFFICE TELEPHONE: HOME TELEPHONE: HOME ADDRESS:	Thomas C. Wallace Married 4 (801)373-4765 (801)225-3847 1051 S. Main Orem, UT 84058
	Time period:	1977-1985
	Title: pathology	Private practice in speech/language
	Employer:	Self
	Time period:	1974-76
	Title:	Coordinator, Speech, Language, and Hearing Services
	Employer:	Umatilla Intermediate Education District Pendleton, Oregon
	Time period:	1967-1974
	Title:	Speech pathologist
	Employer:	Umatilla Intermediate Education District Pendleton, Oregon
EDUCATIONAL BACKGROUND:		
	1970	M.S. Speech Pathology and Audiology Western Oregon State College Monmouth, Oregon
	1067	

1967 B.A. Speech Walla Walla College College Place, Washington